LV 5750

MULTI SDI MONITOR

INSTRUCTION MANUAL

LEADER ELECTRONICS CORP.

TABLE OF CONTENTS

GE	INER	AL SAFE	TY SUMMARY	I
1				1_1
1.	1 1	Scope	of Warranty	1_1
	1.1	Operati	na Precautions	1_1
	1.2	1 2 1	Power Supply	1_1
		1.2.1	Damage Caused by Static Electricity	· · 1_1
		123	Maximum Allowable Input Voltage	1-2
		1.2.0		1_2
		1.2.4	Mechanical Shock	1-2
		126	Last Memory and Default Settings	1_3
		127	East Memory and Delaut Cettings	1_4
		128	I CD Monitor	1-4
		129	Elickering of the LCD	1-4
		1 2 10	About the CE card function	1-5
		1.2.10		
2.	SPE	CIFICAT	IONS	2-1
	2.1	Genera	1	2-1
	2.2	Feature	9S	2-1
	2.3	Specific	ations	2-4
		2.3.1	Video Formats and Corresponding Standards	2-4
		2.3.2	Input/Output Connector	2-4
		2.3.3	Display Format	2-6
		2.3.4	Waveform Display	2-6
		2.3.5	Vectorscope Display	2-7
		2.3.6	Picture Display	2-8
		2.3.7	Embedded Audio Display	2-8
		2.3.8	Status Display	2-8
		2.3.9	Screen Capture	2-10
		2.3.10	Presets	2-10
		2.3.11	Other Display Settings	2-10
		2.3.12	General Specifications	2-11
3.	PAN	EL DESO	CRIPTION	3-1
	3.1	Front P	anel	3-1
	3.2	Rear Pa	anel	3-3
	3.3	Top and	Bottom Panels	3-5
	3.4	Note		3-7
	3.5	Power S	Supply	3-8

	3.6 Power Switch			
	3.7	Connec	tions to the Rear Panel Connectors	3-9
		3.7.1	SDI INPUT Connector	3-9
		3.7.2	EXT REF Connector	3-9
		3.7.3	SDI OUTPUT Connector	3-9
	3.8	Externa	I IF Slot	3-10
		3.8.1	Compact Flash Card Unit	3-10
		3.8.2	Ethernet Unit	3-10
		3.8.3	Remote Control Unit	3-11
4.	MEN	IU STRU	ICTURE	4-1
	4.1	Wavefo	rm Display Menu	4-1
	4.2	Vectors	cope Display Menu	4-3
	4.3	Picture	Display Menu	4-4
	4.4	Audio D	Display Menu	4-5
	4.5	Multi Di	splay Menu	4-7
	4.6	Status [Display Menu	4-15
	4.7	Capture	e Menu	4-18
	4.8	System	Menu	4-19
	4.9	Preset I	Menu	4-21
5.	SYS	ТЕМ СО	NFIGURATION BEFORE WAVEFORM MONITORING	5-1
	5.1	SDI Inp	ut Signal Selection	5-1
		5.1.1	SDI Input Signal	5-1
		5.1.2	SDI Input Selection	5-1
	5.2	Externa	I Synchronization Signal Setting	5-2
		5.2.1	External Sync Signals	5-2
		5.2.2	External Sync Signal Switching	5-3
	5.3	System	Menu	5-3
	5.4	Video S	Signal Format Setting	5-4
		5.4.1	Auto/Manual Selection	5-5
		5.4.2	Interlace and Segment Frame Selection	5-5
		5.4.3	Scanning Selection	5-6
		5.4.4	Field/Frame Frequency Selection	5-6
		5.4.5	Format Selection of the Pseudo-Composite Display	5-7
	5.5	Screen	Display Settings	5-7
		5.5.1	Information Display	5-8
				F O
		5.5.2	Backlight Switching	5-9
	5.6	5.5.2 Interfac	e Setting	5-9 5-10
	5.6	5.5.2 Interfac 5.6.1	Backlight Switching e Setting Recalling Presets through the REMOTE Connector	5-9 5-10 5-10

	5.7	Date an	d Time Settings	5-11
	5.8	Shortcu	t Key Setting	5-12
	5.9	Initializa	tion	5-13
6.	VIDE	O SIGN/	AL WAVEFORM DISPLAY	6-1
	6.1	Video S	ignal Waveform Display	6-1
	6.2	Display	Channel Settings	6-2
	6.3	Display	Mode Settings	6-2
		6.3.1	Overlay Display	6-3
		6.3.2	Parade Display	6-3
		6.3.3	Timing Display	6-4
	6.4	Position	Adjustment	6-4
	6.5	Brightne	ess Adjustment	6-5
		6.5.1	Brightness Adjustment of the Video Signal Waveform	6-5
		6.5.2	Scale Brightness Adjustment	6-5
	6.6	Gain/Fil	ter Adjustment	6-6
		6.6.1	Gain Adjustment	6-6
		6.6.2	Gain Selection	6-7
		6.6.3	Filter Selection	6-7
	6.7	Sweep \$	Settings	6-8
		6.7.1	Sweep Selection	6-8
		6.7.2	H Sweep Selection	6-8
		6.7.3	V Sweep Selection	6-9
		6.7.4	Field Selection	6-10
		6.7.5	Sweep Magnification	6-10
	6.8	Line Sel	lector	6-11
		6.8.1	Line Select	6-11
		6.8.2	Field Selection	6-11
	6.9	Cursor I	Measurement	6-12
		6.9.1	Cursor Display	6-12
		6.9.2	X-Axis/Y-Axis Cursor Selection	6-12
		6.9.3	Cursor Measurement Unit Selection	6-13
		6.9.4	Cursor Movement	6-13
		6.9.5	Reference Amplitude Setting	6-14
	6.10	Color Sy	ystem Setting	6-14
		6.10.1	Color Matrix Conversion	6-15
		6.10.2	Y-GBR (R, G, B) Display	6-16
		6.10.3	Setup Selection	6-16
		6.10.4	Pseudo-Composite Display	6-17

	6.11	Scale S	ettings	6-17
		6.11.1	Scale Unit Selection	6-18
		6.11.2	Display Selection of the 75 % Color Bar Scale	6-18
		6.11.3	Scale Color Selection	6-19
	6.12	Blanking	g Display Settings	6-19
	6.13	Mode K	ey Switching	6-20
7.	VEC	TORSCO	OPE DISPLAY	7-1
	7.1	Vector V	Naveform Display	7-1
	7.2	Brightne	ess Adjustment	7-2
		7.2.1	Brightness Adjustment of Vector Waveforms	7-2
		7.2.2	Scale Brightness Adjustment	7-2
		7.2.3	IQ Axis Display	7-3
		7.2.4	Scale Color Selection	7-3
	7.3	Gain Ad	ljustment	7-4
		7.3.1	Gain Variable	7-4
		7.3.2	Gain Selection	7-5
	7.4	Line Sel	lector	7-5
		7.4.1	Line Select	7-5
		7.4.2	Field Selection	7-5
	7.5	Color Sy	ystem Settings	7-6
		7.5.1	Composite and Component Display	7-6
		7.5.2	Setup Selection	7-7
		7.5.3	Color Bar Saturation Selection	7-7
8.	PICT	URE DIS	SPLAY	8-1
	8.1	Marker	Display	8-2
		8.1.1	Aspect Marker (4:3)	8-3
		8.1.2	Aspect Marker (16:9)	8-4
		8.1.3	Safe Action Marker	8-4
		8.1.4	Safe Title Marker	8-4
		8.1.5	Center Marker	8-4
	8.2	Line Sel	lect Marker Display	8-5
		8.2.1	Line Select Marker Display	8-5
		8.2.2	Field Selection of Line Select	8-5
	8.3	Picture	Size Change	8-6

9.	AUD	AUDIO DISPLAY			
	9.1	Audio W	Vaveform Display Selection	9-2	
	9.2	Embedo	ded Audio Group Selection	9-3	
	9.3	Audio W	Vaveform Display Setting	9-3	
		9.3.1	Brightness Adjustment of Audio Waveforms	9-3	
		9.3.2	Scale Brightness Adjustment	9-3	
		9.3.3	Sound and Lissajou Display Format	9-4	
		9.3.4	Lissajous Display Channel Selection	9-5	
		9.3.5	Gain Selection	9-5	
	9.4	Audio Lo	evel Meter Setting	9-6	
		9.4.1	Reference Level Selection	9-6	
		9.4.2	Dynamic Range Selection	9-6	
		9.4.3	Scale Selection	9-7	
		9.4.4	Peak Hold	9-8	
	9.5	Headph	one Output Settings	9-8	
		9.5.1	Headphone Output ON/OFF	9-8	
		9.5.2	Headphone Output Volume	9-8	
		9.5.3	Headphone Output Channel Selection	9-8	
	9.6	Channe	I Mapping	9-9	
10.	10. MULTI SCREEN DISPLAY 10-				
	10.1	Five Mu	Iti Screen Modes	10-2	
	10.2	Display	Content Selection in 4 Screen Mode	10-3	
	10.3	Various	Settings on the Multi Screen Display	10-3	
		10.3.1	Video Signal Waveform Display Setting on the Multi Screen Display	10-3	
		10.3.2	Vectorscope Waveform Display Setting on the Multi Screen Display	10-4	
		10.3.3	Audio Setting on the Multi Screen Display	10-4	
		10.3.4	Status Setting on the Multi Screen Display	10-4	
11.	STAT	US DISI	PLAY	11-1	
	11.1	Status D	Display Screen	11-2	
		11.1.1	Status Indication of SDI Signals	11-2	
		11.1.2	Status Display of Ancillary Data	11-5	
		11.1.3	Status Display of the Video Signal Level	11-5	
		11.1.4	Status Display of Embedded Audio	11-6	
		11.1.5	Other Status Displays	11-7	
	11.2	Event Lo	og	11-8	
		11.2.1	Event log list display	11-9	
		11.2.2	Starting/Stopping the Event Log	11-10	
		11.2.3	Clearing the Event Log	11-10	

	11.2.4	Log Operation Mode	11-10
	11.2.5	Storage to the Compact Flash Card	11-10
	11.2.6	File List of the Compact Flash Card	11-12
	11.2.7	Log File Deletion	11-12
	11.2.8	Auto Increment of File Names	11-13
11.3	Data Du	ımp Display	11-13
	11.3.1	Data Dump Display Auto Updating and Hold	11-13
	11.3.2	Data Dump Display Format Selection	11-14
	11.3.3	Auto Search of EAV/SAV	11-15
	11.3.4	Line Number and Sample Number Selection	11-15
	11.3.5	Storage to the Compact Flash Card	11-15
	11.3.6	File List of the Compact Flash Card	11-17
	11.3.7	Data Dump File Deletion	11-17
11.4	Status D	Display of Embedded Audio	11-18
	11.4.1	Channel Selection	11-18
	11.4.2	Voice Control Packets	11-19
	11.4.3	Channel Status	11-19
	11.4.4	Channel Status List Display	11-20
11.5	Ancillary	y Data Display	11-20
	11.5.1	EDH Status Display	11-21
11.6	Error De	etection Setting	11-22
	11.6.1	Error Alarm Setting	11-23
	11.6.2	Error Count Rate Setting	11-23
	11.6.3	Detection ON/OFF of Errors	11-23
11.7	Detectio	on Value Setting of Errors	11-26
	11.7.1	Gamut Error Threshold Level	11-26
	11.7.2	Composite Gamut Error Threshold Level	11-27
	11.7.3	Detection Value of Equivalent Cable Length Meter Error	11-28
11.8	Error Di	splay Operation	11-29
11.9	Cable S	election of the Cable Length Meter	11-29
	11.9.1	HD-SDI Cable Selection	11-29
	11.9.2	SD-SDI Cable Selection	11-30
11.10) Error F	Reset	11-30

12. CAF	TURE FU	JNCTION	12-1
12.1	Screen	capture	12-2
12.2	2 Display	Selection	12-2
12.3	Storage	to the Compact Flash Card	12-2
12.4	Recallin	g from the Compact Flash Card	12-5
12.5	File List	of the Compact Flash Card	12-5
12.6	6 Deletion	of Capture Files	12-5
12.7	' Auto Inc	crement of File Names	12-5
12.8	Storage	File Selection	12-5
13. PRE	ESET FUN	NCTION	13-1
13.1	Recallin	g of Settings	13-1
13.2	Registra	ation of Settings	13-1
	13.2.1	Comment Entry	13-2
	13.2.2	Storing of the Settings	13-3
	13.2.3	Deleting of the Settings	13-3
	13.2.4	Editing of Presets	13-3
13.3	Copying	of All Presets	13-3
	13.3.1	Copying from the CF Card to the Internal Memory	13-3
	13.3.2	Copying from the Internal Memory to the CF Card	13-4
14. ADI	DITIONAL	FUNCTION	14-1
14.1	Specific	ations	14-1
	14.1.1	General	14-1
14.2	New Fu	nctions	14-1
14.3	8 Menu S	tructure	14-3
14.4	System	Configuration	14-8
	14.4.1	Additional Video Signal Formats	14-8
	14.4.1.1	Video Signal Formats That Have Been Added	14-8
	14.4.1.2	2 Video Signal Format Setting	14-8
	14.4.2	Time Code	14-9
	14.4.2.1	Time Code Display	14-9
	14.4.2.2	2 LTC/VITC Selection	14-9
	14.4.3	Remaining Battery Power Display	14-10
	14.4.4	SNMP Support	14-10

	14.5	Bar Disp	olay	14-11
		14.5.1	Limit Setting	14-12
		14.5.2	Gamut Display Frequency Characteristics	14-12
		14.5.3	Miscellaneous	14-12
		14.5.4	Details of Bar Display	14-12
	14.6	Status D	Display	14-14
		14.6.1	Supplementary Information Concerning the Status Display	14-14
		14.6.2	Audio CRC Detection Function	14-15
		14.6.2.1	Status Display of the Audio CRC	14-15
		14.6.2.2	2 Event Log	14-15
		14.6.2.3	Turning the Error Detection ON/OFF	14-15
		14.6.3	V-ANC Monitor Function	14-16
		14.6.3.1	ANC_PACKET Summary Display	14-16
		14.6.3.2	P Format ID Display	14-19
		14.6.3.3	Closed Caption Data Display	14-20
		14.6.3.4	Inter-Stationary Control Signal	14-23
		14.6.4	Displaying Ancillary Data Markers Using Data Dump	14-26
15.	SDI	EXT RE	F PHASE DIFFERENCE DISPLAY FUNCTION	15-1
	15.1	General		15-1
	15.2	Feature	S	15-1
	15.3	Specific	ations	15-1
	15.4	MENUE	Structure	15-4
	15.5	Operatio	on	15-5
	15.6	Zero Re	ference	15-7

GENERAL SAFETY SUMMARY

To Avoid Personal Injury

It is recommended that only qualified personnel with technical knowledge use this instrument only after reading and fully understanding all functions of the instrument described this instruction manual.

This instrument is not designed and manufactured for consumers.

If you do not have enough knowledge on electricity, to avoid personal injury and prevent damage to this product, please be sure to use this product only under the supervision of an engineer who has sufficient knowledge about electronics.

Precautions on Contents

Should you find the contents in this manual and any of its technical terms confusing, please feel free to contact your local LEADER agent.

Symbols and Terms

Following terms and symbols indicate necessary warnings and cautions used in this manual and on the product are there for safe operation.

< Symbol >	The sections where this symbol is marked in this manual or instrument, if not correctly performed or practiced, could result in personal injury or cause serious danger to the instrument. Misuse could also produce unintentional movement to create an operational impediment on the instrument or other products that might be connected to it. Be sure to refer to the safety precautions in this manual to safely use the part of the instrument where the symbol is marked.
< Term >	Warning statements identify warning conditions that if disregarded or not correctly performed or adhered to, could result in serious personal injury or even loss of life.
< Term >	Caution statements identify caution conditions that if disregarded or not correctly performed or adhered to, could result in personal injury or damage to the instrument.

GENERAL SAFETY SUMMARY

Review the following safety precautions to avoid operator's injury and loss of life and prevent damage and deterioration to this instrument. To avoid potential hazards, use this product as specified.



Warnings on the Cases and Panels of the Instrument

Operator should not remove any cases or panel for any reasons. If you touch inside the instrument it could result personal shock or fire hazard. Refrain from spilling any liquid on or inserting anything flammables or piece of metal into the ventilation of the instrument. Such actions could cause fire, shock, malfunction and be an accident hazard while the power is on.

Warnings Concerning the Power Line

Use a line voltage in the range of 10 VDC to 18 VDC. A line voltage exceeding this range may cause fire.

Warning on Installation Environments

About the Operating Temperature Range

Operate the instrument between the temperature range of 0 to 40 °C. Operating the instrument at higher temperatures could cause a fire hazard.

Rapid changes of temperatures from cold to warm can create internal moisture or condensation and could damage the instrument. If there is a possibility of moisture condensation allow the instrument to sit for 30 minutes without the power on.

About the Operating Humidity Range

Operating humidity range is \leq 85 % RH.

Do not operate the instrument with wet hands, this could cause a shock and fire hazard.

About the Operation in the Presence of Gasses

Operating the instrument in and near the presence or storage locations of flammable, explosive gasses or fumes could create an explosion and fire hazard. Do not operate the instrument anywhere near such environments.

Avoid Insertions

Do not insert metals or flammable objects or drop liquid on or into the instrument. To do so could cause fire, shock, malfunction and create a dangerous accident hazard.

Warning while Operating

While operating the instrument if smoke, fire, or a bad smell occurs, turn off the instrument at once for it could cause a fire hazard. When such a case occurs, turn off the power switch and pull the plug of the cord from the plug socket. Contact your local LEADER agent after confirming there is no fire.

GENERAL SAFETY SUMMARY



Warnings Concerning the LCD Panel

Glass is used on the LCD panel. If you break the glass, the fragments may cause injury. Do not apply strong shock to the glass or damage the surface of the glass with sharp metal objects.

Caution on Input/Output Terminals

Input Terminals are rated with a maximum input. Do not supply an input over the specified rating in the standard section of the instruction manual. Also, do not supply external power to Output terminal, this could cause the instrument to malfunction.

Caution when Not to Use the Instrument for a Long Time

Make sure to disconnect the power cord from the socket or remove a battery from the instrument when you do not use the instrument for a long time.

Please conform to the above warnings and cautions for safe operation. There are cautions in each area of this instruction manual, so please conform to each caution. If you have any questions about this manual, please feel free to contact your local LEADER agent.

Calibration

This instrument is produced under the strictest quality controls at the factory, but accuracy may gradually deteriorate due to worn components.

Therefore, periodic calibration should be performed.

When service or calibration is required, contact your local LEADER agent.

Routine Maintenance

Remove the power cord plug from the socket when cleaning the instrument.

Avoid the use of thinner or benzene solvents for cleaning cases, panels and knobs since this might remove the paint or damage plastic surfaces.

Wipe cases, panels, and knobs lightly with a soft cloth damped with neutral detergent. Do not allow water, detergent, or other foreign objects to enter the instrument while cleaning.

If a liquid or metal object enters the instrument, it can cause electric shock or fire.

1. INTRODUCTION

Thank you for purchasing LEADER's measuring instruments.

Please read this instruction manual carefully to ensure correct and safe operation. If you have any difficulties or questions on how to use the instrument after you have read this manual, please feel free to contact your local LEADER agent.

After you have read the manual, keep the manual in a safe place for quick reference.

1.1 Scope of Warranty

This LEADER instrument has been manufactured under the strictest quality control guidelines. LEADER shall not be obligated to furnish free service during the warranty period under the following conditions.

- 1. Repair of malfunction or damages resulting from fire, natural calamity, or improper voltage applied by the user.
- 2. Repair of an instrument that has been improperly repaired, adjusted, or modified by personnel other than a factory-trained LEADER representative.
- 3. Repair of malfunctions or damages resulting from improper use.
- 4. Repair of malfunctions caused by devices other than this instrument.
- 5. Repair of malfunctions or damages without the presentation of a proof of purchase or receipt bill for the instrument.

1.2 Operating Precautions

1.2.1 Power Supply

The LV 5750 uses a DC power supply. It will not operate on AC power. If you apply AC power by mistake, it may cause a fire or malfunction.

The voltage range of the DC power supply is 10 V to 18 V. Do not apply voltage exceeding this range as it can cause a malfunction or fire.

There are two power supply input connectors, a V-mount adapter, which is assumed for battery use, and an XLR connector.

If you apply power to both connectors simultaneously, the LV 5750 will operate on the power supplied by the XLR connector. However, please avoid switching the power supply during operation as it may cause erroneous operation. If you need to switch the power supply, turn OFF the power to do so.

1.2.2 Damage Caused by Static Electricity

Electronic parts are susceptible to damage from electrostatic discharge (ESD). The core wire of the coaxial cable may be charged with static electricity. If you are connecting a coaxial cable that is not connected on either end to an input/ output connector of the instrument, short the cable's core wire and external conductor beforehand.

1.2.3 Maximum Allowable Input Voltage

The maximum allowable input voltage to the input connectors is shown in the table below.

Do not apply excessive voltage to prevent damage to the instrument. Refer to Installation Category I (IEC 1010-1, Annex J).

Input Connector	Maximum Allowable Input Voltage
SDI INPUT	±2 V (DC + peak AC)
EXT REF	±5 V (DC + peak AC)
REMOTE	Input TTL level Within +5 V to -0.3 V

The maximum input voltage of " \pm 5 V (DC + peak AC)" is as shown in Figure 1.1.



Figure 1.1 DC + Peak AC

1.2.4 Installation

Do not use the instrument in the following environments.

• High temperature environments

Do not place the instrument under direct sunlight or near a heater (e.g., stove). Avoid using the instrument in a way that leads to drastic changes in temperature such as moving the instrument from a cold environment to a hot environment; this may cause condensation.

Operating temperature range: 0 to 40 °C

- High humidity environments
 Do not place the instrument in a high humidity environment (e.g., bathroom, near a humidifier).
 Operating humidity range: ≤ 85 % RH (without condensation)
- Dusty environments
- Strong magnetic fields Do not place the instrument near a strong magnetic source (e.g., high-power transformer).

1.2.5 Mechanical Shock

Do not apply strong shock to the instrument. A strong shock can break precision parts of the instrument such as the internal crystal oscillator or LCD and can cause a malfunction.

If the LCD breaks, grass fragments may cause injury.

1.2.6 Last Memory and Default Settings

The last memory function retains the front panel settings that exist immediately before the power is turned off. When the power is turned on again, the front panel settings are set to the previous settings.

If the battery runs out or the power supply is cut off with the power turned ON, the LV 5750 will be ON the next time the power is supplied.

However, if the backup battery for the last memory function runs out, a message as shown in the figure below appears when the LV 5750 starts. If this happens, contact your local LEADER agent and request repairs.

If you press the WFM key, the LV 5750 will operate using initial settings.



Figure 1.2 Message that appears when the backup battery runs out

1.2.7 Fan Stop Message

The LV 5750 has a fan near the rear panel for internal cooling.

If the fan stops due to a malfunction, the following message blinks at 1-second intervals. If this happens, turn OFF the power immediately, contact your local LEADER agent, and request repairs.



Figure 1.3 Fan Stop Message

1.2.8 LCD Monitor

There may be pixels on the LCD monitor that do not turn ON or those that remain ON at all times. This is not a malfunction.

1.2.9 Flickering of the LCD

The LCD on the LV 5750 supports many types of video signals. The LCD displays the input SDI signal asynchronously and may flicker on the waveform display or picture display. The LV 5750 stores the input SDI signal once to the frame memory, and then reads the memory using the LCD sync signal, which is asynchronous to the input SDI signal. Therefore, flickers may occur due to frame skip, which skips over certain frame memories, or frame repeat, which reads certain frame memories twice.

1.2.10 About the CF card function

Explain the CompactFlash (henceforth, CF) memory card function that can be used in the LV 5750.

About a CF memory card, operation is being confirmed up to 2GB now. The file system supports only FAT16.

The operation of the reading and writing of the file has been confirmed with the CF memory card in the table below.

In the CF memory cards other than being described here, reading and writing may be normally impossible.

CF memory card manufacture	Model number	Capacity
SanDisk	SDCFB-32MB	32MB
SanDisk	SDCFB-64MB	64MB
SanDisk	SDCFB-128MB	128MB
SanDisk	SDCFB-256MB	256MB
SanDisk	SDCFB-512MB	512MB
SanDisk	SDCFB-2.0GB	2.0GB

* Use a CF memory card that complies with CFA standards.

- * A Microdrive can not be used.
- * CF memory card of high-speed correspondence is not supported.
- * CompactFlashTM is the trademark of U.S. SanDisk.
- * In addition, company names, product names, etc. that have been described are the trademarks, and registered trademarks of each company.

2. SPECIFICATIONS

2.1 General

The LV 5750 is a waveform monitor for HD-SDI and SD-SDI signals. The LV 5750 employ color TFT LCD screen.

The functions of waveform monitors, vectorscopes, audio lissajous, and simple picture monitors are achieved with a single unit.

Complete digital processing of SDI signals enables highly accurate measurements. In addition, extensive error detection functions and analysis functions are provided which allows the LV 5750 to be used as SDI signal monitors.

2.2 Features

Two serial digital input systems

The SDI input connector on the LV 5750 supports free rates. Thus, either HD-SDI or SD-SDI signals can be applied to the same connector. You can select auto or manual setting for the input signal format.

SDI Output

Equipped with an active output connector that reclock the input signal.

Display

Employs a color TFT LCD monitor with XGA resolution (1,024 x 768).

The LCD monitor can show various displays such as waveform display, vectorscope display, picture display, audio display, and status display. These displays can also be placed side by side on a single screen (multi display).

Waveform display

Displays waveforms of serial digital signals side by side or overlaid.

The amplitude can be set to variable or expanded up to 5 times.

The time axis can be expanded 10, 20 or 40 times.

Enables level measurements and time measurements using cursors.

In addition, a composite signal conversion function that artificially converts component signals into composite signals such as NTSC or PAL is provided. This function allows the level of component signals to be managed in a similar fashion as conventional analog signals.

Vectorscope function

Displays chrominance difference signal in vector format.

The amplitude can be set to variable, IQ-MAG, or expanded up to 5 times.

You can turn on/off the IQ axis display, which is convenient for monitoring signals.

Picture display

Displays SDI signals on a simple color picture display.

Line selector

Enables the selection and display of arbitrary video signal lines.

Can be used on the waveform display, vectorscope display, picture display, and data dump display.

• Embedded audio signal display function

Separates the embedded audio signal from the serial digital signal and shows lissajous display, sound image, and level meter display.

Level display shows eight channels, and the range and response can be changed. * The audio quantization precision of SD-SDI is 20 bits.

Screen capture

Enables the displayed screen to be captured to the internal memory or *1 compact flash memory card. The BMP data of the screen image can be retrieved into a PC via the network using *2 FTP.

When the internal memory or *1compact flash memory card is used, the captured screen and the signals that are input sequentially can be superimposed on the screen.

Extensive Analysis Functions

- Various types of error detection Detects *³ CRC errors, *³ EDH errors, *³ BCH errors, checksum errors. Error detection is useful in the monitoring of SDI signals.
- Event log function of SDI signals

Logs detected errors and events and stores the log list to the *1 compact flash memory card. The log can be retrieved as text data into a PC via the network using *2 FTP.

Digital data dump function

The digital data after parallel conversion can be displayed in hexadecimal or binary notation. This is useful when problems occur or when analyzing various types of data.

The data dump can be stored to a ^{*1} compact flash memory card. The data dump can be retrieved as text data into a PC via the network using ^{*2} FTP.

Analysis display

Analyzes and displays voice control packets embedded in the SDI signal, the channels status, EDH packets, and other packets in the AES/EBU signal.

Operation

Remote control

The LV 5750 can be controlled through the panel and remotely controlled through a computer via the ^{*2} Ethernet network. In addition, presets can be recalled using the ^{*4} remote connector on the rear panel.

Preset function

Stores up to 30 sets of frequently used panel settings.

You can easily recall stored settings from the panel, *2 Ethernet network connector or *4 remote connector.

External Synchronization

Allows tri-level sync signals or B. B signals of NTSC and PAL to be input.

Illumination of all key LEDs

You can turn on all the key LEDs. This is useful when operating the LV 5750 in a very dark place.

Power supply

A V-mount battery adapter is available on the rear panel of the LV 5750 which enables the use of batteries used on video cameras and other equipment. An XLR DC input connector is also available for operation on a 12-VDC power supply.

Tripod attachment

A screw hole for attaching a camera tripod is provided on the bottom panel of the LV 5750. The LV 5750 can be mounted on a tripod.

Dedicated rack mount adapter (sold separately)

By using the dedicated rack mount adapter that is sold separately, the LV 5750 can be rack mounted on a rack with short depth.

To rack mount the LV 5750, remove the outer case and use the dedicated rack mount adapter.

- * 1 A compact flash memory card unit that comes standard is required. In this case, the Ethernet Unit and the Remote Control Unit that are sold separately cannot be used.
- * ² An Ethernet Unit sold separately is required.
 In this case, the compact flash memory card unit that comes standard and the Remote Control Unit that is sold separately cannot be used.
- *³ CRC error and BCH error detection is used for HD-SDI; EDH error detection is used for SD-SDI.
- * 4 A Remote Control Unit sold separately is required.
 In this case, the compact flash memory card unit that comes standard and the Ethernet Unit that is sold separately cannot be used.

2.3 Specifications

2.3.1 Video Formats and Corresponding Standards

(1) Video Signal Standards

	Format Name	Standard Supported
1	1080i/60	
2	1080i/59.94	
3	1080i/50	
4	1080p/30	SMPTE 274M,292M
5	1080p/29.97	
6	1080p/25	
7	1080p/24	
8	1080p/23.98	
9	1080PsF/30	
10	1080PsF/29.97	
11	1080PsF/25	SMPTE RP211,292M
12	1080PsF/24	
13	1080PsF/23.98	
14	720p/60	
15	720p/50	SMPTE 296M
16	720p/59.94	
17	525i/59.94	SMDTE 250M
18	625i/50	

(2) Other Standards

Ancillary data standard

SMPTE 291M

- Embedded audio standard
 - HD-SDI SMPTE 299M
 - SD-SDI SMPTE 272M

(3)	Format Setting	
	SDI Signal	Select auto setting or manual setting
	Sampling Frequency	74.25 MHz (HDTV), 74.25/1.001 MHz (HDTV), 13.5
		MHz (SDTV)
	External Synchronization	Auto setting by HDTV tri-level sync signal or SDTV
		black burst.

2.3.2 Input/Output Connector

(1) SDI Input

,		
Input Connector	BNC connector	2 systems A and B
Input Impedance	75 Ω	
Input Return Loss	≥ 15 dB	5 MHz to serial clock frequency
Maximum input voltage	±2 V (DC + AC pe	eak)

(2)	External Reference Input		
	Input Signal	Tri-level sync signa	al or NTSC/PAL black burst
	Input Connector	BNC connector	1 system 2 connectors
	Input Impedance	15 KΩ	Passive Loop-through
	Input Return Loss	≥ 30 dB	
	Maximum input voltage	$\pm 5 V (DC + AC pea$	ak)
	 If the video signal wave reference, the waveforr 	m phase of 1 clock b	before and after the video signal
(3)	SDI Output	DNO I	
	Output Connector	BNC connector	1 connector
		Reclocks and outp	uts the selected SDI input signal
	Output Impedance	/5Ω	
	Output Voltage	800 mvp-p ±10 %	E MUT to corial cleak fragmanay
	Oulput Return Loss	≥ 10 UB	5 MHZ to serial clock frequency
(4)	Headphone Output		
	Output Signal	Separates and out	puts the embedded audio signal
		in the SDI signal	
	Sampling Frequency	Only supports 48 k	Hz (must be synchronized to the
		video signal)	
	Output Connector	Miniature Jack	1 connector (stereo)
	Volume Adjustment	Set from the menu	
		JZ 52 (10 52 10 000	52)
	Quantization Accuracy	SD-SDI: 24 bits	
		5D-5D1. 20 bits	
(5)	IF Slot		
	Installable Units	Select one unit from	m the following:
		Compact Flash Me	emory Card Unit
		Remote Control Un	lit (Sold Separately)
		Ethemet Unit (sold	separately)
(6)	Compact Flash Memory C	ard Unit	
	Function	Saves screen capt	ures, error logs, preset data, and
		data dumps	
	* A Microdrive cannot be	used.	
(7)	Remote Control Unit (Sold	l separately)	
	Function	Recalling of preset	s, monitoring of errors
	Control Signal	TTL level (LOW ac	tive)
	Control Connector	25-pin D-sub	1 connector (female)
* The compact flash memory card unit that comes standard and the Ethernet			
	Unit that is sold separately cannot be used.		

(8) Ethernet Unit (Sold separately)

Function	Remote control from an external computer and
	monitoring of errors, etc.
Туре	10BASE-T/100BASE-TX Auto switching

Input/Output connector RJ-45

Standard

Conforms to IEEE802.3

* The compact flash memory card unit that comes standard and the Remote Control Unit that is sold separately cannot be used.

1 connector

2.3.3 Display Format

Display Format	XGA	Effective area 1024 x 768 dots
Display	1-screen disp	blay
	Waveform display, au	n display, vectorscope display, picture Judio display, and status display
	2-screen disp	blay
	Waveform	display and vectorscope display
	Waveform	display and picture display
	Waveform	display and audio waveform display
Waveform d		ı display and audio level display
	4-screen disp	blay
	Select aud	dio waveform display or status display
	in additio display, ar	n to waveform display, vectorscope nd picture display.

2.3.4 Waveform Display

(1)	Waveform Operation	
	Display Mode	Overlay display: Displays component signals overlaid
		Parade display: Displays component signals side by side
		Timing display: Displays by calculating Y-C_B and Y-C_R
		Uses bowtie signals (authorized by Tektronix, Inc.)
	EAV-SAV Period	Select show or hide
	G, B, R Conversion	Converts Y, C_B , C_R signals into G, B, R and displays
		the result
	Pseudo-Composite Display	Artificially converts component signals into composite
		signals and displays the result
	Channel Assignment	Select GBR order or RGB order during GBR conversion
		display
	Scale	Select V scale or % scale
		You cannot select scales when pseudo-composite display is selected.
	Line Select	Displays the selected line

(2)	Vertical Axis		
	Gain	Select x1, x5, or variable	
	Variable gain	x0.2 to x10.0	
	Amplitude Accuracy	±0.5 %	
	Frequency Response HD	ΓV	
	Y signal	+0.5 %	1 MHz to 30 MHz
	C ₂ C ₂ Signal	±0.5 %	0.5 MHz to $15 MHz$
		±0.0 /0	
	Low-pass Allenuation	≥ 20 UD	al 20 MHz
	Frequency Response SD1	V	
	Y signal	±0.5 %	1 MHZ to 5.75 MHZ
	C _B C _R Signal	±0.5 %	0.5 MHz to 2.75 MHz
	Low-pass Attenuation	≥ 20 dB	at 3.8 MHz
(3)	Horizontal Axis		
(0)	Line Display	Display format	Overlay: 1H, 2H
			Parade: 1H 2H 3H
			Timing: 2H
		Magnification	Select v1 v10
		Active Dicplay	Displays only the video period
		Active Display	Displays only the video period
		Biarik Display	Displays blanking period
			expanded when set to overlay 2H
	Field Display	Display format	Overlay: 1V, 2V
			2V display not allowed for
			progressive
			Parade: 1V, 2V, 3V
		Magnification	Select x1, x20, x40
	Time Base Accuracy	±0.5 %	
(4)	Cursor Measurement		
()	Configuration	Horizontal cursors:	2 cursors (BEE and DELTA)
	Comgutation	Vertical cursors:	2 cursors (REE and DELTA)
	Amplitude Meesurement	Maggurad in [9/1 or	
		Disalaured in [%] an	
	Time Measurement	Displayed in [usec]	or [msec]
	Frequency Display	Displays the freque	ency in which the time between
		cursors is consider	ed a cycle.
235 V	Actorecone Disnlay		
2.5.5 V	Sonsitivity	Select 75 % or 100	% Using a color bar
	Coin	Select 75 % 01 100	A converiable
	Variable gain		IAG OF VARIABLE
	variable gain	x0.2 to x10.0	
	Amplitude Accuracy	±0.5 %	
	Blanking Period	Displayed by mask	ing
	IQ Axis	Select show/hide	
	Pseudo-Composite Display	Artificially converts of	component signals into composite
		signals and displays	s the result
	Line Select	Displays the select	ed line

2.3.6 Picture Display

	HDTV Display SDTV Display Frame Rate Marker Display	Displayed by sampling the pixels (when set to FIT) Displayed by interpolating pixels (when set to FIT) Converts the frame rate using the internal synchronization signal and displays the result Center marker 4:3 or 16:9 marker display Safe action marker display Safe title marker display Displays a marker the selected line
2.3.7 E	mbedded Audio Display	
2.0.7 L		
(1)	Quantization Accuracy	SDTV: 20 bits
(2)	Lissajous Display Display Channel Display Method	Select 2-ch or 8-ch display Select X-Y or L-R
(3)	Sound Image Display Display Channel	Select 3-1 ch, 3-2 ch, or 3-2-2 ch.
(4)	Level Meter Display Display Channel Meter	Simultaneous 8 ch display Select 60 dB peak level, 90 dB peak level, or average response meter Peak level meter has a hold function
(5)	Correlation Meter Display	Displays the phase between channels using a correlation meter
(6)	Channel Group Selection Ch Mapping	Select two arbitrary groups from groups 1, 2, 3, and 4 Mapping of L, R, SL(S), SR, C, LFE, RL, and RR is possible.
2.3.8 S	Status Display	

(1)	SDI Signal Status Display	
	Signal Detection	Detects the presence or absence of SDI signals
	CRC Error	Transmission error of HD-SDI signals
	EDH Error	Transmission error of SD-SDI signals
	BCH Error	Transmission error of embedded audio signals in
		the HD-SDI signal
	TRS Error	Detects TRS errors
	Checksum Error	Transmission error of ancillary data
	Parity Error	Detects parity errors in the ancillary data header

	Gamut Error	Detects gam	ut errors
	Detection Range	Upper limit:	90.0 % to 109.4 %
		Lower limit:	-7.2 % to +6.0 %
			0.1 % steps
	Composite Gamut Error	Monitors the	level error when the component signal
		is converted	into composite signal
	Detection Range	Upper limit:	90.0 % to 135.0 %
		Lower limit:	–40 % to –20 %
			0.1 % steps
	Equivalent Cable Length		
	Measurement	Displays the	signal attenuation of the SDI signal by
		converting to	cable length
	Supported cables	HD-SDI	Select L-7CHD, LS-5CFB, or 1694A
		SD-SDI	Select LS-5C2V, 8281, or 1505A
	Accuracy	±20 m	
	Resolution	5 m (10 m fo	r L-7CHD)
	Error Count	Up to 100,00	00 errors
		Counts only	the specified errors
	Count period	1 count even	if multiple errors occur within 1 second
(2)	Data Dump Display		
	Display format	Displayed s	eparately by serial data sequence or
		channel.	
	Line Select	Displays the	selected line
	Sample Select	Displays sele	ected samples
	Jump Function	Move to EAV	or SAV by one-key operation
	Data Output	Save data in	text format to a PC via a compact flash
		memory care	d or Ethernet network.
(3)	Event Log		
	Number of Logs	Up to 1,000	events
		Select over	write or update when the maximum
		number of ev	vents is exceeded
	Operation	Logs events	that occur from start to stop
	Logged Items	Error items, i	nput switching operation, time stamps,
		etc.	
	Data Output	Save data in	text format to a PC via a compact flash
		memory care	d or Ethernet network.
(4)	Audio Status		
()	Voice Control Packets	Analyzes an	d displays the voice control packets of
		the SDI sign	al
	Channel Status	Dumps, ana	lyzes, and displays the channel status
		of the embed	dded audio signal
(5)	EDH display		
(0)	EDH	Displays the	status of the EDH packets

2.3.9 Screen Capture

Capture	Captures the display screen
	Records 1 screen in the internal memory
Media	Internal memory (RAM) or compact flash card
Display	Displays the captured image by itself or by overlapping
	with the input signal
Data Output	Save data in B.M.P. format to a PC via a compact
	flash memory card or Ethernet network.

2.3.10 Presets

Number of Presets	30
Media	Internal memory or compact flash memory card
Recall Method	Through the front panel, remote connector, and
	Ethernet network
	Switch 8 points and 30 points for recalling through
	the remote connector
Сору	Copy presets collectively to the compact flash
	memory card or from the compact flash memory card
	to the LV 5750

2.3.11 Other Display Settings

(1)	LCD	
	Backlight brightness	Select high or low
	Auto Shutoff	Set the time for the backlight to shut off automatically
(2)	Display	
	Format	Displays the input signal format at the top section of the screen
		Select show or hide
	Color System	Displays YC_BC_R , GBR, RGB, or COMP depending on the setting
		Select show or hide
	Date	Displays the data using the built-in calendar function Select year/month/day, month/day/year, day/month/ year, or hide
	Time	Displays the time using the built-in time function or the ATC embedded in the SDI signal (frame count is not displayed) Select show or hide

(3) Illumination of Key LEDs Illumination of All Key LEDs Turns on all the LEDs corresponding to the keys

2.3.12 General Specifications

Environmental Conditions	
Operating Temperature	0 to 40 °C
Operating Humidity	≤ 85 % RH (without condensation)
Spec-Guaranteed Temperature	10 to 30 °C
Spec-Guaranteed Humidity	≤ 85 % RH (without condensation)
Operating Environment	Indoor and outdoor use
Operating Altitude	Up to 2,000 m
Overvoltage Category	Ι
Pollution Degree	2
Power Requirements	DC 12V (10 to 18V), 30 Wmax.
Dimensions and Weight	215 (W) x 133 (H) x 103 (D) mm
	(excluding projections)
	224 (W) x 147 (H) x 180 (D) mm
	(including projections)
	2.5 kg
Accessories	Instruction manual1

3. PANEL DESCRIPTION

3.1 Front Panel



Figure 3.1 Front Panel

1 POWER switch

Power switch. (The word "POWER" illuminates when the power can be turned ON.) If a battery is attached when the power is OFF, the LED blinks.

If a power supply is connected to the XLR connector, the LED stays lit (not blinking).

2 RECALL key

Displays a menu used to recall the settings saved to the preset memory.

③ MEMORY key

Stores the settings to the preset memory or sets the preset memory.

④ SYSTEM key

Displays the system menu. Enters settings related to the LV 5750 system.

⑤ CAPTURE key

Captures the screen. The capture image can also be saved.

6 REFERENCE key

Press this key to switch the synchronization of the video signal waveform display to INTERNAL (synchronization signal regenerated from the SDI signal) or EXTERNAL (external sync signal applied to the EXT REF connector 32).

- SDI signal input selection key Switches the input between INPUT A and INPUT B.
- B Hood attachment holes
 Holes for attaching the optional viewing hood sold separately.
- (9) V POS control

Adjusts the vertical position on the waveform display. Press the control to reset the vertical position of the waveform to default.

10 MODE key (MODE)

Switches the display mode of WFM waveforms. OVERLAY displays channels 1, 2, and 3 overlaid. PARADE displays channels 1, 2, and 3 side by side. TIMING is used when performing timing measurements such as for bowtie displays.

1 H POS control

Adjusts the horizontal position on the waveform display. Press the control to reset the horizontal position of the waveform to default.

12 CH1, CH2, and CH3 keys

Selects the channel from CH1, CH2, and CH3 to be shown on the waveform display. The keys can be turned ON/OFF independently; the LED of the selected key turns on. However, you cannot turn OFF all the channels simultaneously.

- Waveform key (WFM)Displays waveforms.
- VECTOR key (VECTOR) Displays vector waveforms.
- (5) PICTURE key (PICTURE) Displays pictures.
- 16 AUDIO key

Shows the lissajous display, level meter, and other displays of embedded audio.

- 1) STATUS key
 - Displays the status of the input signal.
- MULTI key Displays multiple items at once.
- 19 Function Dial (F·D)

Function dial for entering values and settings on each menu. Press the dial to reset the step (resolution) setting in various menus to its default.

20 SHORTCUT key

One of several functions can be assigned through user setting.

21 Headphone socket

Outputs the embedded audio signal in the SDI signal to the headphone.

The output channel and volume are set in chapter 9, "Audio Display."

All audio signals synchronize to the video clock of the SDI signal. If the signal is out of synchronization, noise will be generated.

22 Function keys (F·1 to F·7)

Function keys used to enter settings on the menu.

3.2 Rear Panel



Figure 3.2 Rear Panel

23 DC power input connector

An XLR DC power input connector. For details, see section 3.4, "Power Supply."

24 External IF slot

By inserting a dedicated interface card, various functions can be added such as saving/ loading data from the compact flash card, controlling the LV 5750 via the Ethernet interface, and recalling presets using the remote connector. The Compact Flash Card Unit is installed as standard.

Only one unit can be installed to the external IF slot.

* The number of times a unit can be inserted and removed is 500. If this number is exceeded, the unit may fail to operate properly due to poor contact or other reasons.

25 Unit attachment screw

A screw used when installing or removing a unit from the external IF slot 29.

26 SDI signal input connector

Input connector for serial digital signals. Two systems (A and B) are provided. The input is internally terminated at 75 Ω .

Do not apply a voltage exceeding ± 2 V (DC + peak AC) to the SDI signal input connector. If you do, the LV 5750 may malfunction.

2 SDI signal output connector (SDI OUTPUT)

HD-SDI or SD-SDI signal output. Retransmits the signal by reclocking the signal selected using the SDI signal input selection key (SDI) \bigcirc .

Use this signal by connecting to the picture monitor with an SDI input, for example.

Do not apply signals to the output connector. If you do, the instrument may malfunction.

Do not apply a voltage exceeding ± 5 V (DC + peak AC) to the external synchronization input connector.

If you do, the instrument may malfunction.

28 Serial number label

Instrument serial number. Provide the number on the label when contacting LEADER.

29 Fan

Exhaust fan for internal cooling. Be sure not to obstruct air circulation.

Using the instrument with the fan air circulation blocked for an extended time can cause a malfunction.

30 Battery plate

Attachment plate for V-mount batteries. For details, see section 3.4, "Power Supply."

31 Vent holes

Inlet vent holes for internal cooling. Be sure not to obstruct air circulation.

Using the instrument with the fan air circulation blocked for an extended time can cause a malfunction.

32 EXT REF connector

Accepts external sync signal. The input configuration is loop-through. Supports HD tri-level sync signal or NTSC/PAL black burst signal.

3.3 Top and Bottom Panels





Figure 3.4 Bottom panel

33 Hexagonal screws for removing the case

Screws for removing the case when rack mounting the instrument. Remove the case using a hexagonal socket wrench with 3 mm width. You can remove the case by unfastening the screws and shifting the case backward. If you are using the instrument by carrying it around different locations, be sure use the instrument with the cover attached.

34 Screw hole for tripod

A hole used to mount the instrument on a tripod.



Figure 3.5 Direction for removing the case (side panel)

3.4 Note

The hexagonal screws for removing the case are designed so that the screws cannot be inserted more than 5 mm from the inner chassis. If you are using the LV 5750 with the case removed, do not insert the hexagonal screws back into the screw holes. In addition, do not use these screw holes for purposes other than those specified by LEADER. Inserting a screw longer than 5 mm can damage the LV 5750.



3.5 Power Supply

The input voltage range is 10 V to 18 VDC.

If you connect both the power supply (battery) attached to the battery plate 30 and the DC power input connector, the power from the DC power input connector takes precedence.

If you connect a battery or the DC power supply, the internal microprocessor is set to standby condition even when the power switch is turned OFF. Consequently, a small amount of power is consumed. If you are not using the LV 5750 for an extended time, disconnect the battery and DC power supply.



Pin No.	Pin Name
1	GND
2	NC
3	NC
4	+12V

* Do not connect anything to NC.

The case is negative ground. When using DC input, use caution when connecting equipment that is positive ground. If you reverse the DC input connection or if a positive power supply touches the case, the internal circuit may break.

Use the DC power supply within the ratings. Otherwise, a malfunction can occur. The LV 5750 may not operate properly when a power supply outside the ratings is used.

3.6 Power Switch

The POWER switch 1 turns ON/OFF the power.

The power to the LV 5750 is turned OFF by holding down the switch for 1 second to prevent the power from being turned OFF inadvertently while the monitoring is in progress. You do not have to hold down the switch when turning ON the power.

When the power is turned ON, the POWER switch 1 LED illuminates and the operation of the LV 5750 starts.

If a battery is attached when the power is OFF, the LED blinks. If a power supply is connected to the XLR connector, the LED stays lit (not blinking).

If the battery runs out and the LV 5750 stops while the power is turned ON, the LV 5750 will start at the power-on condition when you install a new battery.

	switch LED (1) operation			
OFF	Standby mode. Only the internal microprocessor is running.			
Blinking	Standby mode. A battery is installed in the V-mount adapter.			
	The battery voltage is within the ratings.			
ON	Running. The line voltage is within the ratings and the LV 5750 is running.			
3.7 Connections to the Rear Panel Connectors

3.7.1 SDI INPUT Connector

The SDI INPUT connector B (INPUT SDI A, B) is terminated internally at 75 Ω . You do not need to connect a terminator. Use a cable with a characteristic impedance of 75 Ω for the connection.

Applying excessive input voltage to the connector can cause damage to the instrument.

Do not apply excessive voltage.

3.7.2 EXT REF Connector

The EXT REF connectors ③ are of loop-through configuration. Loop-through connectors are connected as shown in Figure 3.6.

Apply the input signal to either connector and terminate the other connector with a 75 Ω terminator or 75 Ω equipment. Make sure to terminate the end of the chain. Use a cable with a characteristic impedance of 75 Ω for the connection.



Figure 3.6 Loop-through input

3.7.3 SDI OUTPUT Connector

Use the SDI OUTPUT connector O by connecting it to a picture monitor that supports SDI input. The output impedance is 75 Ω . Thus, terminate the other end at 75 Ω .

Do not apply signals to the output connector. Doing so can lead to malfunction.

3.8 External IF Slot

There are three types of dedicated external interface units available for the external IF slot @.

The Compact Flash Card Unit is installed on the standard model. Ethernet Unit and Remote Control Unit are available as options.

Only one of the units above can be installed to the external IF slot. When installing an optional unit, remove the Compact Flash Card Unit.

To remove a unit, turn the unit attachment screws (25) using a driver or a similar tool and remove the unit carefully. To install a unit, check the direction and carefully insert the unit board along the slot guide.

When installing or removing a unit from the external IF slot, be sure to disconnect the DC power supply or battery (with no power supplied to the LV 5750).

In particular, installing or removing a unit with the power turned ON can cause a malfunction.

* The number of times a unit can be inserted and removed is 500.

If this number is exceeded, the unit may fail to operate properly due to poor contact or other reasons.

3.8.1 Compact Flash Card Unit

The Compact Flash Card Unit is a unit that is installed to the external IF slot 24 as standard. It can be used to save captured data, data dumps, event logs, preset data, etc.

Format the compact flash card to FAT16 before use. Other formats such as FAT32 and NTFS will not be detected.



Figure 3.7 CF CARD slot 35

3.8.2 Ethernet Unit

If you install the Ethernet Unit, the LV 5750 can be controlled remotely using a PC via the Ethernet network.

The Ethernet Unit is an option sold separately. Remove the Compact Flash Card Unit that comes standard and insert the Ethernet Unit in its place.



Figure 3.8 Ethernet Connector (ETHER) 36

3.8.3 Remote Control Unit

The Remote Control Unit can be used to recall preset information using remote control. The Remote Control Unit is an option sold separately.

Remove the Compact Flash Card Unit that comes standard and insert the Remote Control Unit in its place.



Figure 3.9 REMOTE Connector 3

4. MENU STRUCTURE

The structure of the menu assigned to the front panel keys is shown below. Underlined sections indicate initial settings.

4.1 Waveform Display Menu





- * 1 Menu shown during pseudo-composite display.
- * 2 The selectable items vary depending on the video signal format, display mode, and other parameters.
- * 3 Menu shown during G, B, R or R, G, B display.

4.2 Vectorscope Display Menu



* 1 Menu shown during pseudo-composite display.

4.3 Picture Display Menu



* 1 Varies depending on whether the input signal is HD-SDI or SD-SDI.

4.4 Audio Display Menu





- * 1 Varies depending on whether the display mode is sound image, single lissajous, or multi lissajous.
- * 2 Varies depending on the 1st GROUP and 2nd GROUP selections.

4.5 Multi Display Menu

















- * 1 Menu shown during pseudo-composite display.
- * 2 The selectable items vary depending on the video signal format, display mode, and other parameters.
- * 3 Menu shown during G, B, R or R, G, B display.
- * 4 Varies depending on whether the display mode is sound image, single lissajous, or multi lissajous.
- * 5 Varies depending on the 1st GROUP and 2nd GROUP selections.
- * 6 Menu shown when the Compact Flash Memory Card Unit that comes standard is inserted.

* 7 Menu shown when the Remote Control Unit that is sold separately is inserted. Items indicated by italics are common settings for MULTI WFM and MULTI VEC. They cannot be set individually.



4.6 Status Display Menu



4-16



- * 1 Menu shown when the Compact Flash Memory Card Unit that comes standard is inserted.
- * 2 Menu shown when the Remote Control Unit that is sold separately is inserted.

4.7 Capture Menu



- * 1 Menu shown when the Compact Flash Memory Card Unit that comes standard is inserted.
- * 2 Displayed when the cursor is on a recallable file (.BSX extension).

4.8 System Menu





* 1 Menu shown when the Remote Control Unit that is sold separately is inserted.

* 2 Menu shown when the Ethernet Unit that is sold separately is inserted.

4.9 Preset Menu



* 1 Menu shown when the Compact Flash Memory Card Unit that comes standard is inserted.

5. SYSTEM CONFIGURATION BEFORE WAVEFORM MONITORING

5.1 SDI Input Signal Selection

5.1.1 SDI Input Signal

• The SDI signal input connectors (INPUT SDI A, B) 26 are dedicated to serial digital video signals.

Do not input other types of signals such as analog video signals.

- <u>The signal amplitude is within 800 mVp-p±10 %.</u> Do not apply signals that exceed this amplitude.
- Below are the supported formats. Only component SDI signals are supported; Composite SDI signals are not supported.
- There are two methods for setting the format. One is to detect the format automatically, and the other is to set the format manually. For details, see section 5.4, "Video Signal Format Setting."
 - 1 1080i/60
 - 2 1080i/59.94
 - 3 1080i/50
 - 4 1080p/30
 - 5 1080p/29.97
 - 6 1080p/25
 - 7 1080p/24
 - 8 1080p/23.98
 - 9 1080PsF/30
 - 10 1080PsF/29.97
 - 11 1080PsF/25
 - 12 1080PsF/24
 - 13 1080PsF/23.98
 - 14 720p/60
 - 15 720p/59.94
 - 16 525i/59.94
 - 17 625i/50



Figure 5.1 SDI signal input connectors (INPUT SDI A,B)

5.1.2 SDI Input Selection

The SDI signal input selection key \bigcirc is switches the SDI signal input connectors (INPUT SDI A,B) B to select the channel corresponding to the SDI input signal to be monitored on the LV 5750.

The A and B indicators to the left of the SDI signal input selection key O illuminates according to the selected channel.

If you press the SDI signal input selection key \bigcirc when an error is being observed, a NO SIGNAL or ERROR message appears, and the error count is incremented. If you switch the input signal by pressing the SDI signal input selection key (SDI) \bigcirc , the PHONES jack output and various displays will not appear for a few seconds.

5.2 External Synchronization Signal Setting

5.2.1 External Sync Signals

- The EXT REF connector 32 supports HDTV analog tri-level sync signals and SDTV analog black burst signals.
- The format of the signal input to the EXT REF connector 3 is automatically determined.
- <u>The signal amplitude is within ±5 V.</u> Do not apply signals that exceed this voltage.
- When applying HDTV tri-level sync signals, input signals that have the same frame frequency and line frequency as HD-SDI signals.
- The video signal waveform on the reference of an external synchronize signal indicates that the waveform phase of approximately 1 clock is not decided.



Figure 5.2 EXT REF connector 32 (EXT REF)

- If the video signal waveform is displayed using an external sync signal as reference, the waveform phase of 1 clock before and after the video signal clock is indefinite.
- Below are black burst signal formats for which the waveform display can be synchronized.

525i/59.94 black burst signal 1080i/59.94 1080p/29.97 1080PsF/29.97 1080PsF/23.98 (10 field ID is required for black burst) 1080p/23.98 (10 field ID is required for black burst) 720p/59.94 525i/59.94 625i/50 black burst signal 1080i/50

1080p/25 1080PsF/25 625i/50 When an external sync signal is applied, the format of the sync signal is automatically detected.

If the 10 field ID is embedded and the SDI format is 1080PsF/23.98 or 1080p/ 23.98 when an NTSC black burst is applied, the 10 field ID is automatically detected.

5.2.2 External Sync Signal Switching

To display waveforms with respect to the external sync signal, press the REFERENCE key (6) to illuminate the EXT LED.



If the SDI signal and the external sync signal are not synchronized, the waveform display moves.

Figure 5.3 REFERENCE key 6

If you select INT using the REFERENCE key 6 , the waveform is displayed using the sync signal on the SDI signal.

The displays that are affected by switching INT and EXT using the REFERENCE key (6) are waveform display and vectorscope display.

For picture display, audio display, and data dump display, synchronization is achieved using the SDI sync signal regardless of the INT/EXT setting.

5.3 System Menu

Settings related to the operation of the entire system are configured from the system menu by pressing the SYSTEM key ④.



Press the SYSTEM key ④ to display the System Setup Menu shown below.

Figure 5.4 SYSTEM key ④

INTERFACE is a menu for options.

If the standard Compact Flash Memory Card Unit is installed, the interface setup menu does not appear.



Figure 5.5 System setup menu

Function Key	Description
F·1 FORMAT	Sets the video signal format
F·3 DISPLAY	Sets the display
F·4 INTERFACE	Sets the input/output interface
F·5 DATE & TIME	Sets the date/time
F·6 SHORTCUT KEY SET	Sets the shortcut key operation
F·7 SETUP INIT	Initializes settings

Table 5.1 System setup menu description

5.4 Video Signal Format Setting

To set the video signal format, press the $\fbox{F\cdot1}$ FORMAT key from the system setup menu.

The video signal format setup menu appears. Then, set the items as desired.



Figure 5.6 Video signal format setup menu

Function Key	Description
F·1 MODE	Selects auto/manual setting of the video
	signal format
F·2 i/PsF SELECT	Selects interlace and segment frame
F·3 SCANNING	Selects scanning when manual is selected
F·4 FIELD(FRAME)FREQ.	Selects the field/frame frequency when
	manual is selected
F·6 COMPOSIT FORMAT	Selects the format for pseudo-composite
	display

Table 5.2 Description of the video signal format setup menu

5.4.1 Auto/Manual Selection

You can select auto or manual setting for the SDI signal video format.

(1) Auto Selection of the Video Signal Format

[SYSTEM F-1 FORMAT F-1 MODE :AUTO]

From the video signal format setup menu, press F-1 MODE to select AUTO. The LV 5750 detects the format of the input SDI signal and automatically sets itself.

The detected video signal format is displayed at the upper left corner of the screen (the video signal format display must be turned ON as described in section 5.5.1, "Information Display").

* The LV 5750 cannot automatically distinguish HD-SDI interlace and segment frame.

See section 5.4.2, "Interlace and Segment Frame Selection" and set it manually. Below are video signal formats that the LV 5750 cannot automatically distinguish.

1080i/60 and 1080PsF/30 1080i/59.94 and 1080PsF/29.97 1080i/50 and 1080PsF/25

(2) Manual Selection of the Video Signal Format

[SYSTEM F-1 FORMAT F-1 MODE :MANUAL]

If you press the $\overline{F\cdot 1}$ MODE key to select MANUAL from the video signal format setup menu, you must set the video signal format manually according to the input SDI signal.

To do so, carry out the procedures described in sections 5.4.3, "Scanning Selection," and 5.4.4, "Field/Frame Frequency Selection" in order. If the scanning and field/frame frequency selections are not correct, disturbance will occur on the display.

When an SDI signal of the specified video signal format is received, the video signal format is displayed at the upper left corner of the screen (the video signal format display must be turned ON as described in section 5.5.1, "Information Display").

5.4.2 Interlace and Segment Frame Selection

 $\begin{bmatrix} SYSTEM & F\cdot 1 \\ FORMAT & F\cdot 2 \\ i/PsF SELECT : INTERLAC / SEG.FRM \end{bmatrix}$ Even when auto setting of the video signal format is used, 1080i (interlace) and 1080PsF (segment frame) cannot be distinguished automatically. You must manually select interlace or segment frame.

Press the $\boxed{F\cdot2}$ i/PSF SELECT key from the video signal format setup menu and select interlace or segment frame. This selection is possible only when AUTO is selected in section 5.4.1, "Auto/Manual Setting Selection." The menu does not appear when MANUAL is selected.

INTERLAC: Selects interlace. SEG.FRM: Selects segment frame.

5.4.3 Scanning Selection

[<u>SYSTEM</u> <u>F·1</u> FORMAT <u>F·3</u> SCANNING :1080i / 1080PsF / 1080p / 720p / 525i / 625i]

Manually selects the scanning of the video signal format (as a video signal format setting).

The scanning selection is possible only when MANUAL is selected in section 5.4.1, "Auto/Manual Setting Selection." The menu does not appear when AUTO is selected.

1080i: Selects HD-SDI with 1080 active interlaced lines.

1080PsF: Selects HD-SDI with 1080 active segment frames.

1080p: Selects HD-SDI with 1080 active progressive lines.

720p: Selects HD-SDI with 720 active progressive lines.

525i: Selects SD-SDI with 525 interlaced lines.

625i: Selects SD-SDI with 625 interlaced lines.

5.4.4 Field/Frame Frequency Selection

[SYSTEM]	F-1 FORMAT	F·4 FIELD FREQ.	:60 / 59.94 / 50]
[SYSTEM]	F·1 FORMAT	F·4 FRAME FREQ.	:30 / 29.97 / 25 / 24 / 23.98]
[SYSTEM]	F·1 FORMAT	F·4 FRAME FREQ.	:30 / 29.97 / 25 / 24 / 23.98]
SYSTEM	F-1 FORMAT	F·4 FRAME FREQ.	:60 / 59.94

Manually selects field frequency or frame frequency as a video signal format setting.

The selectable field/frame frequencies vary depending on the selection made in section 5.4.3, "Scanning Selection" (among the video signal formats that the LV 5750 supports).

The selected value is displayed as video signal format at the upper right corner of the screen (the video signal format display must be turned ON as described in section 5.5.1, "Information Display").

The field/frame frequency selection is possible only when MANUAL is selected in section 5.4.1, "Auto/Manual Setting Selection." The menu does not appear when AUTO is selected.

Below are the selectable ranges of the field/frame frequency with respect to the item selected in section 5.4.3, "Scanning Selection." The field frequency is fixed when 525i or 625i is selected.

		Field Frequency	Frame Frequency
ion	1080i	60 / 59.94 / 50	
ect	1080PsF		30 / 29.97 / 25 / 24 / 23.98
Sel	1080p		30 / 29.97 / 25 / 24 / 23.98
ing	720p		60 / 59.94
ann	525i	59.94	
Sc	625i	50	

Table 5.3 Field/Frame frequencies according to the scanning selection

5.4.5 Format Selection of the Pseudo-Composite Display

Press the $\boxed{F \cdot 6}$ COMPOSIT FORMAT key from the video signal format setup menu to make the selection.

NTSC: Artificially converts all video signal formats into NTSC composite signal.

- PAL: Artificially converts all video signal formats into PAL composite signal.
- AUTO: Artificially converts video signal formats with frame/field rate of 25 Hz or 50 Hz into PAL and all other video signal formats into NTSC composite signal.



Figure 5.7 NTSC and PAL when displaying pseudo-composite vector waveforms

5.5 Screen Display Settings

To set the screen display, press the $\boxed{F\cdot3}$ DISPLAY key from the system menu. The screen display setup menu appears. Then, set the items as desired.



Figure 5.8 Screen display setup menu

Function Key	Description
F-1 INFORMATION	Sets the information display at the upper of the screen
F·2 BACK LIGHT	Selects the brightness of the LCD backlight
F·3 AUTO OFF	Selects the auto shutoff function of the LCD backlight

Table 5.4 Description of the screen display setup menu

5.5.1 Information Display

The LV 5750 can display information such as the video signal format and date/time at the top section of the screen. You can use the menu to turn on or off the function.



Figure 5.9 Information Display

(1) Showing the Video Signal Format

[SYSTEM] F·3 DISPLAY F·1 INFORMATION F·1 FORMAT :ON/ OFF]

From the information display menu, press $\boxed{F \cdot 1}$ FORMAT to select whether to show the video signal format at the top section of the screen.

(2) Showing the Date

[SYSTEM F·3 DISPLAY F·1 INFORMATION F·2 DATE : Y/M/D / M/D/Y / D/M/Y / OFF]

From the information display menu, press $\boxed{F\cdot 2}$ DATE to select the format of the date to be displayed at the top section of the screen.

Y/M/D: Displayed in the order year, month, and day.

M/D/Y: Displayed in the order month, day, and year.

D/M/Y: Displayed in the order day, month, and year.

OFF: Does not show the date.

(3) Displaying the Time

REAL: Displays the current time (built-in realtime clock)

OFF: Does not show the time.

(4) Showing the Color System

 $\begin{bmatrix} SYSTEM & F\cdot3 & DISPLAY & F\cdot1 & INFORMATION & F\cdot4 & COLOR & :ON/OFF \end{bmatrix}$ From the information display menu, press $\hline F\cdot4 & COLOR & to select whether to show the color system at the top section of the screen.$

The color system is displayed as follows when ON is selected.

YCbCr: Displaying the waveform using Y, CB, CR signal

GBR: Converting to G, B, R signal and displaying the waveform

RGB: Converting to R, G, B signal and displaying the waveform

COMP: Converting to pseudo-composite signal and displaying the waveform

5.5.2 Backlight Switching

You can adjust the brightness of the LCD backlight and select auto shutoff.

(1) Adjusting the Brightness of the Backlight

(2) Auto Shutoff

[SYSTEM] $F\cdot3$ DISPLAY $F\cdot3$ AUTO OFF :OFF / 5 min / 30 min / 60 min] From the screen display menu, press $F\cdot3$ AUTO OFF to set the auto shutoff function of the LCD backlight.

OFF: Does not automatically shut off the backlight.

5 min: The backlight is shut off 5 minutes after pressing the last key.

30 min: The backlight is shut off 30 minutes after pressing the last key.

60 min: The backlight is shut off 60 minutes after pressing the last key.

To recover from the backlight off condition, press any key.

The backlight turns ON, and the screen returns to the original condition.

Note Allow the backlight to turn off and on only a few times per day. Repeating the action frequently shortens the backlight life.

5.6 Interface Setting

If the optional Remote Control Unit or Ethernet Unit is installed, you must enter the remote connector or Ethernet connector interface settings. Enter these settings by selecting $\boxed{F\cdot4}$ INTERFACE from the system menu.

If the standard Compact Flash Memory Card Unit is installed, the interface setup menu does not appear.



Figure 5.10 Interface setup menu

Function Key	Description	
F·1 REMOTE	Selects the preset recall from the remote connector on	
	the rear panel.	
F·2 ETHERNET	Sets the Ethernet parameters.	

 Table 5.5
 Description of the interface setup menu

5.6.1 Recalling Presets through the REMOTE Connector

[SYSTEM] F-4 INTERFACE F-1 REMOTE : BIT / BINARY]

If you install the Remote Control Unit that is sold separately, you will be able to recall preset information using remote control. The remote connector has 8 pins assigned for recalling presets.

Select BIT when using the 8 pins to recall up to 8 presets. Select BINARY when using the lower 5 pins to recall up to 30 presets using binary code. For details, see the "LV 5750-02 Instruction Manual".

5.6.2 Ethernet Settings

[SYSTEM] F·4 INTERFACE F·2 ETHERNET]

You can set the Ethernet Unit that is sold separately by pressing $\boxed{F\cdot 2}$ ETHERNET from the interface setup menu.

For details, see the "LV 5750-01 Instruction Manual".

5.7 Date and Time Settings

The LV 5750 has a calendar and time function. Thus, date/time can be displayed and error logs can be taken with time information. For details on displaying the date/time, see section 5.5, "Screen Display Settings." For details on error logging, see section 11.2, "Event Log".



Figure 5.11 Date and Time Setup Menu

(1) Adjusting the Year

[SYSTEM F·5 DATE & TIME F·1 YEAR]

You can set the year from 1900 to 2200. From the menu, select $F\cdot 1$ YEAR and then adjust the year using the function dial ($F\cdot D$) (19).

(2) Adjusting the Month

[SYSTEM]	F·5 DATE & TIME	F·2	MON	TH]	
		_		_	

You can set the month from January to December. From the menu, select F-2	
MONTH and then adjust the month using the function dial $(F \cdot D)$ (9.	

(3) Adjusting the Date

```
[SYSTEM F·5 DATE & TIME F·3 DAY]
```

You can set the date from 1st to 31st. From the menu, select $\overline{F\cdot3}$ DAY and then adjust the date using the function dial ($\overline{F\cdotD}$) (19).

(4) Adjusting the Hour

[SYSTEM	F·5 DATE & TIME	F·4 HOUR]
You can set the	e hour from 1 to 24. From	m the menu, select $F\cdot 4$ HOUR and then
adjust the hou	r using the function dia	I (F·D) 19.

(5) Adjusting the Minute

```
\begin{bmatrix} SYSTEM & F \cdot 5 \\ DATE & TIME & F \cdot 5 \\ MINUTE \end{bmatrix}You can set the minute from 0 to 59. From the menu, select F \cdot 5 MINUTE and then adjust the minute using the function dial (F \cdot D) 19.
```

(6) Adjusting the Second

[SYSTEM	F·5 DATE & TIME	F·6 SECOND]
You can set th	ne second from 0 to 59.	From the menu, select F·6 SECOND
and then adju	st the second using the	function dial $(F \cdot D)$ (19).

(7) Date/Time Confirmation

[SYSTEM] F·5 DATE & TIME F·7 CLOCK SET]

When you are done adjusting the date and time, press $\boxed{F \cdot 7}$ CLOCK SET to write to the calendar function of the LV 5750.

The date/time setup menu is closed as the date/time is written, and the screen returns to the system menu.

To cancel the writing, press the SYSTEM key ④.

5.8 Shortcut Key Setting

(1) Illumination of All Key LEDs

By assigning the key LED illumination function to the SHORTCUT key 20, you can illuminate all the key LEDs of the LV 5750 using the SHORTCUT key. When you perform the next operation, all keys other than the functioning key are turned OFF.

From the short cut key setup menu, press $\boxed{F \cdot 6}$ SHORTCUT KEY SET to select KEY LED. The shortcut key is assigned to the key LED illumination function.

(2) Direct Recall

By assigning the direct recall function to the SHORTCUT key 20, you can recall presettings using the SHORTCUT key.

Direct recall is useful when a setting that is used often but deep in the menu tree is assigned (event log screen or data dump display, for example).

From the short cut key setup menu, press $\boxed{F \cdot 6}$ SHORTCUT KEY SET to select DIRECT_K. The shortcut key is assigned to the direct recall function.

The preset procedure is as follows:

- Set the LV 5750 in a condition you wish to assign to the direct recall function.
- Press the MEMORY key 3 to display the preset memory menu.
- Press the SHORTCUT key 20. The key LED blinks twice, and the setting is registered.

You are done assigning the direct recall function.
(3) PHONES jack volume adjustment

By assigning the volume adjustment function to the SHORTCUT key (20), you will be able to adjust the PHONES jack volume using the function dial (F·D) (19) by pressing the SHORTCUT key regardless of the menu that is being displayed.

To assign the function, select VOLUME using the $\boxed{F \cdot 6}$ SHORTCUT KEY SET from the system menu. This will make the function dial ($\boxed{F \cdot D}$) (19) the PHONES jack volume adjustment dial when the SHORTCUT key is pressed. The volume adjustment function is cleared when another key is pressed. Use the AUDIO menu for normal volume adjustment.

5.9 Initialization

The LV 5750 stores the last operation before the power is turned off, and starts with the previous condition when the power is turned on the next time.

(1) Normal Initialization Procedure

[SYSTEM	F·7 SETUP INIT	F·1 INIT YES]				
[SYSTEM	F·7 SETUP INIT	F·3 INIT CANCEL]				
Press the SYSTEM key ④, F·7 SETUP INIT, and then F·1 INIT YES. Press						
F·3 INIT CANCEL to return to the system setup menu without initializing.						

This initialization procedure does not initialize the Ethernet settings such as the IP address or the presets.

(2) Hardware Reset

To clear the backup memory and clear all settings including the Ethernet settings and presets, you must carry out a hardware reset.

To carry out a hardware reset, turn ON the power while holding down the V POS control (9) and the H POS control (1).

When you carry out a hardware reset, a message appears stating that the backup memory has been cleared. Press any key. The initialization procedure terminates.

* The Ethernet Unit is sold separately.

6. VIDEO SIGNAL WAVEFORM DISPLAY

6.1 Video Signal Waveform Display

Press the WFM key 13 to display video signal waveforms.

The screen shows the video signal waveform, scale, and waveform display menu.

The waveform display menu consists of two pages. Use $\boxed{F \cdot 7}$ next menu or prev menu to switch between page 1 and 2.



Figure 6.1 Waveform display setup key (3)



(a) Page 1



(b) Page 2

Figure 6.2 Waveform Display and Waveform Display Menu

Function Key	Description
F·1 INTEN	Adjusts the intensity of the displayed waveform
F·2 GAIN FILTER	Sets the gain and low-pass filter
F·3 SWEEP	Sets the sweep time
F·4 LINE SELECT	Line select
F·5 CURSOR	Cursor measurement
F·6 COLOR SYSTEM	Sets the color system
F·1 SCALE	Sets the size and unit of the scale
F·2 EAV-SAV	Selects the waveform display of the blanking
	period
F·4 TIMING MODE	Selects the timing display

Table 6.1 Description of the waveform display men	Table 6.1	Description	of the	waveform	display	menu
---	-----------	-------------	--------	----------	---------	------

6.2 Display Channel Settings

Use the CH1, CH2, and CH3 keys 12 to select the waveform you wish to display. Signals assigned to CH1, CH2, and CH3 keys 12 depend on whether Y, C_B, C_R display, G, B, R display, or R, G, B display is used.

For details, see section 6.10.1, "Color Matrix Conversion."

By default, CH1, CH2, and CH3 are assigned to Y, C_B, and C_R, respectively.

Press the CH1, CH2, or CH3 key (2) to illuminate the corresponding LED and display the waveform assigned to that key. Press the key again to turn the LED off and clear the waveform display.

However, you cannot turn off all the channels.

If you select TIMING display using the MODE key (10), the conditions of the CH1, CH2, and CH3 keys (12) are ignored, and timing display is activated. In addition, the CH1, CH2, and CH3 keys (12) do not affect displays other than waveform displays such as vectorscope display and picture display.

The CH1, CH2, CH3 keys (2) do not function when the pseudo-composite display is selected as described in section 6.10.1, "Color Matrix Conversion."

6.3 Display Mode Settings

Press the MODE key 10 to switch the waveform display format between overlay, parade, and timing (time difference measurement).

If the timing display is prohibited (PASS) as described in section 6.13, "Mode Key Switching," the timing display does not appear even when you press the MODE key ⁽¹⁾.

If the pseudo-composite display is selected as described in section 6.10.1, "Color Matrix Conversion," the display does not change even when you press the MODE key 10.

6.3.1 Overlay Display

Select OVERLAY using the MODE key 10 to display the waveforms of CH1, CH2, and CH3 overlaid.



Figure 6.3 Overlay Display

6.3.2 Parade Display

Select PARADE using the MODE key 10 to display the waveforms of CH1, CH2, and CH3 side by side.

The waveforms are displayed in the order CH1, CH2, and CH3 from the left. In addition, if you select PARADE, the screen is automatically set to 1H or 1V display even if 2H display or 2V display is selected as described in section 6.7, "Sweep Settings." 2H display or 2V display is not possible in parade display.



Figure 6.4 Parade Display

6.3.3 Timing Display

Select TIMING using the MODE key 0 to measure the time and amplitude differences between channels with respect to CH1 (Y/G). A bowtie signal is used as a signal source.

Timing display shows two bowtie displays side by side as shown in Figure 6.5. The conditions of CH1, CH2, and CH3 2 are ignored.

If the timing display is prohibited as described in section 6.13, "Mode Key Switching," the timing display does not appear even when you press the MODE key ⁽¹⁾. By default, the timing display is prohibited.

* Bowtie signal waveform measurement

If you set the LV 5750 to TIMING display, CH1 (Y/G) and CH2 (C_B/B) are displayed on the left and CH1 (Y/G) and CH3 (C_B/R) are displayed on the right. The vertical lines of bowtie signals are marker signals. The long marker at the center is the reference marker (no time difference).

The interval between markers represents a timing difference of 1 ns (when Leader's LT 443D is used). The timing difference is read from the interval between the null and the reference marker. If the null is positioned to the left of the reference marker, this indicates that C_B/B or C_R/R is advanced relative to Y/G.

If there is an amplitude difference between channels, the depth of the null is decreased.



(Bowtie signal: use authorized by Tektronix, Inc.)

Figure 6.5 Timing Display

6.4 Position Adjustment

You can adjust the vertical position of the video signal waveform on the screen using the V POS control (9) and the horizontal position using the H POS control (11).

The V POS control 9 and H POS control 1 also function as switches. Press the switch to reset the display position of the video signal waveform to the reference position.

6.5 Brightness Adjustment

To adjust the brightness of the video signal waveform or the scale, press the WFM key 3 to display the waveform display menu. Then, select $\boxed{F \cdot 1}$ INTEN. The brightness adjustment menu appears. Set the items as desired.



Figure 6.6 Brightness Adjustment Menu

6.5.1 Brightness Adjustment of the Video Signal Waveform

[WFM F-1 INTEN F-1 WFM F-D]

To adjust the intensity (brightness) of the video signal waveform, press $\boxed{F \cdot 1}$ INTEN from the brightness adjustment menu.

While the $\overline{F\cdot 1}$ WFM section is selected in blue, turn the function dial ($\overline{F\cdot D}$) (9 to adjust the brightness between -128 and 127.

The function dial ($\overline{F \cdot D}$) (19) also functions as a switch. Press the switch to reset the brightness of the video signal waveform to the reference value (0).

6.5.2 Scale Brightness Adjustment

[WFM F-1 INTEN F-2 SCALE INTEN F-D]

To adjust the brightness of the scale on the video signal waveform display, press F:2 SCALE INTEN from the brightness adjustment menu and turn the function dial $(\overline{F\cdot D})$ 19.

The adjustment range of the scale brightness is -8 to 7 (16 levels).

The function dial ($\overline{F \cdot D}$) (19) also functions as a switch. Press the switch to reset the brightness of the scale on the video signal waveform display to the reference value (4).

6.6 Gain/Filter Adjustment

To adjust the gain (amplitude) of the video signal waveform or set the filter, press the WFM key 3 to display the waveform display menu. Then, select $\fbox{F2}$ GAIN FILTER. The gain/filter adjustment menu appears. Set the items as desired.



Figure 6.7 Gain/Filter Adjustment Menu

6.6.1 Gain Adjustment

[WFM F-2 GAIN FILTER F-1 GAIN VAR :CAL / VAR]

To continuously change the gain of the video signal waveform, press $F\cdot 1$ GAIN VAR from the gain/filter adjustment menu. Turn the function dial ($F\cdot D$) (9) with VAR selected to change the gain continuously. Select CAL to set the gain to the reference value.

The gain adjustment range is x0.20 to x2.00 with respect to the reference gain.

- CAL: Reference gain
- VAR: Continuous change of gain (variable)

The function dial $(\overline{F \cdot D})$ (19) also functions as a switch. Press the switch while VAR is selected to reset the gain to the reference value.

The gain reference is set to the value selected in section 6.6.2, "Gain Selection."



Figure 6.8 Variable Gain Display

6.6.2 Gain Selection

 $\begin{bmatrix} WFM & F\cdot 2 \\ GAIN FILTER & F\cdot 2 \\ GAIN MAG & X1 / X5 \end{bmatrix}$ To change the gain of the video signal waveform to 1x or 5x, press F·2 MAG from the gain/filter adjustment menu. Select X5 to set the gain to 5x. Select X1 to set the gain to the reference value (1x).

X1: 1x gain

X5: 5x gain

6.6.3 Filter Selection

(1) Filter Selection on the Component Display

 $\begin{bmatrix} WFM & F\cdot 2 & GAIN FILTER & F\cdot 3 & FILTER & FLAT / LOW PASS \end{bmatrix}$ To apply a low-pass filter on the component waveform display such as Y, C_B, C_R or G, B, R, press the $F\cdot 3$ FILTER key from the gain/filter adjustment menu. Select LOW PASS to apply a low-pass filter to the video signal waveform display. Select FLAT to produce a flat frequency response within the frequency range. The low-pass filter response is as shown in Table 6.2.

FLAT: Flat frequency response within the signal frequency range LOW PASS: Low-pass filter response with the frequency response shown in the table below

	Frequency Range	Low-Pass Filter Response
HDTV	30 MHz	Attenuation of 20 dB or more at 20 MHz
SDTV	5.75 MHz	Attenuation of 20 dB or more at 3.8 MHz



Table 6.2 Filter Response

Figure 6.9 Low-pass Filter Waveform

(2) Filter Selection on the Pseudo-Component Display

 $\begin{bmatrix} WFM & F\cdot2 & GAIN FILTER & F\cdot3 & FILTER & FLAT / FLAT+LUM \end{bmatrix}$ If the pseudo-composite display is selected as described in section 6.10.1, "Color Matrix Conversion," the pseudo-composite waveform and luminosity component waveform can be displayed side by side. Press the $F\cdot3$ FILTER from the gain/filter adjustment menu and select FLAT to display only the pseudocomposite waveform. Select FLAT+LUM to display the composite waveform and luminosity component using parade display.



Figure 6.10 Filter Waveform of the Pseudo-composite Display

6.7 Sweep Settings

To set a sweep on the video signal waveform, select $\boxed{F \cdot 1}$ SWEEP from the waveform display menu that appears when you press the WFM key 3.

The sweep setup menu appears. Select the appropriate items. The contents of the sweep setup menu vary depending on the video signal format.



Figure 6.11 Sweep Setup Menu

6.7.1 Sweep Selection

[WFM] F·3 SWEEP F·1 SWEEP : H / V]

To switch the line display or field display (for interlace format) or frame display (for progressive format) on the video signal waveform display, press $\boxed{F \cdot 1}$ SWEEP from the sweep setup menu.

- H: Line display
- V: Field display for interlace and segment frame Frame display for progressive

6.7.2 H Sweep Selection

[WFM F·3 SWEEP F·2 H_SWEEP : 1H / 2H]

If H (line display) is selected in section 6.7.1, "Sweep Selection," you can select 1H display, which sets the sweep time to 1 line, or 2H display, which sets the sweep time to 2 lines. However, 2H display is valid only when OVERLAY is selected using the MODE key ⁽¹⁰⁾. If PARADE is selected, the display is set to 1H. In addition, the sweep setting for bowtie signal monitoring is enabled if TIMING is selected using the MODE key ⁽¹⁰⁾ regardless of the H sweep selection. To switch the 1H/2H display, press $\boxed{F \cdot 2}$ H_SWEEP from the sweep setup menu.

- 1H: 1 line display
- 2H: 2 line display (valid only when OVERLAY is selected as described in section 6.3, "Display Mode Settings")



Figure 6.12 H Sweep Selection

6.7.3 V Sweep Selection

[WFM F·3 SWEEP F·2 V_SWEEP : 1V / 2V]

If V (field/frame display) is selected in section 6.7.1, "Sweep Selection," you can select 1V display, which sets the sweep time to 1 field, or 2V display, which sets the sweep time to 2 fields.

The V sweep selection is displayed only when the video signal format is interlace or segment frame; it is not displayed for progressive format.

In addition, 2V display is valid only when OVERLAY is selected using the MODE key 10. If PARADE is selected, the display is set to 1V. The sweep setting for bowtie signal monitoring is enabled if TIMING is selected using the MODE key 10 regardless of the V sweep selection.

To switch the 1V/2V display, press $\boxed{F\cdot 2}$ V_SWEEP from the sweep setup menu.

- 1V: 1 field display
- 2V: 2 field display (valid only when OVERLAY is selected as described in section 6.3, "Display Mode Settings")



(a) 1V Display





6.7.4 Field Selection

[WFM] F·3] SWEEP F·3] FIELD :FIELD1 / FIELD2] You can select the displayed field, if 1V is selected in section 6.7.3, "V Sweep Selection."

From the sweep setup menu, press $\boxed{F\cdot3}$ FIELD to select FIELD 1 or FIELD 2.

FIELD1: Field 1 is displayed.

FIELD2: Field 2 is displayed.

You cannot select the field if the video format is progressive.

6.7.5 Sweep Magnification

[WFM] F·3 SWEEP F·4 SWEEP MAG :X1 / X10 / X20 / X40 / ACTIVE / BLANK]

To expand the horizontal axis of the video signal waveform display, press $\boxed{F\cdot 4}$ SWEEP MAG from the sweep setup menu and select sweep magnification.

You can select the following magnification settings when H is selected as described in section 6.7.1, "Sweep Selection."

- x1: Turns off sweep magnification.
- x10: Magnifies the sweep 10x.
- x20: Magnifies the sweep 20x.

ACTIVE: Displays only the active picture area.

BLANK: Displays the line blanking period without masking.

You can select the following magnification settings when OVERLAY is selected as described in section 6.3, "Display Mode Settings" and 2H is selected as described in section 6.7.2, "H Sweep Selection."

- x1: Turns off sweep magnification.
- x10: Magnifies the sweep 10x.
- x20: Magnifies the sweep 20x.

BLANK: Displays the line blanking period.

If BLANK is selected, the waveform of the data in the blanking period is displayed regardless of whether EAV-SAV is to PASS/REMOVE as described in section 6.12, "Blanking Display Settings."

You can select the following sweep magnification settings when V Sweep is selected in section 6.7.1, "Sweep Selection."

- x1: Turns off sweep magnification.
- x20: Magnifies the sweep 20x.
- x40: Magnifies the sweep 40x.

Sweep	Sweep Magnification							
Selection	X1	X1 X10 X20 X40 ACTIVE BLANK						
1H	Yes Yes Yes No		Yes Yes					
2H	Yes	Yes	Yes	No	No	Yes		
1V	Yes	No	Yes	Yes	No	No		
2V	Yes	No	Yes	Yes	No	No		

Yes indicates that it is selectable; No indicates that it is not.

However, 2V display in the progressive format cannot be selected. Table 6.3 Relationship between Sweep Selection and Sweep Magnification

If the pseudo-composite display is selected as described in section 6.10.1, "Color Matrix Conversion," magnification settings of x10 and BLANK are not possible on 1H display and 2H display.

6.8 Line Selector

To select a line on the video signal waveform to be displayed, press the WFM key (3) to display the waveform display menu. Then, select $\overline{F \cdot 4}$ LINE SELECT. The line select menu appears. Set the items as desired.



Figure 6.14 Line Select Menu

6.8.1 Line Select

 $\begin{bmatrix} WFM & F\cdot4 \\ LINE SELECT & F\cdot1 \\ LINE SELECT : ON / OFF \end{bmatrix}$ To enable the line select display on the video signal waveform display, press the F·1 LINE SELECT key from the line select menu and select ON. Then, turn the function dial (F·D) (19) to display an arbitrary line. Select OFF to disable the line select display.

If line select is turned ON when OVERLAY is selected in section 6.3, "Display Mode Settings" and 2H is selected in section 6.7.2, "H Sweep Selection," 2-line line select is enabled. In this case, the selected line is shown on the left, and the next line is shown on the right.

6.8.2 Field Selection

[WFM F·4 LINE SELECT F·2 FIELD :FIELD1 / FIELD2 / FRAME] When the line select display is enabled on the video signal waveform display, you can set the variable range of the function dial ($\overline{F \cdot D}$) (19) to field or frame. You cannot select the field if the video signal format is progressive.

- FIELD1: Limits the range of lines that can be selected using the function dial $(F \cdot D)$ (9) to field 1.
- FIELD2: Limits the range of lines that can be selected using the function dial $(F \cdot D)$ (9) to field 2.
- FRAME: Sets the range of lines that can be selected using the function dial $(\overline{F \cdot D})$ (9) to the entire frame.

6.9 Cursor Measurement

Cursor measurement is used to measure voltage, time, and other parameters. The REF and Δ cursors are used to measure the voltage and time between two points on a waveform.

To perform cursor measurements, press the WFM key 3 to display the waveform display menu. Then, select $\boxed{F \cdot 5}$ CURSOR.

The cursor measurement menu appears. Set the items as desired.

If the pseudo-composite display is selected as described in section 6.10.1, "Color Matrix Conversion," cursor measurements on the time axis is not possible.



Figure 6.15 Cursor Measurement Menu

Function Key	Description
F·1 CURSOR	Turns ON/OFF the cursor display
F·2 XY SEL	Switches between the X-axis and Y-axis of the cursor.
F·3 X(Y) UNIT	Selects the cursor unit of the X-axis and Y-axis
F·4 FD VAR	Selects REF, DELTA, and TRACK using the function dial
F·5 REF SET	Sets the reference value for the relative measurement
F·D	Moves the cursor

Table 6.4 Cursor measurement menu description

6.9.1 Cursor Display

[WFM F·5 CURSOR F·1 CURSOR:ON / OFF]

To display the cursor, press $\boxed{F \cdot 1}$ CURSOR from the cursor measurement menu and select ON.

Select OFF to clear the cursor.

6.9.2 X-Axis/Y-Axis Cursor Selection

[WFM F·5 CURSOR F·2 XY SEL :X / Y]

Select the axis to perform the cursor measurement from X-axis (time) and Y-axis (amplitude).

If the pseudo-composite display is selected as described in section 6.10.1, "Color Matrix Conversion," you cannot select the X-axis (time). Measurement is made only on the Y-axis (amplitude).

- X: Measures the time using the cursor.
- Y: Measures the amplitude using the cursor.

6.9.3 Cursor Measurement Unit Selection

(1) Selecting the X-Axis Cursor Unit

[WFM F·5 CURSOR F·3 X UNIT :sec / Hz]

You can set the X-axis cursor unit using the $\boxed{F\cdot3}$ X UNIT key when X is selected as described in section 6.9.2, "X-Axis/Y-Axis Cursor Selection."

- sec: Displays using units of seconds.
- Hz: Displays using units of frequency where the interval between the cursors is taken to be one period.

If the pseudo-composite display is selected as described in section 6.10.1, "Color Matrix Conversion," cursor measurements on the X-axis is not possible.

(2) Selecting the Y-Axis Cursor Unit

[WFM F·5 CURSOR F·3 Y UNIT :mV / % / R%]

You can set the Y-axis cursor unit using the $\boxed{F:3}$ Y UNIT key when Y is selected as described in section 6.9.2, "X-Axis/Y-Axis Cursor Selection."

- mV: Displays using units of millivolts.
- %: Displays using a ratio where 700 mV is taken to be 100 % during component display.

Displays using a ratio where 714 mV is taken to be 100 % during pseudo-composite display (NTSC).

R%: Displays the relative value with respect to the reference amplitude using a ratio.

The reference amplitude is specified using the procedure given in section 6.9.5, "Reference Amplitude Setting."

6.9.4 Cursor Movement

 $\begin{bmatrix} WFM & F\cdot5 \\ CURSOR & F\cdot4 \\ FD \\ VAR:REF / DELTA / TRACK \end{bmatrix}$ To move the desired cursor, select $F\cdot4$ FD VAR from the cursor measurement menu.

- REF: Moves the reference cursor using the function dial $(F \cdot D)$ (19).
- DELTA: Moves the Δ cursor using the function dial (F·D) (19.
- TRACK: Moves the reference cursor and Δ cursor simultaneously using the function dial ($\overline{F \cdot D}$) (19.

6.9.5 Reference Amplitude Setting

[WFM F·5] CURSOR F·5] REF SET]

In cursor measurement, you can make an arbitrary amplitude a reference and display the relative value using a ratio.

Relative measurements can be carried out according to the procedure below.

- ① Select R% as described in section 6.9.3 (2), "Selecting the Y-Axis Cursor Unit."
- ② Set the cursors to the reference amplitude as described in section 6.9.4, "Cursor Movement."
- ③ Press F·5 REF SET to confirm the reference amplitude.
- ④ Measure the amplitude as described in section 6.9.4, "Cursor Movement."

6.10 Color System Setting

The SDI signals that can be monitored on the LV 5750 are limited to component signals consisting of intensity and chrominance difference signals.

Intensity and chrominance difference signals are usually represented by Y, C_B , C_R . The LV 5750 also provides a functionality that computes and displays Y, C_B , C_R signals as G, B, R signals. Furthermore, component signals can be artificially converted into NTSC or PAL signals to be displayed.

In addition, G, B, R signals can be displayed as R, G, B signals (in the order red, green, and blue). The waveforms corresponding to CH1, CH2, and CH3 keys 2 vary according to these settings.

To set the color system, press the WFM key 3 to display the waveform display menu. Then, select $\boxed{\texttt{F}{\cdot}\texttt{6}}$ COLOR SYSTEM.

The color system setup menu appears. Set the items as desired.



Figure 6.16 Color System Setup Menu

6.10.1 Color Matrix Conversion

[WFM F·6 COLOR SYSTEM F·1 COLOR MATRIX:YCbCr / GBR / RGB / COMPOSIT]

To select the color matrix of the video signal waveform display, press $\boxed{F \cdot 1}$ COLOR MATRIX from the color system setup menu.

YCbCr: Waveforms are displayed using intensity and chrominance difference signals. CH1, CH2, and CH3 keys ⁽¹⁾/₂ are assigned as follows: CH1:Y CH2:C_B CH3:C_R
GBR: Waveforms are displayed by converting the Y, C_B, C_R signal into G, B, R. CH1, CH2, and CH3 keys ⁽¹⁾/₂ are assigned as follows: CH1:G CH2:B CH3:R
RGB: Waveforms are displayed by converting the Y, C_B, C_R signal into R, G, B.

CH1, CH2, and CH3 keys 12 are assigned as follows: CH1:R CH2:G CH3:B

COMPOSIT: The component signal is artificially converted into a pseudocomposite signal such as NTSC or PAL and displayed. CH1, CH2, and CH3 keys (12) do not function.







Figure 6.17 Waveform Display by Color Matrix

6.10.2 Y-GBR (R, G, B) Display

WFMF·6COLOR SYSTEMWFMF·6COLOR SYSTEM

F·2 YGBR :ON / OFF]

I F·2 YRGB :ON / OFF]

To display the intensity signal (Y) and the G, B, R or R, G, B signal that has been computed through matrix conversion simultaneously, press $\boxed{F\cdot2}$ YGBR (YRGB) from the color matrix setup menu to select ON. In this case, the color matrix conversion as described in section 6.10.1, "Color Matrix Conversion," must be set to G, B, R or R, G, B. Y-GBR display is activated if G, B, R is selected; Y-RGB display is activated if R, G, B is selected. This selection is not possible if Y, C_B, C_R or pseudo-composite display is selected.

The Y-GBR display and Y-RGB display are shown only on a single screen of the video signal waveform display; it cannot be shown on the multi screen display. When Y-GBR display or Y-RGB display is selected, the CH1, CH2, and CH3 keys 1 are disabled.





(a) Y-GBR Display





6.10.3 Setup Selection

 $\begin{bmatrix} WFM & F \cdot 6 \\ COLOR SYSTEM & F \cdot 3 \\ SETUP :0\% / 7.5\% \end{bmatrix}$ If you select NTSC pseudo-composite display as described in section 6.10.1, "Color Matrix Conversion," you can select the setup level of the waveform display. From the color system setup menu, press F \cdot 3 SETUP. If you select 0 %, setup is not added. If you select 7.5 %, a pseudo-composite waveform display with 7.5 % setup added is shown.

You can select setup only when using the NTSC pseudo-composite display. The menu does not appear during component display or PAL display.







Figure 6.19 Setup Display

6.10.4 Pseudo-Composite Display

[WFM] F·6 COLOR SYSTEM F·1 COLOR MATRIX:YCbCr / GBR / RGB / COMPOSIT]

Only component SDI signals can be input to the LV 5750. However, by selecting COMPOSIT as described in section 6.10.1, "Color Matrix Conversion," you can display component signals as pseudo-composite signals.

If AUTO is selected as described in section 5.4.5, "Format Selection of the Pseudo-Composite Display," video signal format having a field frequency is 50 Hz or a frame frequency is 25 Hz or 50 Hz is converted into PAL. Video signal formats of other field/frame frequencies are converted into NTSC.

In the pseudo-composite display, the color burst frequency does not match the NTSC or PAL frequency; it is 1/4th the frequency of the input signal clock. In addition, when HD-SDI signals are displayed as pseudo-composite signals, the width and position of the sync signal and color burst signal are slightly different from those of the actual NTSC or PAL signals.

The line selector and V sweep are displayed using the number of lines of the original component signal.

In addition, the following limitations exist in the pseudo-composite display.

- ① The time axis cannot be expanded using sweep 1H and 2H.
- 2 The time axis cannot be measured in cursor measurement.
- ③ For composite display of HD-SDI, the line select is set to that of HDTV.
- ④ Waveform display of the EAV-SAV (blanking period) is not possible.
- (5) The MODE key, and CH1, CH2, and CH3 keys do not function.
- ⑥ The scale is set to % (IRE) scale during NTSC display and V (voltage) scale during PAL display.

6.11 Scale Settings

The LV 5750 provides various scale settings on the video signal waveform display. To set the scale, press the WFM key 3 to display the waveform display menu. Then, press $\fbox{F:7}$ next menu followed by $\fbox{F:1}$ SCALE.

The scale setup menu appears. Set the items as desired.



Figure 6.20 Scale Setup Menu

6.11.1 Scale Unit Selection

[WFM F·7] next menu F·1] SCALE F·1] SCALE UNIT :HDV,SD% / HDV,SDV / HD%,SD%]

On the video signal waveform display, you can set the scale unit to match HDTV and SDTV separately.

From the scale setup menu, press $\boxed{F\cdot 1}$ SCALE UNIT and select the scale unit of HDTV and SDTV.

HDV,SD%: V (voltage) unit for HDTV format and % unit for SDTV. HDV,SDV: V (voltage) unit for HDTV and SDTV formats. HD%,SD%: % unit for HDTV and SDTV formats.

The scale unit during pseudo-composite display is fixed to % (IRE) for NTSC display and V (voltage) for PAL display.



Figure 6.21 Scale unit selection

6.11.2 Display Selection of the 75 % Color Bar Scale

[WFM] $F\cdot7$ next menu $F\cdot1$ SCALE $F\cdot2$ 75% COLOR SCALE :ON / OFF] You can select a scale to match the peak level of the chrominance difference signal when the 75 % color bar is displayed as a waveform.

From the scale setup menu, press $\overline{F2}$ 75% COLOR SCALE. Select ON to show the scale, OFF to hide the scale.

The scale for 75 % color bar cannot be displayed during pseudo-composite display or G, B, R or R, G, B display.



Figure 6.22 75 % Color Bar Scale

6.11.3 Scale Color Selection

[WFM F·7] next menu F·1] SCALE F·3] SCALE COLOR: WHITE / YELLOW / CYAN / GREEN / MAGENTA / RED / BLUE]

You can select the scale color on the video signal waveform display. You can select the scale color from the seven colors below.

From the scale setup menu, press $\boxed{F\cdot3}$ SCALE COLOR to select the color.

WHITE: Sets the scale color to white.
YELLOW: Sets the scale color to yellow.
CYAN: Sets the scale color to cyan.
GREEN: Sets the scale color to green.
MAGENTA: Sets the scale color to magenta.
RED: Sets the scale color to red.
BLUE: Sets the scale color to blue.

6.12 Blanking Display Settings

[WFM F·7 next menu F·2 EAV-SAV :REMOVE / PASS]

You can select whether to display waveforms or mask to the black level during the blanking period on the video signal waveform display.

Press the WFM key (13) to display the waveform display menu. Then, press $\overline{F \cdot 7}$ next menu followed by $\overline{F \cdot 2}$ EAV-SAV to select REMOVE or PASS.

If the pseudo-composite display is selected as described in section 6.10.1, "Color Matrix Conversion," the waveform of the blanking period cannot be displayed.

REMOVE: Signals are masked to the black level during the blanking period. PASS: All waveforms are displayed during the blanking period.



Figure 6.23 Waveform Display during the Blanking Period

6.13 Mode Key Switching

[WFM] $F \cdot 7$ next menu $F \cdot 4$ TIMING MODE :NORMAL / PASS] You can select whether to allow the timing display to be selected when the MODE key (10) is pressed.

Press the WFM key 3 to display the waveform display menu. Then, press $\boxed{F \cdot 7}$ next menu followed by $\boxed{F \cdot 4}$ TIMING.

NORMAL: The timing display is also shown when the MODE key is pressed.

PASS: The timing display is not shown even when the MODE key is pressed.

Only overlay and parade displays are selected.

7. VECTORSCOPE DISPLAY

7.1 Vector Waveform Display

To show vector waveforms, press the VECTOR key 14.

The screen shows the vector waveforms, scale, and the vectorscope display menu.

The vector display of the component signal is achieved using X-Y display of C_B (horizontal) and C_R (vertical) signals.



Figure 7.1 Area around the VECTOR key 1



Figure 7.2 Vectorscope display and vectorscope display menu

Function Key	Description
F·1 INTEN/SCALE	Adjusts the intensity of the displayed waveform
	or sets the scale
F·2 GAIN	Sets the gain
F·3 LINE SELECT	Line select
F·4 COLOR SYSTEM	Sets the composite display and the color bar
	saturation

Table 7.1 Vectorscope display menu description

Graticule	Video Signal Format Setting	Applicable Calorimetry Standard		
1	1080i/60,50 1080p/30,25,24	SMPTE 274M, 296M		
	1080PsF/30,25,24			
2	525i/60,625i/50 720p/60	SMPTE 125M		

• Frame Amplitude ±3 % of full scale (0.7 V)

• Circle +20 % (HDTV) with respect to green with maximum color amplitude, +20 % (SDTV) with respect to red with maximum color amplitude

Table 7.2 Video signal format and applicable calorimetry standard

7.2 Brightness Adjustment

To adjust the brightness of the vectorscope display or scale, press the VECTOR key 14 to display the vectorscope display menu. Then, press $\fbox{F-1}$ INTEN/SCALE. The brightness and scale adjustment menu appears. Make the appropriate adjustment.



Figure 7.3 Brightness and scale setup menu

7.2.1 Brightness Adjustment of Vector Waveforms

[VECTOR] [F·1] INTEN/SCALE [F·1] VECTOR INTEN [F·D]]

To adjust the brightness of the vectorscope display, press $F\cdot 1$ VECTOR INTEN from the brightness and scale setup menu. Then, turn the function dial ($F\cdot D$) (9) to adjust the brightness.

The adjustment range is from -128 to 127.

The function dial $(F \cdot D)$ (19) also functions as a switch. Press the switch to reset the brightness of vector waveforms to the reference value (0).

7.2.2 Scale Brightness Adjustment

 $\begin{bmatrix} VECTOR & F\cdot1 \\ INTEN/SCALE & F\cdot2 \\ SCALE \\ INTEN & F\cdotD \end{bmatrix}$ To adjust the brightness of the scale on the vectorscope display, press $F\cdot2$ SCALE INTEN from the brightness and scale adjustment menu and turn the function dial $(F\cdotD)$ (19 to make the adjustment.

The adjustment range is from -8 to 7 (16 levels).

The function dial $(F \cdot D)$ (19) also functions as a switch. Press the switch to reset the brightness of the scale on the vectorscope display to the reference value (4).

7.2.3 IQ Axis Display

 $\begin{bmatrix} VECTOR & F\cdot1 & INTEN/SCALE & F\cdot3 & IQ AXIS :ON / OFF \end{bmatrix}$ To display the IQ axis on the vectorscope display, press $F\cdot3 & IQ AXIS$ from the scale setup menu and select ON or OFF. The IQ axis cannot be displayed for 625/50i.

• IQ axis on HDTV The following axis are displayed when the full scale value of 0.7 V is taken to be 100 %.

I axis G=44.559 % Q axis G=37.056 % B=27.865 % B=84.085 % R=69.120 % R=62.417 %



Figure 7.4 IQ axis on the vectorscope display

7.2.4 Scale Color Selection

[VECTOR F-1 INTEN/SCALE F-4 SCALE COLOR :WHITE / YELLOW / CYAN / GREEN / MAGENTA / RED / BLUE]

You can select the scale color on the vectorscope display. You can select the scale color from the seven colors below.

From the brightness and scale setup menu, press $\boxed{F \cdot 4}$ SCALE COLOR to select the color.

WHITE:	Sets the scale color to white.
YELLOW:	Sets the scale color to yellow.
CYAN:	Sets the scale color to cyan.
GREEN:	Sets the scale color to green.
MAGENTA:	Sets the scale color to magenta.
RED:	Sets the scale color to red.
BLUE:	Sets the scale color to blue.

7.3 Gain Adjustment

To adjust the gain of the vector waveforms, press the VECTOR key (14). Then, select F-2 GAIN from menu that appears.

The gain adjustment menu appears. Set the items as desired.



Figure 7.5 Gain adjustment menu

7.3.1 Gain Variable

[VECTOR F·2 GAIN F·1 GAIN VAR :CAL / VAR]

To continuously change the gain of the vector waveform (variable), press $\boxed{F \cdot 1}$ GAIN VAR from the gain adjustment menu.

When you select VAR, you can change the gain continuously using the function dial $(\overline{F \cdot D})$ (9.

Select CAL to set the gain to the reference value. The variable range of gain is x0.20 to x2.00 with respect to the reference gain.

CAL: Sets the reference gain

VAR: Sets the variable gain

The function dial $(F \cdot D)$ (19) also functions as a switch. Press the switch while VAR is selected to reset the gain to the reference value.



Figure 7.6 Gain Variable Display

7.3.2 Gain Selection

 $\begin{bmatrix} VECTOR & F\cdot2 \end{bmatrix} GAIN & F\cdot2 \end{bmatrix} GAIN MAG :X1 / X5 / IQ-MAG \end{bmatrix}$ To change the gain of the video signal waveform to X1 or X5, press F·2 GAIN MAG from the gain adjustment menu.

Select X5 to set the gain to X5. Select X1 to set the gain to the reference value (X1). If you select IQ-MAG, the IQ signal is positioned on the circumference of the vectorscope graticule when the NTSC SMPTE color bar is up-converted to HDTV. The magnification of the IQ-MAG gain is as follows:

COLOR BARS

when set to 75 % 3.1 times

7.4 Line Selector

You can monitor a particular line by using the line selector function of the vectorscope display.

From the vectorscope display menu, press $\boxed{F\cdot3}$ LINE SELECT.

The line select menu appears. Set the items as desired.



Figure 7.7 Line select menu

7.4.1 Line Select

[VECTOR F·3 LINE SELECT F·1 LINE SELECT :ON / OFF]

To select a line to be displayed on the vectorscope display, press $F\cdot 1$ LINE SELECT from the line select menu and select ON. Select OFF to clear the line selector function and display all the lines.

Select the line using the function dial ($\overline{F \cdot D}$) (19).

7.4.2 Field Selection

 $\begin{bmatrix} VECTOR & F\cdot3 \\ LINE SELECT & F\cdot2 \\ FIELD : FIELD1 / FIELD2 / FRAME \end{bmatrix}$ When the line select display is enabled on the vectorscope display, you can set the variable range of the function dial (F·D) (9) to field or frame.

You cannot select the field if the video signal format is progressive.

- FIELD1: Limits the range of lines that can be selected using the function dial $(\overline{F \cdot D})$ 19 to field 1.
- FIELD2: Limits the range of lines that can be selected using the function dial $(\overline{F \cdot D})$ (19 to field 2.
- FRAME: Sets the range of lines that can be selected using the function dial $(\overline{F \cdot D})$ (19) to the entire frame.

7.5 Color System Settings

The vectorscope display of component signals is X-Y display of chrominance difference signals. However, you can also convert the signal into pseudo-composite signal to be displayed on the vectorscope.

You can also set the setup and the color bar saturation.

To set these color system parameters, press $\boxed{F\cdot 4}$ COLOR SYSTEM from the vectorscope display menu.

The color system setup menu appears. Set the items as desired.



Figure 7.8 Color system setup menu

7.5.1 Composite and Component Display

[VECTOR F-4 COLOR SYSTEM F-1 COLOR MATRIX :COMPONEN / COMPOSIT]

To display component signals artificially as composite signals on the vectorscope display, press $\boxed{F \cdot 1}$ COLOR MATRIX from the color system setup menu and select COMPOSIT.

If you select COMPONEN, vectorscope display using chrominance difference signals is shown.

- COMPONEN: Displays the chrominance difference signals of the component signal on the X-Y vectorscope display.
- COMPOSIT: The component signal is artificially converted into a composite signal and shown on the vectorscope display.



(a) Chrominance difference signal vectorscope display



(b) Pseudo-composite vectorscope display

Figure 7.9 Chrominance difference signal vectorscope display and pseudo-composite vectorscope display

7.5.2 Setup Selection

 $\begin{bmatrix} VECTOR & F\cdot4 \\ COLOR SYSTEM & F\cdot2 \\ SETUP :0 \% / 7.5 \% \end{bmatrix}$ If composite display is selected as described in section 7.5.1, "Composite and Component Display," you can select the setup level of the vectorscope display. From the color system setup menu, press F·2 SETUP. If you select 0 %, setup is not added. If you select 7.5 %, a pseudo-composite waveform display with 7.5 % setup added is shown.

You can select setup only when using the pseudo-composite display. The menu does not appear during component display.

7.5.3 Color Bar Saturation Selection

 $\begin{bmatrix} VECTOR & F\cdot4 \\ COLOR SYSTEM & F\cdot3 \\ COLOR BAR :100 \% / 75 \% \end{bmatrix}$ To select the 100 % color bar or 75 % color bar, press $F\cdot3 \\ COLOR BAR$ from the color system setup menu and select 100 % or 75 %.

8. PICTURE DISPLAY

Press the PICTURE key 15 to display the picture.

SDTV is displayed with an aspect ratio of 4 to 3; HDTV is displayed with an aspect ratio of 16 to 9.

The picture display on the LV 5750 shows pictures using simplified resolution conversion. Therefore, it cannot display all the details included in the SDI signal.

The LCD on the LV 5750 shows pictures asynchronously with the input SDI signal by performing frame rate conversion on the internal circuit. Therefore, the screen may flicker due to phenomena known as frame repeat and frame skip.

There are two modes available, FIT mode which displays the picture on full screen and x1 mode which displays the signal pixels without magnification.



Figure 8.1 Area around the PICTURE key (5)



Figure 8.2 Picture Display Menu

8.1 Marker Display

To show the safety marker on the picture display, press $\boxed{F \cdot 1}$ MARKER to show the marker display menu.

As shown in the figure above, four marker sizes are available according to the input signal and aspect marker.

Markers cannot be displayed when showing the multi screen display.



Fligure 8.3 Marker Display Menu



Figure 8.4 HD-SDI Marker Display (4:3 OFF)



Figure 8.5 HD-SDI Marker Display (4:3 ON)



Figure 8.6 SD-SDI Marker Display (16:9 OFF)



Figure 8.7 SD-SDI Marker Display (16:9 ON)

8.1.1 Aspect Marker (4:3)

[PICTURE F-1 MARKER F-1 4:3: ON / OFF]

When displaying HD-SDI signals with the picture aspect ratio of 16:9, the marker can be displayed at the position corresponding to 4:3. Depending on whether the 4:3 aspect marker display is ON, the sizes of the safe action marker and safe title marker are different.

The aspect marker can be turned ON/OFF by pressing $\boxed{F \cdot 1}$ 4:3 from the marker display menu.

8.1.2 Aspect Marker (16:9)

[PICTURE F·1 MARKER F·1 16:9: ON / OFF] When displaying SD-SDI signals with the picture aspect ratio of 4:3, the marker can be displayed at the position corresponding to 16:9. Depending on whether the 16:9 aspect marker display is ON, the sizes of the safe action marker and safe title marker are different.

The aspect marker can be turned ON/OFF by pressing $\boxed{F\cdot 1}$ 16:9 from the marker display menu.

8.1.3 Safe Action Marker

[PICTURE F·1 MARKER F·2 SAFE ACTION: ON / OFF] Safe action markers can be displayed at the 90 % positions vertically and horizontally with respect to the active area on the picture display. If the aspect marker is displayed, the safe action markers are displayed at the

90% positions vertically and horizontally with respect to the aspect marker.

The safety action marker display is selected by pressing $\boxed{F\cdot 2}$ SAFE ACTION from the marker display menu.

8.1.4 Safe Title Marker

[PICTURE F-1 MARKER F-3 SAFE TITLE: ON / OFF]

Safe title markers can be displayed at the 80 % vertical and horizontal positions with respect to the active area on the picture display.

If the aspect marker is displayed, the safe title markers are displayed at the 80% positions vertically and horizontally with respect to the aspect marker.

The safety title marker display is selected by pressing $\boxed{F\cdot3}$ SAFE TITLE from the marker display menu.

8.1.5 Center Marker

[$\underline{PICTURE}$ $\underline{F\cdot 1}$ MARKER $\underline{F\cdot 4}$ CENTER: ON / OFF] A center marker (cross hairs) can be displayed at the center of the picture display.

The center marker display is selected by pressing $\boxed{F \cdot 4}$ CENTER from the marker display menu.

8.2 Line Select Marker Display

You can check the position of a particular line by using the line selector function of the picture waveform.

From the picture display menu, press $\boxed{F\cdot 2}$ LINE SELECT.

The line select menu appears. Set the items as desired.



Figure 8.8 Line select menu

8.2.1 Line Select Marker Display

[PICTURE] F·2 LINE SELECT F·1 LINE SELECT :ON / OFF]

The line select function allows the selected line to be displayed highlighted on the picture display.

The selectable range of line select is within the active picture area; Line select is not displayed during the blanking period.

From the line select menu, press $\boxed{F\cdot 1}$ LINE SELECT to select ON. Then, turn the function dial ($\boxed{F\cdot D}$) (19) to select the line.

8.2.2 Field Selection of Line Select

You cannot select the field if the video signal format is progressive.

- FIELD1: Limits the range of lines that can be selected using the function dial $(\overline{F \cdot D})$ (19 to field 1.
- FIELD2: Limits the range of lines that can be selected using the function dial (F·D) 19 to field 2.
- FRAME: Sets the range of lines that can be selected using the function dial $(\overline{F \cdot D})$ (19) to the entire frame.

When interlace format is displayed as a picture, each field is converted into the XGA rate to be displayed. Since the interlace display is not actually used, the marker position does not change even if you change the field with the line select function.

8.3 Picture Size Change

The picture display on the LV 5750 is adjusted to the number of pixels on the LCD. If an HD-SDI signal, which has more pixels than the LCD, is applied, the image is simply sampled and reduced in size. If an SD-SDI signal, which has less pixels than the LCD, is applied, the pixels are interpolated and the image is enlarged. When an image is simply sampled, narrow lines and points of the image may be lost and not displayed. When an image is displayed by interpolation, the image may appear out of focus. You can display the picture without magnification to prevent these effects.

[PICTURE F·3 SIZE : FIT / x1]

Press $\boxed{F\cdot3}$ SIZE from the picture display menu and select FIT to display the SDI signal adjusted to the LCD size. Select x1 to display the SDI signal using the original number of pixels (not adjusted).

- FIT: Displays the picture by converting the resolution
- x1: Displays the picture without converting the resolution

If x1 is selected for HD-SDI signal input, the picture runs off the LCD. If this happens, you can turn the V POS 9 and H POS 1 dials to move the position of the picture display.

For SD-SDI signals, the pixels are not square. As a result, if x1 is selected the picture will appear distorted, because the aspect ratio will not be 4:3.

9. AUDIO DISPLAY

The embedded audio signal can be separated from the SDI signal and displayed as a sound image or shown on lissajous and level meter displays. Of the 16 channels of embedded audio, two groups where one group consists of four channels can be displayed.

Press the AUDIO key (16) to display audio signals. The screen shows the audio signal waveform, level meter, audio display menu, and other items. Only sampling frequency of 48 kHz is supported for audio signal. The signal must also be synchronized to the video signal.



Figure 9.1 Area around the AUDIO key 16



Figure 9.2 Audio Display Menu

Function Key	Description		
F·1 MODE	Selects the audio waveform display format		
F·2 SDI GROUP	Selects the group from which audio is decoded		
	from the SDI signal		
F·3 SOUND & LISSAJOU	Sets the sound image and lissajous display		
F·4 LEVEL METER	Sets the level meter display		
F·5 PHONES	Sets the headphone output		
F·6 CHANNEL MAPPING	Maps channels on the sound image display		

Table 9.1 Audio display menu description

9.1 Audio Waveform Display Selection

The right-most meter on the lissajous display is called a phase meter. When the meter points to +1, it indicates that the L and R channels are in-phase. Indications of -1 and 0 correspond to reverse phase and no correlation, respectively.

SOUND: Displays the sound image using waveforms.

LISSAJOU: Displays lissajous using waveforms.

MLT_LISS: Displays four-screen (eight channels simultaneously) lissajous using waveforms.

VALUE: Displays audio levels using values.





(b) Lissajous display

СН	1 L	-20.0dB	LR	SL SR	C LFE	RL AR
СН	2 :R	-20.0dB			-6	
СН	8 :S L	-20.0dB	-10	-10	-10	-10
СН	4 :S R	-20.04B	-15	-15	-15	-15
СН	5 ×C	-20.04B	-20	-20	-20	-20
СН	6 LFF	-20.0dB	-30	30	30	30
СН	7 :RL	-20.04B	-40	-40	-40	40
СН	8 :RR	-20.04B	二部 211 2	3	-50 5 6	

(c) Four-screen lissajous display

(d) Value level display

Figure 9.3 Types of audio waveform displays
9.2 Embedded Audio Group Selection

9.3 Audio Waveform Display Setting

To set the audio waveform display, press the AUDIO key 6 to display the audio display menu. Then, select $\boxed{F\cdot 3}$ SOUND & LISSAJOU.

The audio waveform display menu appears. Set the items as desired.





9.3.1 Brightness Adjustment of Audio Waveforms

 $\begin{bmatrix} AUDIO & F\cdot3 \\ SOUND & LISSAJOU & F\cdot1 \\ AUDIO & INTEN & F\cdotD \end{bmatrix}$ To adjust the brightness of audio waveforms, press $F\cdot1 \\ AUDIO & INTEN \\ from the audio waveform display menu and turn the function dial (<math>F\cdotD$) (9) to adjust the level in the range of -8 to 7.

The function dial ($\overline{F \cdot D}$) (19) also functions as a switch. Press the switch to reset the brightness of audio waveforms to the reference value (0).

9.3.2 Scale Brightness Adjustment

 $\begin{bmatrix} AUDIO & F\cdot3 \end{bmatrix} SOUND & LISSAJOU & F\cdot2 \end{bmatrix} SCALE INTEN & F\cdotD \end{bmatrix} \\ To adjust the brightness of the scale, press & F\cdot2 \\ SCALE INTEN from the audio waveform display menu and turn the function dial (F·D) (19) to adjust the level in the range of -8 to 7. \\ \end{bmatrix} \\ \begin{bmatrix} AUDIO & F\cdot3 \\ F\cdot2 \\ F\cdot2$

The brightness adjustment of the scale also affects the brightness of the level meter display.

The function dial $(F \cdot D)$ (19) also functions as a switch. Press the switch to reset the brightness of the scale to the reference value (4).

9.3.3 Sound and Lissajou Display Format

[AUDIO F·3 SOUND & LISSAJOU F·3 SURROUND : 3-1 / 3-2 / 3-2-2] If sound image display (SOUND) is selected in section 9.1, "Audio Waveform Display Selection," you can select the surround system from 3-1 system, 3-2 system, and 3-2-2 system.

The surround system selection is also valid on the single lissajous display in which the L and R of lissajous channels are set to Lt and Rt, respectively. In this case, single lissajous display is shown by performing down-mixing computation according to the surround system. For details, see section 9.3.4, "Channel Selection of Lissajous Display."

To select the surround system, press $\boxed{F\cdot3}$ SOUND from the audio display menu and select 3-1, 3-2, or 3-2-2.



[AUDIO F·3 SOUND & LISSAJOU F·3 LISSAJOU:X-Y / MATRIX] If lissajous display (LISSAJOU) or multi-lissajous (MULTI LISSAJOU) is selected in section 9.1, "Audio Waveform Display Selection," you can select the display format from X-Y display and matrix display.

Press $\overline{F\cdot 3}$ LISSAJOU from the audio waveform display menu and select X-Y or MATRIX.





(a) X-Y lissajous display
 (b) Matrix lissajous display
 Figure 9.6 Lissajous display format

9.3.4 Lissajous Display Channel Selection

[AUDIO F·3 SOUND & LISSAJOU F·4 SINGLE LISSAJ-L : 1st GROUP / 2nd GROUP / Lt]

[AUDIO] F·3 SOUND & LISSAJOU F·5 SINGLE LISSAJ-R : 1st GROUP / 2nd GROUP / Rt]

If lissajous display (LISSAJOU) is selected in section 9.1, "Audio Waveform Display Selection," you can select the vertical and horizontal axes for each embedded audio channels of the lissajous display. The channel selected in section 9.2, "Embedded Audio Group Selection" is displayed on the menu.

If you select Lt, Rt, the following down-mixing computation is carried out according to the name selected in section 9.6, "Channel Mapping" and assigned to Lt and Rt. Always select Lt and Rt in pairs. Otherwise, the lissajous waveforms will not be displayed correctly.

Below are the down-mixing computing equations for each surround system.

Down-mixing computing equation for the 3-1 system

Lt: 0.707 x (L + 0.707C + 0.707S)

Rt: 0.707 x (R + 0.707C + 0.707S)

Down-mixing computing equation for the 3-2 system

Lt: 0.707 x (L + 0.707C + 0.707SL)

Rt: 0.707 x (R + 0.707C + 0.707SR)

Down-mixing computing equation for the 3-2-2 system

Lt: 0.55 x (L + 0.707C + 0.707RL + 0.707SL)

Rt: 0.55 x (R + 0.707C + 0.707RR + 0.707SR)

9.3.5 Gain Selection

 $\begin{bmatrix} AUDIO & F\cdot3 & SOUND & LISSAJOU & F\cdot6 & GAIN:X1 / X2 / X10 / X0.5 / AUTO \end{bmatrix}$ If you selected an item other than numerical display (VALUE) in section 9.1, "Audio Waveform Display Selection," you can set the gain of the waveform display. Press F·6 & GAIN from the audio waveform display menu to select the gain factor.

- X1: Reference gain value. When a reference level is applied, the display matches the scale marker.
- X2: Displays at twice the reference gain.
- X10: Displays at ten times the reference gain.
- X0.5: Displays at 1/2 the reference gain.

AUTO: Automatically adjusts the gain.

9.4 Audio Level Meter Setting

To set the audio level meter, select $\boxed{F \cdot 4}$ LEVEL METER from the function menu that appears by pressing the AUDIO key $\boxed{6}$.

The audio level meter menu appears as shown in Figure 9.7. Select the appropriate items.



Figure 9.7 Audio Level Meter Menu

9.4.1 Reference Level Selection

 $\begin{bmatrix} AUDIO & F\cdot4 \end{bmatrix} LEVEL METER & F\cdot1 \end{bmatrix} REF LEVEL : -20 dB / -18 dB / -12 dB \\ \end{bmatrix}$ Select the reference level of the level meter from -20 dB, -18 dB, and -12 dB. Press $F\cdot1$ REF LEVEL from the audio waveform display menu to select the reference level.

On the level meter display, levels below the reference level are indicated in white, and levels above the reference level are indicated in red.

9.4.2 Dynamic Range Selection

[AUDIO] F·4] LEVEL METER F·2] RANGE: PEAK60 / PEAK90 / AVERAGE] You can select the dynamic range and response of the level meter.

Press $\boxed{F\cdot 2}$ RANGE from the audio waveform display menu to select the dynamic range and response.

PEAK60: Displays a peak level meter capable of measuring down to -60 dB.

PEAK90: Displays a peak level meter capable of measuring down to -90 dB.

AVERAGE: Displays an average-response level meter from -20 dB to +3 dB with the reference level taken to be 0 dB.



The figure shows the scale when the reference level is set to -20 dB and the scale is set to TYPE-A.



9.4.3 Scale Selection

TYPE-A: Level meter is displayed independent of the reference level. TYPE-B: Level meter is displayed with the reference level as 0 dB.



Figure 9.9 TYPE-B level meter according to the reference level Figure 9.9 shows the scale when PEAK60 is selected.

9.4.4 Peak Hold

 $\begin{bmatrix} AUDIO & F\cdot4 \end{bmatrix} LEVEL METER & F\cdot4 \end{bmatrix} PEAK HOLD:0.5 to 5.0 / HOLD \end{bmatrix}$ If you selected PEAK60 or PEAK90 in section 9.4.2, "Dynamic Range Selection," you can select the response time of the peak hold on the level meter. You can set the time in the range of approximately 0.5 s to 5.0 s in 0.5 steps. When you select HOLD, the peak level is held.

Adjust the response time using the function dial $(F \cdot D)$ (19.

9.5 Headphone Output Settings

To set the headphone output, press the AUDIO key 6 to display the audio display menu. Then, select $\boxed{F\cdot5}$ PHONES.

The headphone output menu shown in Figure 9.10 appears. Set each item.

The headphone output generates noise for a few seconds until the internal circuit stabilizes when the signal is inserted or removed or when the SDI signal input is switched.



Figure 9.10 Audio Level display menu

9.5.1 Headphone Output ON/OFF

9.5.2 Headphone Output Volume

[AUDIO] $F \cdot 5$ PHONES $F \cdot 2$ VOLUME : 0 ~ 128] You can control the headphone output volume.

You can also control the headphone output volume using the short cut key. For details, see section 5.8, "Shortcut Key Setting."

9.5.3 Headphone Output Channel Selection

9.6 Channel Mapping



Figure 9.11 Audio display menu page 1

С	LFE	RL	RR	up	up
CH5	CH6	CH7	СН8	menu	menu
F·1	F·2	F·3	F·4 F·5	F·6	F·7

Figure 9.12 Audio display menu page 2

You can map the surround positions to the audio signal channels to produce a correct waveform display on the sound image display.

The selectable groups of channels can be switched as described in section 9.2, "Embedded Audio Group Selection."

10. MULTI SCREEN DISPLAY

Press the MULTI key (18) to display various information such as video signal waveform display and picture display and the multi screen menu.



Figure 10.1 Area around the MULTI key 18



Figure 10.2 Multi screen menu

10.1 Five Multi Screen Modes

[MULTI F-1 MODE :4SCREEN / WFM_VEC / WFM_PIC / WFM_AUD / WFM_LVL]

The following five modes are available for the multi screen display.

- 4SCREEN: 4 screen display. The screen is divided into four sections. The upper right quadrant shows the video signal waveform. The lower right shows the picture display. The upper left quadrant shows the vectorscope display or audio waveform. The lower left quadrant shows the audio level meter display or status display.
- WFM_VEC: Shows the vectorscope and video signal waveform on the left and right sides of the screen, respectively.
- WFM_PIC: Displays the videos signal waveform with the picture at the lower right of the screen.
- WFM_AUD: Shows the audio waveform and video signal waveform on the left and right sides of the screen, respectively.
- WFM_LVL: Shows the audio level meter and video signal waveform on the left and right sides of the screen, respectively.





(a) 4 screen multi display



(c) Waveform display and picture display



 (e) Waveform display and audio level display



(b) Waveform display and vectorscope display

(d) Waveform display and audio waveform display



10.2 Display Content Selection in 4 Screen Mode

[MULTI F·6 UPPER:VECTOR / AUDIO]

[MULTI F·7 LOWER:STATUS / AUD_LVL]

On the 4 screen multi display, the video signal waveform is shown in the upper right quadrant and the picture in the lower right quadrant.

You can select the vectorscope waveform or audio waveform for the upper left quadrant, and status or audio level meter for the lower left quadrant.

Press $\overline{F \cdot 6}$ UPPER of the multi screen menu to select the content to be displayed in the upper left quadrant. Press $\overline{F \cdot 7}$ LOWER to select the content to be displayed in the lower left quadrant.



(a) Select vectorscope display and status (b) Select audio waveform and audio level meter

Figure 10.4 Display content selection of 4 screen display

10.3 Various Settings on the Multi Screen Display

 $\overline{F\cdot 2}$ to $\overline{F\cdot 5}$ keys on page 1 of the multi screen menu is used to set various display items on the multi screen display.

10.3.1 Video Signal Waveform Display Setting on the Multi Screen Display

[MULTI F·2 MULTI WFM]

To set items related to the video signal waveform display on the multi screen display, press F:2 MULTI WFM from page 1 of the multi screen menu. The video signal waveform display menu that appears when the WFM key (3) is pressed appears. Enter the appropriate settings.

The settings related to the video signal waveform display on the multi screen display are passed on when the single video signal waveform display is shown.

For a detailed setup procedure, see the items in chapter 6, "Video Signal Waveform Display."

10.3.2 Vectorscope Waveform Display Setting on the Multi Screen Display

[MULTI F·3 MULTI VEC]

To set items related to the vectorscope display on the multi screen display, press $\overline{F:3}$ MULTI VEC from page 1 of the multi screen menu. The vectorscope display menu that appears when the VECTOR key (1) is pressed appears. Enter the appropriate settings.

The settings related to the vectorscope display on the multi screen display are passed on when the single vectorscope display is shown.

For a detailed setup procedure, see the items in chapter 7, "Vectorscope Display."

10.3.3 Audio Setting on the Multi Screen Display

[MULTI F·4 MULTI AUDIO]

To set items related to the audio display on the multi screen display, press $F\cdot 4$ MULTI AUDIO from page 1 of the multi screen menu. The audio display menu that appears when the AUDIO key (6) is pressed appears. Enter the appropriate settings.

The settings related to the audio display on the multi screen display are passed on when the single audio display is shown.

For a detailed setup procedure, see the items in chapter 9, "Audio Display."

10.3.4 Status Setting on the Multi Screen Display

[MULTI F·5 MULTI STATUS]

To set items related to the status display on the multi screen display, press $\boxed{F\cdot5}$ MULTI STATUS from page 1 of the multi screen menu. The status display menu that appears when the STATUS key $\boxed{17}$ is pressed appears. Enter the appropriate settings.

The settings related to the status display on the multi screen display are passed on when the single status display is shown.

For a detailed setup procedure, see the items in chapter 11, "Status Display."

11. STATUS DISPLAY

The status display can be used to check the status of various parameters of the SDI signal.

Press the STATUS key 1 to display the status screen and the status display menu.



Figure 11.1 Area around the STATUS key 🗇

1080i / 59.94 YCbCr				
SDI	SIGNAL TRS LINE NUMBER	DETECT NORMAL NORMAL	FORMAT	NORMAL
	CRC LUMA	NORMAL	CRC CHROMA	NORMAL
VIDEO	GAMUT	NORMAL	COMP. GAMUT	NORMAL
ANC	PARITY	NORMAL		
	CHECKSUM	NORMAL		
AUDIO	BCH	NORMAL		
	CHANNEL	1, 2, 3, 4, 5	5, 6, 7, 8,	
		9, 10, 11, 12, 13	3, 14, 15, 16	
ETC	CABLE LENGTH	50m		
	ERROR COUNT	0	FROM RESET	00: 10: 00
	LOG MODE	NOW LOGGING	3	
LOG	DATA AUD DUMP	IO ANC PACKE	ERROR CONFIG	CABLE ERROR LENGTH RESET
F·1	F·2 F·3	3 F-4	F-5	F·6 F·7

Figure 11.2 Status Screen and Status Display Menu

Function Key	Description
F·1 LOG	Displays the error log.
F·2 DATA DUMP	Displays the data dump.
F·3 AUDIO	Displays the status of the embedded audio.
F·4 ANC PACKET	Displays the ancillary data
F·5 ERROR CONFIG	Displays the error detection setup menu.
F·6 CABLE LENGTH	Selects the cable type on the cable length meter.
F·7 ERROR RESET	Resets the error occurrence such as the error count.

Table 11.1 Description of the Status Display Menu

11.1 Status Display Screen

Press the STATUS key 1 to display the status screen and the status display menu. Below is the description of the status display.

Display Name	Description		
SIGNAL	Detecs the presence or absence of	SDI signals.	
FORMAT	Auto detection items of video signa	l format.	
TRS	Detect TRS errors.		
LINE NUMBER	Detects line number errors.	(only for HD-SDI)	
CRC LUMA	Detects Y video signal errors.	(only for HD-SDI)	
CRC CHROMA	Detects C_B , C_R video signal errors.	(only for HD-SDI)	
EDH	Detects SD-SDI errors.		
GAMUT	Detects Gamut errors.		
COMP. GAMUT	Detects level errors when the signal is converted into		
	composite signal.		
PARITY	Detects parity errors of ANC data.		
CHECKSUM	Detects checksum errors of ANC da	ata.	
BCH	Detects embedded audio errors.		
CHANNEL	Detects the presence or absence o	of embedded audio	
	on each channel.		
CABLE LENGTH	Converts the SDI signal level into a	coaxial cable length	
	and displays the result.		
ERROR COUNT	Counts the number of errors up to 100000 errors.		
FROM RESET	Time elapsed since error reset.		
LOG MODE	Displays the operation status of the error log function.		

Table 11.2 Description of the Status Screen Display

11.1.1 Status Indication of SDI Signals

(1) SIGNAL Display

SIGNAL on the status display screen indicates whether the SDI signal exists at the input connector.

The presence/absence of SDI signals is detected by the lock/unlock condition of the internal reclocker. If the amplitude of the SDI signal is small or if the amount of jitter is large, the indicator may show NO SIGNAL even when a signal is being applied.

DETECT: The presence of the SDI signal is confirmed. NO SIGNAL: The SDI signal cannot be detected. (2) Format Display

If auto selection of the video signal is enabled in section 5.4.1, "Auto/Manual Selection," the FORMAT indication on the status display screen shows whether the video format of the input SDI signal can be monitored on the LV 5750.

If manual selection is used, it shows whether the selected video format is being input.

For both auto selection and manual selection, the current video format is shown at the upper right corner of the screen.

FORMAT indication

When the video format selection is set to AUTO

- NORMAL: A signal of a video format that can be monitored on the LV 5750 is being input.
- UNKNOWN: A signal of a video format that cannot be monitored on the LV 5750 is being input.

When the video format selection is set to MANUAL

NORMAL: The video format selected manually is being input.

UNKNOWN: A video format other than the one selected manually is being input.

(3) TRS Display

TRS on the status display screen can be used to detect errors in the TRS (Timing Reference Signal) of the input SDI signal.

TRS error detects whether the EAV (End of Active Video) and SAV (Start of Active Video) positions in the SDI signal are in the correct place.

To detect TRS errors, error detection in section 11.6.3, "Detection ON/OFF of Erros" must be turned ON.

TRS indication

NORMAL: The TRS in the SDI signal is normal.

ERROR: There is an error in the TRS in the SDI signal.

Blank: TRS errors in the SDI signal are not being detected.

(4) Line Number Display

LINE NUMBER on the status display screen indicates whether the line number multiplexed in the input HD-SDI signal matches the number of lines that is being counted internally by the LV 5750.

Since the line number is multiplexed only in HD-SDI signals, it is not indicated for SD-SDI signals.

To detect line number errors, error detection in section 11.6.3, "Detection ON/ OFF of Erros" must be turned ON.

LINE NUMBER indication

NORMAL: The line number in the HD-SDI signal is normal.

ERROR: There is an error in the line number in the HD-SDI signal.

Blank: Line number errors in the HD-SDI signal are not being detected.

(5) CRC LUMA and CRC CHROMA Display

CRC LUMA and CRC CHROMA on the status display screen indicate whether the CRC multiplexed in the input HD-SDI signal matches the CRC result that is calculated internally by the LV 5750 for the intensity signal and the chrominance difference signal separately.

Since the CRC is multiplexed only in HD-SDI signals, it is not indicated for SD-SDI signals.

To detect CRC errors, error detection in section 11.6.3, "Detection ON/OFF of Erros" must be turned ON.

CRC LUMA indication

- NORMAL: The CRC of the intensity signal in the HD-SDI signal is normal.
- ERROR: There is an error in the CRC of the intensity signal in the HD-SDI signal.

Blank: CRC errors in the HD-SDI signal are not being detected.

- CRC CHROMA indication
 - NORMAL: The CRC of the chrominance difference signal in the HD-SDI signal is normal.
 - ERROR: There is an error in the CRC of the chrominance difference signal in the HD-SDI signal.

Blank: CRC errors in the HD-SDI signal are not being detected.

(6) EDH Display

EDH on the status display screen indicates whether an error occurred in the EDH that is multiplexed in the input SD-SDI signal. An error occurs when the error flag of the received EDH packet is 1 (error somewhere in the SDI transmission path) or when the CRC in the EDH packet does not match the CRC value calculated in the LV 5750.

Since the EDH is multiplexed only in SD-SDI signals, it is not indicated for HD-SDI signals.

To detect EDH errors, error detection in section 11.6.3, "Detection ON/OFF of Erros" must be turned ON.

For details on the EDH error, see section 11.5.1, "EDH Status Display."

EDH indication

NORMAL:	The EDH in the SD-SDI signal is normal.
ERROR:	There is an error in the EDH in the SD-SDI signal.
NOT FOUND:	The EDH in the SD-SDI signal cannot be found.
Blank:	EDH errors in the SD-SDI signal are not being detected.

11.1.2 Status Display of Ancillary Data

(1) Parity Display

PARITY on the status display screen indicates the result of the error detection using the parity bit included in the header section of the ancillary data in the applied SDI signal.

The parity error detection does not detect errors in the UDW (user data word) section.

To detect parity errors, error detection in section 11.6.3, "Detection ON/OFF of Erros" must be turned ON.

PARITY indication

NORMAL: The parity of the ancillary data is normal. ERROR: Parity error occurred in the ancillary data.

(2) Checksum Display

CHECKSUM on the status display screen indicates the result of the error detection using the checksum included in the ancillary data in the applied SDI signal.

To detect checksum errors, error detection in section 11.6.3, "Detection ON/ OFF of Erros" must be turned ON.

CHECKSUM indication

NORMAL: The checksum of the ancillary data is normal. ERROR: Checksum error occurred in the ancillary data.

11.1.3 Status Display of the Video Signal Level

(1) Gamut Display

GAMUT on the status display screen indicates whether gamut errors have occurred.

You can set upper and lower limits for the gamut error detection values.

To detect gamut errors, error detection in section 11.6.3, "Detection ON/OFF of Erros" must be turned ON.

For the procedure in setting the gamut error detection values, see section 11.7.1, "Gamut Error Threshold Level".

GAMUT indication

- NORMAL: The upper and lower limits do not exceed the specified level.
- ERROR: The specified level is exceeded.

Blank: Gamut errors are not being detected.

(2) Composite Gamut Display

C.GAMUT on the status display screen indicates whether level errors have occurred when the signal was converted into composite signal.

Signals that meet the prescribed levels as Y, C_B , C_R signals may exceed the prescribed levels when converted to composite signal. Composite gamut error monitors the level using the signal converted to composite signal.

You can set upper and lower limits for the composite gamut error detection values.

To detect composite gamut errors, the composite gamut error detection in section 11.6.3, "Detection ON/OFF of Errors" must be turned ON.

For the procedure in setting the threshold levels of composite gamut errors, see section 11.7.2, "Composite Gamut Error Threshold Level".

COMP. GAMUT indication

- NORMAL: The specified upper and lower limits are not exceeded after converting to a composite signal.
- ERROR: The specified level is exceeded after converting to a composite signal.
- Blank: Composite gamut errors are not being detected.

11.1.4 Status Display of Embedded Audio

(1) BCH Display

BCH on the status display screen indicates errors in the BCH code of the embedded audio that is multiplexed in the HD-SDI signal.

Since the BCH is multiplexed only in HD-SDI signals, it is not indicated for SD-SDI signals.

The LV 5750 does not have an error correction function. Therefore, errors will not be corrected even when BCH errors occur.

To detect BCH errors, error detection in section 11.6.3, "Detection ON/OFF of Erros" must be turned ON.

BCH indication

NORMAL: BCH error is not occurring in the embedded audio.

ERROR: BCH error occurred in the embedded audio.

Blank: BCH errors in the embedded audio are not being detected.

(2) Channel Display

CHANNEL on the status display screen indicates embedded audio channels that are multiplexed in the SDI signal.

The multiple channel display can show all 16 channels as defined by the standard.

The presence of each embedded audio channel is detected by monitoring the voice control packets. If the voice control packet flag is not valid, multiplexed channels are not indicated even if embedded audio packets are multiplexed in the SDI signal

CHANNEL indication

- -: Embedded audio is not multiplexed.
- Value: Audio channels corresponding to the indicated values are multiplexed.

11.1.5 Other Status Displays

(1) Cable Display

CABLE LENGTH on the status display indicates the attenuation by converting the attenuation of the input SDI signal to an equivalent cable length by assuming the SDI signal intensity reference to be 800 mV.

1) For HD-SDI signals

The equivalent cable length measurement for HD-SDI signals converts the signal attenuation to an LS-5CFB, L-7CHD or 1694A cable length. Select the type of cable length to be converted into in section 11.9.1, "HD-SDI Cable Selection."

The display range is as follows:

LS-5CFB and 1694A

```
<5 m, 5 m, ..., 125 m , >130 m The display resolution is 5 m.
```

• L-7CHD

<10 m, 10 m, ..., 200 m , >200 m The display resolution is 10 m.

② For SD-SDI signals

The equivalent cable length measurement for SD-SDI signals converts the signal attenuation to a 5C2V, 8281, or 1505A cable length.

Select the type of cable length to be converted into in section 11.9.2, "SD-SDI Cable Selection."

The display range is as follows:

<50 m, 50 m, ..., 295 m, >300 m The display resolution is 5 m.

(2) Error Count Indication

ERROR COUNT on the status display indicates the number of detected errors. The indication range is between 0 and 100,000. You can switch the error count indication to a mode that counts once per second or a mode that counts once per field. For details, see section 11.6.2, "Error Count Rate Setting."

(3) Elapsed Time since Error Reset

FROM RESET on the status display indicates the current elapsed time since error reset.

The built-in realtime clock is used to display the elapsed time.

(4) Log Mode Display

LOG MODE on the status display indicates the operation status of the event log function.

You can select the event log operation in section 11.2.2, "Event Log Start/Stop."

LOG MODE indication

NOW LOGGING: Event log function is enabled.

LOG STOPPED: Event log function is disabled.

11.2 Event Log

Select $\boxed{F \cdot 1}$ LOG from the status display screen to show a log list display of error detection information.



Figure 11.3 Event log display menu

11.2.1 Event log list display

[STATUS F·1 LOG]

Select $\boxed{F \cdot 1}$ LOG from the status display menu that appears by pressing the STATUS key 1 to show the event log list.

The event log handles consecutive errors of the same type as a single log incident. Up to 1,000 errors can be stored in the log.



Figure 11.4 Event Log Display

Error information display

The error information column in the event log only takes the log of the items whose error detection has been turned ON.

The error items are displayed using abbreviations. See the table below for the abbreviations and the error detection information.

The event log records one event per line. If multiple errors occur simultaneously, all the errors cannot be displayed. If this happens, you can save the errors to a compact flash memory card as text and view all the error items.

Abbreviation	Error Description
TRS	TRS error
LINE	HD-SDI line number error
CRC_L	HD-SDI Y video signal transmission error
CRC_C	HD-SDI C _B , C _R video signal transmission error
EDH	SD-SDI transmission error
GMUT	Gamut error
CGMUT	Level error when the signal is converted into composite signal
PRTY	Ancillary data parity error
СНК	Ancillary data checksum error
BCH	Embedded audio transmission error
CABL	Cable length error

Table 11.3 Abbreviations in the event log

11.2.2 Starting/Stopping the Event Log

 $\begin{bmatrix} STATUS & F \cdot 1 \\ LOG & F \cdot 2 \\ LOG & START / STOP \end{bmatrix}$ To start or stop the event log, press $F \cdot 2$ LOG from the event log display menu to select START or STOP.

11.2.3 Clearing the Event Log

 $\begin{bmatrix} STATUS & F \cdot 1 \\ LOG & F \cdot 3 \\ CLEAR \end{bmatrix}$ To clear the event log, press $F \cdot 3 \\ CLEAR$ from the event log display menu. If you clear the event log, the error count on the status display is reset to 0, and all errors are cleared.

11.2.4 Log Operation Mode

[STATUS F-1 LOG F-4 LOG MODE: OVER WR / STOP]

Up to 1000 items can be recorded in the event log.

You can select the operation when the log exceeds 1000 items. You can select to overwrite old logs or stop logging when 1000 items have been recorded.

To select the log operation mode, press $F\cdot 4$ LOG MODE from the event log display menu and select OVER WR (for overwrite) or STOP.

11.2.5 Storage to the Compact Flash Card

The event log can be stored to a compact flash card as a text data. The procedure is indicated below.

* Use FAT16 for the file system on the compact flash card. Writing is not possible on other file systems such as FAT32 and NTFS. ① Showing the save menu to the compact flash card

[STATUS F·1 LOG F·6 CF CARD]

Press $\boxed{F \cdot 6}$ CF CARD from the event log display menu to show the compact flash card save menu.



Figure 11.5 Compact Flash Card Save Menu

2 Specifying the file name

F·1 LOG

[STATUS

Ε.1 ΝΔΜΕ ΙΝΡΙ ΙΤ Ι

From the compact flash card storage menu, press $F\cdot 1$ NAME INPUT to show the file name entry screen.

Enter the file name using the function key $(F \cdot 1 \text{ to } F \cdot 7)$ 2 and the function dial $(F \cdot D)$ (1). After you finish entering the file name, press $F \cdot 7$ up menu. If a space is included in the file name, it is converted to an underscore.



Figure 11.6 File name entry screen

Function Key	Description
F-1 CLEAR ALL	Clears the file name
F·2 DELETE	Clears the character at the file name cursor
F·3 INSERT	Enter a single space at the file name cursor
F·4 ←	Moves the file name cursor to the left by a character
F·5 →	Moves the file name cursor to the right by a character
F·6 CHAR SET	Applies the character at the character selection cursor
	to the file name.
F·7 up menu	Confirms the file name

Table 11.4 File Name Entry Menu

③ Saving the event log to the card

[STATUS F-1 LOG F-6 CF CARD F-2 STORE CARD]

Check that a compact flash card is inserted in the dedicated interface card for the LV 5750. (The interface card is inserted in the external IF slot @ on the rear panel.) Then press F-2 STORE CARD.

The event log is saved to the compact flash card as a text file.

If a compact flash card is not inserted in the dedicated interface card, a message "No Card In Slot" is displayed.

If a text file with the same name already exists on the compact flash card, $\overline{F\cdot 1}$ OVER WR YES, $\overline{F\cdot 3}$ OVER WR NO appears. Press the $\overline{F\cdot 1}$ key to overwrite the file and $\overline{F\cdot 3}$ key to cancel.

After the writing of the event log is complete, a file list of the compact card is displayed.

The file list only displays the event log text files.

11.2.6 File List of the Compact Flash Card

[STATUS] $F\cdot 1$ LOG $F\cdot 6$ CF CARD $F\cdot 3$ FILE LIST] To display the file list of the compact flash card, press $F\cdot 3$ FILE LIST from the compact flash card storage menu. The file list only displays the event log text files.

If a compact flash card is not inserted in the dedicated interface card, a message "No Card In Slot" is displayed.

11.2.7 Log File Deletion

 $\begin{bmatrix} STATUS & F\cdot1 & LOG & F\cdot6 & CF & CARD & F\cdot4 & FILE & DELETE \end{bmatrix}$ To delete an event log file, press $F\cdot4 & FILE & DELETE & from the compact flash card save menu. If you then press <math>F\cdot1 & DELETE & YES$, the selected log file is deleted. If you press $F\cdot3 & DELETE & NO$, the file is not deleted, and the screen returns to the compact flash card save menu.

11.2.8 Auto Increment of File Names

[STATUS F-1 LOG F-6 CF CARD F-5 FILENAME AUTO INC : ON / OFF]

A number can be added automatically to the file name of data stored to the compact flash memory card.

11.3 Data Dump Display

Select $\boxed{F\cdot 2}$ DATA DUMP from the status display menu to show the data dump display screen. The data dump of the selected line and the data dump display menu are displayed.

1080i / 59.94	值 2004/03/	/01 🕒	10:00:00				
DATA DUN [EAV] [EAV] [EAV]	AP LINE # SAMPLE <1920> <1921> <1922> <1922> <1924> <1925> <1926> <1926> <1927> <1928> <1929> <1930> <1931> <1931> <1932> <1933> <1934> <1935>	1 Y 3FF 000 2D8 204 200 2BB 23C 040 040 040 040 040 040 040 040	Cb/Cr 3FF 000 2D8 204 200 2F7 1E8 200 200 200 200 200 200 200 200 200				
MODE RUN F·1	DISPLA SERIAL	Y 	EAV JUMP F·3	SAV JUMP	F.D LINE	CF CARD F·6	up menu F·7

Figure 11.7 Data dump display and data dump display menu

11.3.1 Data Dump Display Auto Updating and Hold

[STATUS] F·2 DATA DUMP F·1 MODE : RUN / HOLD]

The data dump display can be automatically updated or held. From the data dump display menu, press $\boxed{F \cdot 1}$ MODE to select RUN to automatically update the data dump or select HOLD to hold the data dump.

11.3.2 Data Dump Display Format Selection

- SERIAL: Displays the data dump using a serial data array after parallel conversion.
- COMPONENT: Displays the data dump using a serial data array after parallel conversion that has been separated into Y, C_B, C_R.
- BINARY: Displays the data dump using a serial data array after parallel conversion in binary notation.

1080i / 59.94	1 2004/03/	01 🕒	10:00:00	
DATA DUI	VIP LINE #	1		
	SAMPLE	Υ	Cb/Cr	
[EAV]	<1920>	3FF	3FF	
[EAV]	<1921>	000	000	
[EAV]	<1922>	000	000	
[EAV]	<1923>	2D8	2D8	
	<1924>	204	204	
	<1925>	200	200	
	<1926>	2BB	2F7	
	<1927>	23C	1E8	
	<1928>	040	200	
	<1929>	040	200	
	<1930>	040	200	
	<1931>	040	200	
	<1932>	040	200	
	<1933>	040	200	
	<1934>	040	200	
	<1935>	040	200	

1080i / 59.94	值 2004/03/	01 🕒 10	:00:00	
DATA DUN	/IP LINE #	1		
	SAMPLE	Y	Cb	Cr
[EAV]	<1920>	3FF	3FF	
[EAV]	<1921>	000		000
[EAV]	<1922>	000	000	
[EAV]	<1923>	2D8		2D8
	<1924>	204	204	
	<1925>	200		200
	<1926>	2BB	2F7	
	<1927>	23C		1E8
	<1928>	040	200	
	<1929>	040		200
	<1930>	040	200	
	<1931>	040		200
	<1932>	040	200	
	<1933>	040		200
	<1934>	040	200	
	<1935>	040		200

(a) HD-SDI serial data array display

625i / 50					
DATA	DUMP LINE #	1			
	SAMPLE	COLOR	DATA		
[EAV]	< 720>	Cb	3FF		
[EAV]	< 721>	Y	000		
	< 722>	Cr	000		
	< 723>	Y'	2D8		
	< 724>	Cb	000		
	< 725>	Y	3FF		
	< 726>	Cr	3FF		
	< 727>	Y'	2FF		
	< 728>	Cb	2C0		
	< 729>	Y	224		
	< 730>	Cr	290		
	< 731>	Y'	20D		
	< 732>	Cb	21F		
	< 733>	Y	292		
	< 734>	Cr	20D		
	< 735>	Y'	11F		

(c) SD-SDI serial data array display

(b) HD-SDI component display

625i / 50	雪 2004/03/01	اڭ 10:	00:00	
DATA	DUMP LINE #	1		
	SAMPLE	Υ	C.ADR	Cb Cr
[EAV]	< 720>	0	<360>	3FF 000
[EAV]	< 721>	2D8		
[EAV]	< 722>	3FF	<361>	000 3FF
[EAV]	< 723>	2FF		
	< 724>	224	<362>	224 290
	< 725>	20D		
	< 726>	292	<363>	21F 20D
	< 727>	11F		
	< 728>	20D	<364>	294 11F
	< 729>	296		
	< 730>	21F	<365>	20D 170
	< 731>	1BC		
	< 732>	172	<366>	21E 1BC
	< 733>	11E		
	< 734>	1BC	<367>	174 11E
	< 735>	176		

(d) SD-SDI component display

1080i / 59.94	值 2004/03/	01 🕒 10:00:00)				
DATA DUMP LINE # 1							
	SAMPLE	Y	Cb/Cr				
[EAV]	<1920>	11111111111	11111111111				
[EAV]	<1921>	0000000000	0000000000				
[EAV]	<1922>	0000000000	0000000000				
[EAV]	<1923>	1011011000	1011011000				
	<1924>	1000000100	1000000100				
	<1925>	100000000	1000000000				
	<1926>	1010111011	1011110111				
	<1927>	1000111100	0111101000				
	<1928>	0001000000	1000000000				
	<1929>	0001000000	100000000				
	<1930>	0001000000	1000000000				
	<1931>	0001000000	1000000000				
	<1932>	0001000000	100000000				
	<1933>	0001000000	1000000000				
	<1934>	0001000000	1000000000				
	<1935>	0001000000	100000000				

(e) Binary display

Figure 11.8 Data dump display format of SDI signals

11.3.3 Auto Search of EAV/SAV



F·2 DATA DUMP

F·3 EAV JUMP]

You can select whether to display the start sample number of the data dump display using the EAV sample number or SAV sample number. Press $\overline{F\cdot3}$ EAV JUMP from the data dump display menu to display the data dump from the EAV sample number; press $\overline{F\cdot4}$ SAV JUMP to display the data dump from the SAV sample number.

11.3.4 Line Number and Sample Number Selection

- LINE: Sets the range of lines that can be selected using the function dial $(F \cdot D)$ 19 to the entire frame.
- SAMPLE: Selects the sample number of the line specified using the function dial $(\overline{F \cdot D})$ (19.

11.3.5 Storage to the Compact Flash Card

The selected one line of the data dump can be saved to the compact flash card. The procedure is indicated below.

* Use FAT16 for the file system on the compact flash card. Writing is not possible on other file systems such as FAT32 and NTFS. ① Showing the save menu to the compact flash card

[STATUS F·2 DATA DUMP F·6 CF CARD]

Press $\boxed{F \cdot 6}$ CF CARD from the data dump display menu to show the compact flash card save menu.



Figure 11.9 Compact Flash Card Save Menu

② Specifying the file name

[STATUS] F·2 DATA DUMP F·6 CF CARD F·1 NAME INPUT] From the compact flash card save menu, press F·1 NAME INPUT to show the file name entry screen.

Enter the file name using the function key ($\overline{F\cdot 1}$ to $\overline{F\cdot 7}$) 2 and the function dial ($\overline{F\cdot D}$) (9. After you finish entering the file name, press $\overline{F\cdot 7}$ up menu. If a space is included in the file name, it is converted to an underscore.

0	1	2	3	4	5	6	7	8	9	
Α	в	С	D	Е	F	G	н	I	J	
К	L /	М	Ν	0	Р	Q	R	S	т	
U	v	W	X	Y	Z	-				
	Cha	aracter s	election	cursor						
	[F.D_NOB] = CHARSELECT, [F.D_PUSH] = CHAR SET & Function Key EDIT									
DUMP STORE FILE NAME DAT_0505 .TXT										
\ File name cursor										
CLEAR ALLDELETEINSERT←→CHAR SETup menu										
F·1	F	2	F·S	3	F·4		F·5		F·6	F·7

Figure 11.10 File name entry screen

Function Key	Description
F·1 CLEAR ALL	Clears the file name
F·2 DELETE	Clears the character at the file name cursor
F·3 INSERT	Enter a single space at the file name cursor
F·4 ←	Moves the file name cursor to the left by a character
F·5 →	Moves the file name cursor to the right by a character
F·6 CHAR SET	Applies the character at the character selection cursor
	to the file name.
F·7 up menu	Confirms the file name

Table 11.5 File Name Entry Menu

③ Saving the data dump to the card

 $\begin{bmatrix} STATUS & F \cdot 2 \end{bmatrix} DATA DUMP & F \cdot 6 \end{bmatrix} CF CARD & F \cdot 2 \end{bmatrix} STORE CARD \\ \end{bmatrix} Check that a compact flash card is inserted in the dedicated interface card for the LV 5750. (The interface card is inserted in the external IF slot 24 on the rear panel.) Then press F \cdot 2 \\ \\ STORE CARD. \end{bmatrix}$

The data dump is saved to the compact flash card as a text file.

If a compact flash card is not inserted in the dedicated interface card, a message "No Card In Slot" is displayed.

If a text file with the same name already exists on the compact flash card, $\overline{F\cdot 1}$ OVER WR YES, $\overline{F\cdot 3}$ OVER WR NO appears. Press the $\overline{F\cdot 1}$ key to overwrite the file and $\overline{F\cdot 3}$ key to cancel.

After the writing of the data dump is complete, a file list of the compact card is displayed.

The file list only displays the data dump text files.

11.3.6 File List of the Compact Flash Card

 $\begin{bmatrix} STATUS & F\cdot2 \\ DATA DUMP & F\cdot6 \\ CF CARD & F\cdot3 \\ File LIST \end{bmatrix}$ To display the file list of the compact flash card, press $F\cdot3$ FILE LIST from the compact flash card save menu. The file list only displays the text files of the data dump.

If a compact flash card is not inserted in the dedicated interface card, a message "No Card In Slot" is displayed.

11.3.7 Data Dump File Deletion

 $\begin{bmatrix} STATUS & F\cdot 2 \\ DATA DUMP & F\cdot 6 \\ CF CARD & F\cdot 4 \\ FILE DELETE \end{bmatrix}$ To delete a data dump file, press $F\cdot 4$ FILE DELETE from the compact flash card save menu. If you then press $F\cdot 1$ DELETE YES, the selected data dump file is deleted. If you press $F\cdot 3$ DELETE NO, the file is not deleted, and the screen returns to the compact flash card save menu.

11.4 Status Display of Embedded Audio

The LV 5750 enables you to view the bit corresponding to the channel status bit of the AES/EBU audio packet in the embedded audio signal multiplexed in the SDI signal. However, the bit is not displayed when the format is type II data format (consumer mode).

You can display the channel status bit by selecting any channel from the 8 channels of audio signals of 2 groups that were selected in section 9.2, "Embedded Audio Group Selection."

Press $\overline{F\cdot 3}$ AUDIO from the status display screen to display the detailed status of embedded audio and the channel selection menu.



Figure 11.11 Detailed status display of embedded audio

11.4.1 Channel Selection

You can switch the audio channel on the channel status display using $\boxed{F\cdot 1}$ CH SELECT.

The channels that you can select are determined by the selection you make in section 9.2, "Embedded Audio Group Selection."

For details, see section 9.2, "Embedded Audio Group Selection."

11.4.2 Voice Control Packets

For HD-SDI, the voice control packets are normally multiplexed in line 9 or line 571 (interlace only) of the horizontal ancillary data area of the Y channel. For 525i/59.94, the packets are multiplexed in line 12 and line 275 of the horizontal ancillary data area. For 625i/50, the packets are multiplexed in line 8 and line 321. However, if the sampling frequency of the audio signal is 48 kHz, voice control packets do not need to be multiplexed.

The voice control packet display shows the following information from the data multiplexed in the SDI signal as text.

DID indication

Checks the DID of the voice control packets that are multiplexed in the SDI signal and displays the multiplexed audio groups.

Value: The audio groups indicated by the values are multiplexed.

-: Audio group that is not multiplexed.

ACT indication

Checks the ACT of voice control packets and displays the audio channels that are multiplexed in the SDI signal as embedded audio.

Value: The audio channels indicated by the values are multiplexed.

-: Audio channel that is not multiplexed.

RATE indication

Check the RATE of voice control packets and displays the sampling frequency of the selected audio group.

11.4.3 Channel Status

The channel status of the detailed status display of embedded audio shows information by analyzing the channel status bit of embedded audio packets.

FORMAT indication

Indicates whether the signal is an audio signal for the broadcasting studio or a signal for consumer audio devices or the like.

Professional: Indicates that the signal is for the broadcasting studio.

Consumer: Indicates that the signal is for consumer audio devices or audio software manufacturing.

AUDIO DATA indication

Indicates whether the audio packet is an audio signal or data other than the audio signal.

Yes: Audio signal

No: Signal other than audio

EMPHASIS indication

Indicates the audio signal emphasis setting.

Not_indicated:	No emphasis indication
No:	No emphasis
50/15us:	Emphasis time constant of 50/15 μ s
CCIT_J17:	CCITT J.17 (800 Hz insertion loss of 6.5 dB)
Reserved:	Received undefined data

SIGNAL LOCK indication

Lock condition of the source sampling frequency

Yes: Locked

No: Not locked

CH MODE indication

Indicates channel mode setting.

Not_indicated:	No mode indication
Two-channel:	2 channel mode
Single-channel:	1 channel mode
Primary/secondary:	Primary/Secondary mode
Stereo:	Stereo mode
Reserved:	Received undefined data

RESOLUTION indication

Indicates quantization accuracy.

24bits: Indicates audio data of 24-bit quantization accuracy.

20bits: Indicates audio data of 20-bit quantization accuracy.

11.4.4 Channel Status List Display

The channel status list of the detailed status display of embedded audio shows all 192 bits of the channel status of embedded audio packets.

11.5 Ancillary Data Display

The LV 5750 can display the analyzed results of the ancillary data that is multiplexed in the SDI signal. Press $\boxed{F\cdot4}$ ANC PACKET from the status display menu to show the ancillary data display menu.

However, for audio control packets, check the data as described in section 11.1.4, "Status Display of Embedded Audio."



Figure 11.12 Ancillary data display menu

11.5.1 EDH Status Display

In the case of SD-SDI signals, $\boxed{F\cdot 4}$ EDH is displayed in the status display menu, the detailed EDH status can be displayed.

525i / 59.95 🔋 2004/03/01 🕒 1	0:00:00			
STATUS SMPTE RP165 EDH MONITOR DATA				
EDH PACKET N	ORMAL			
UES IDA FF : 0 0 AP : 0 0 ANC : 0 0	IDH EDA 0 0 0 0 0 0	EDH 0 0 0		
RECIEVED CRC FF N AP N	ORMAL ORMAL			
				up menu
F·1 F·2	F·3	F·4 F·5	5 F·6	F·7

Figure 11.13 Detailed EDH status display

The EDH packet flag indicates the contents of the EDH packet that is multiplexed in the input SDI signal. RECEIVED CRC indicates the result of the comparison between the CRC in the EDH packet and the CRC calculated from the data.

Because the SDI output is delivered by passing through only the serial reclock circuit, the packet contents are not overwritten even if an error occurs in the RECEIVED CRC.

The details of each flag are shown below.

- FF: Full field. A CRC code is generated from all the data in a single field, and the result of the error detection is displayed.
- AP: Active picture. A CRC code is generated from the data in the active video period, and the result of the error detection is displayed.
- ANC: Ancillary data. A parity bit and checksum are generated from the ancillary data, and the result of the error detection is displayed.
- EDH: Transmission error detection flag immediately before the LV 5750. A value of 0 indicates normal; 1 indicates an error.
- EDA: Transmission error detection flag from a device before the LV 5750. A value of 0 indicates normal; 1 indicates an error.
- IDH: Error detection flag in the data transmission system inside the LV 5750. A value of 0 indicates normal; 1 indicates an error.

- IDA: Data transmission error detection flag in the device before the LV 5750. A value of 0 indicates normal; 1 indicates an error.
- UES: Indicates whether the previous equipment supports EDH packets.
 - 0: The entire SDI signal system supports EDH packets.
 - 1: An equipment that does not support EDH packets is connected.

RECEIVED CRC FF: Full field CRC reception error

NORMAL: The full field CRC of the EDH packet and the full field CRC recalculated from the received data match (no reception error).

ERROR: The full field CRC of the EDH packet and the full field CRC recalculated from the received data do not match (reception error).

RECEIVED CRC AP: Active picture CRC reception error

- NORMAL: The active picture CRC of the EDH packet and the active picture CRC recalculated from the received data match (no reception error).
- ERROR: The active picture CRC of the EDH packet and the active picture CRC recalculated from the received data do not match (reception error).

11.6 Error Detection Setting

Press $\boxed{F\cdot5}$ ERROR CONFIG from the status display menu to show the error configuration menu as shown in figure 11.14.



Figure 11.14 Error configuration menu

Function Key	Description
F-1 REMOTE ERR OUT	Sets the error signal output when the Remote
	Control Unit sold separately is installed
F·2 COUNT RATE	Sets the count rate of the error count.
F·3 ERROR DETECT	Sets whether to detect each error.
F·4 ERROR LEVEL	Sets the error threshold level.
F·5 ERROR DISPLAY	Selects whether to refresh or hold the error display.

 Table 11.6 Error configuration menu description

11.6.1 Error Alarm Setting

Press F·1 REMOTE ERR OUT from the error setup menu to select the polarity of the error signal that is output from the optional Remote Control Unit that is sold separately.

[STATUS F·5 ERROR CONFIG F·1 REMOTE ERR OUT : OFF / POSITIVE / NEGATIVE]

Error alarm output pins are assigned to the REMOTE connector of the optional Remote Control Unit that is inserted in the external IF slot 2 on the rear panel. This output transmits a pulse when an error occurs in any of the items that have been turned ON in section 11.6.3, "Detection ON/OFF of Errors".

OFF: Errors are not output from the remote connector.

POSITIVE: A high level signal is output when an error occurs.

NEGATIVE: A low level signal is output when an error occurs.

For the location of the output pin, see the instruction manual of the option (Remote Control Unit) that is sold separately.

11.6.2 Error Count Rate Setting

 $\begin{bmatrix} STATUS & F \cdot 5 \end{bmatrix} ERROR CONFIG & F \cdot 2 \\ COUNT RATE V RATE / 1 sec \\ Press & F \cdot 2 \\ COUNT RATE from the error configuration menu to select the count rate of the error count on the status display screen. \\ \end{bmatrix}$

- V RATE: Increments the error count per field (frame for progressive). Even if multiple errors occur within a field, the error count is incremented only by 1.
- 1sec: Increments the error count per second. Even if multiple errors occur within a second, the error count is incremented only by 1.

11.6.3 Detection ON/OFF of Errors

Press $\boxed{F\cdot3}$ ERROR DETECT from the error configuration menu to show the error detection setup menu. You can turn ON/OFF the error detection function of each error from this menu.



(a) Error detection setup menu page 1

PARITY ERROR ON	CHECKSUM ERROR ON		GAMUT ERROR OFF	C.GAMUT ERROR OFF	next menu	up menu
F·1	F·2	F·3	F·4	F·5	F·6	F·7

(b) Error detection setup menu page 2



(c) Error detection setup menu page 3

Figure 11.15 Error detection setup menu

(1) TRS Error ON/OFF

[STATUS] F·5 ERROR CONFIG F·3 ERROR DETECT F·1 TRS ERROR:ON / OFF]

To turn ON/OFF the TRS error detection function, press $\boxed{F \cdot 1}$ TRS ERROR from the error detection setup menu to select ON or OFF.

(2) Line Number Error ON/OFF

[STATUS] F·5] ERROR CONFIG F·3] ERROR DETECT F·2] LINE ERROR :ON / OFF]

To turn ON/OFF the line number error detection function, press $F\cdot 2$ LINE ERROR from the error detection setup menu to select ON or OFF. The detection function only works on HD-SDI signals.

(3) CRC Error ON/OFF

[STATUS F·5] ERROR CONFIG F·3] ERROR DETECT F·3] CRC ERROR:ON / OFF]

To turn ON/OFF the CRC error detection function when HD-SDI signals are input, press $\boxed{F\cdot3}$ CRC ERROR from the error detection setup menu to select ON or OFF.

(4) EDH Error ON/OFF

[STATUS F·5 ERROR CONFIG F·3 ERROR DETECT F·4 EDH ERROR:ON / OFF]

To turn ON/OFF the EDH error detection function when SD-SDI signals are input, press $\boxed{F \cdot 4}$ EDH ERROR from the error detection setup menu to select ON or OFF.

If the function is turned OFF, the result of EDH error detection is not displayed on the status display screen. However, the status of the EDH flag is displayed in the detailed status display of EDH.

(5) Parity Error ON/OFF

[STATUS F·5 ERROR CONFIG F·3 ERROR DETECT F·6 next menu F·1 PARITY ERROR:ON / OFF]

To turn ON/OFF the parity error detection function, press $\boxed{F \cdot 6}$ next menu from the error detection setup menu followed by $\boxed{F \cdot 1}$ PARITY ERROR to select ON or OFF.

(6) Checksum Error ON/OFF

[STATUS] F·5 ERROR CONFIG F·3 ERROR DETECT F·6 next menu F·2 CHECKSUM ERROR : ON / OFF]

To turn ON/OFF the checksum error detection function, press $\boxed{F\cdot 6}$ next menu from the error detection setup menu followed by $\boxed{F\cdot 2}$ CHECKSUM ERROR to select ON or OFF.

(7) Gamut Error ON/OFF

[STATUS] F·5 ERROR CONFIG F·3 ERROR DETECT F·6 next menu F·4 GAMUT ERROR:ON / OFF]

To turn ON/OFF the gamut error detection function, press $\boxed{F\cdot 6}$ next menu from the error detection setup menu followed by $\boxed{F\cdot 4}$ GAMUT ERROR to select ON or OFF.

(8) Composite Gamut Error ON/OFF

[STATUS] F·5 ERROR CONFIG F·3 ERROR DETECT F·6 next menu F·5 C.GAMUT ERROR:ON / OFF]

To turn ON/OFF the composite gamut error detection function, press $\boxed{F \cdot 6}$ next menu from the error detection setup menu followed by $\boxed{F \cdot 5}$ C.GAMUT to select ON or OFF.

(9) BCH Error ON/OFF

[STATUS F·5 ERROR CONFIG F·3 ERROR DETECT F·6 next menu F·6 next menu F·1 BCH ERROR:ON / OFF]

To turn ON/OFF the BCH error detection function, press $\boxed{F \cdot 6}$ next menu twice from the error detection setup menu followed by $\boxed{F \cdot 1}$ BCH ERROR to select ON or OFF.

The detection function only works on HD-SDI signals.
(10) Equivalent Cable Length Meter Alarm ON/OFF

[STATUS] F·5 ERROR CONFIG F·3 ERROR DETECT F·6 next menu F·6 next menu F·4 CABLE ERROR :ON / OFF] To turn ON/OFF the function for sounding an alarm (cable length meter indicated using red characters) when the cable length measured by the equivalent cable length meter function exceeds a specified value, press F·6 next menu twice from the error detection setup menu followed by F·4 CABLE ERROR to select ON or OFF.

11.7 Detection Value Setting of Errors

Select $\boxed{F\cdot 4}$ ERROR LEVEL from the error configuration menu to show the error level selection menu for setting the threshold levels for error detection.



Figure 11.16 Error level item selection menu

Function Key	Description
F·1 GAMUT	Sets the gamut error threshold level.
F·2 COMPOSIT GAMUT	Sets the composite gamut error threshold level.
F·3 CABLE	Set the cable length meter error threshold level.

Table 11.7 Description of the error level item selection menu

11.7.1 Gamut Error Threshold Level

Press $\boxed{F\cdot 1}$ GAMUT from the error level item selection menu to show the gamut error threshold level setup menu. You can set the gamut error threshold level from this menu.



Figure 11.17 Gamut error threshold level setup menu

[STATUS	F·5 ERROR CONFIG	F·4 ERROR LEVEL	F·1 GAMUT	F·1
GAMUT UP	PER:90.8% ~ 109.4%]			
[STATUS	F·5 ERROR CONFIG	F·4 ERROR LEVEL	F·1 GAMUT	F·2
GAMUT LO	NER:-7.2% ~ 6.1%]			

From the gamut error threshold level setup menu, press $\boxed{F\cdot 1}$ GAMUT UPPER or $\boxed{F\cdot 2}$ GAMUT LOWER and set the threshold levels to define signal levels used to determine gamut errors.

If a signal level greater than the UPPER value is detected, a gamut error occurs. If a signal level smaller than the LOWER value is detected, a gamut error occurs.

11.7.2 Composite Gamut Error Threshold Level

Press $\overline{F\cdot 2}$ COMPOSIT GAMUT from the error level item selection menu to show the composite gamut error threshold level setup menu. You can set the composite gamut error threshold level from this menu.



Figure 11.18 Composite gamut error threshold level setup menu

[STATUS]	F·5 ERROR CONFIG	F-4 ERROR LEVEL	F·2 COMPOSIT
GAMUT	F·1 C. GAMUT UPPER:9	0.0% ~ 135.0%]	
[STATUS	F·5 ERROR CONFIG	F-4 ERROR LEVEL	F·2 COMPOSIT
GAMUT	F·2 C. GAMUT LOWER:-	-40.0% ~ -20.0%]	

From the composite gamut error threshold level setup menu, press $\overline{F \cdot 1}$ C. GAMUT UPPER or $\overline{F \cdot 2}$ C. GAMUT LOWER and set the threshold levels to define signal levels used to determine composite gamut errors.

If a signal level greater the specified UPPER value is detected when the SDI signal is converted to composite signal, a composite gamut error occurs. If a signal level less than the specified LOWER value is detected when the SDI signal is converted to composite signal, a composite gamut error occurs.

11.7.3 Detection Value of Equivalent Cable Length Meter Error

Press $\boxed{F\cdot3}$ CABLE from the error level item selection menu to show the cable length meter error threshold level setup menu. You can set the cable length meter error threshold level from this menu.



Figure 11.19 Cable length meter error threshold level setup menu

(1) Cable Length Meter Error of HD-SDI Signals

[STATUS F·5 ERROR CONFIG F·4 ERROR LEVEL F·3 CABLE F·1 HD CABLE LENGTH: 5m ~ 200m]

From the cable length meter error threshold setup menu, press $\boxed{F \cdot 1}$ HD CABLE LENGTH and set the cable length (signal attenuation) used to determine equivalent cable length meter errors in the HD-SDI input signal.

If the cable length meter indication is greater than the specified value, an equivalent cable length meter error occurs.

If an error occurs, the equivalent cable length meter indication for CABLE LENGTH in the status display menu is shown in red.

(2) Cable Length Meter Error of SD-SDI Signals

[STATUS F·5 ERROR CONFIG F·4 ERROR LEVEL F·3 CABLE F·2 SD CABLE LENGTH: 50m ~ 300m]

From the cable length meter error threshold setup menu, press $F\cdot 2$ SD CABLE LENGTH and set the cable length (signal attenuation) used to determine equivalent cable length meter errors in the SD-SDI input signal.

If the cable length meter indication is greater than the specified value, an equivalent cable length meter error occurs.

If an error occurs, the equivalent cable length meter indication for CABLE LENGTH in the status display menu is shown in red.

(3) Cable Length Meter Alarm of HD-SDI Signals

[STATUS] F·5 ERROR CONFIG F·4 ERROR LEVEL F·3 CABLE F·3 HD CABLE WARNING: 5m ~ 200m]

From the cable length meter error threshold setup menu, press $\boxed{F\cdot3}$ HD CABLE WARNING and set the cable length (signal attenuation) used to determine equivalent cable length meter alarms of the HD-SDI input signal.

If the cable length meter indication is greater than the specified value, an equivalent cable length meter alarm is indicated.

If an alarm occurs, the equivalent cable length meter indication for CABLE LENGTH in the status display menu is shown in yellow.

(4) Cable Length Meter Alarm of SD-SDI Signals

[STATUS F·5 ERROR CONFIG F·4 ERROR LEVEL F·3 CABLE F·4 SD CABLE WARNING: 50m ~ 300m]

From the cable length meter error threshold setup menu, press $\boxed{F\cdot 4}$ SD CABLE WARNING and set the cable length (signal attenuation) used to determine equivalent cable length meter alarms of the SD-SDI input signal.

If the cable length meter indication is greater than the specified value, an equivalent cable length meter alarm is indicated.

If an alarm occurs, the equivalent cable length meter indication for CABLE LENGTH in the status display menu is shown in yellow.

11.8 Error Display Operation

 $\begin{bmatrix} STATUS & F \cdot 5 \end{bmatrix} ERROR CONFIG & F \cdot 5 \\ ERROR DISPLAY: REFRESH / HOLD \\ There are two methods of displaying errors on the status display. The error indication can be set to disappear after approximately 1 second or remain displayed until the error is reset. \\ \end{bmatrix}$

From the status display menu, press $\overline{F\cdot5}$ ERROR DISPLAY. Select REFRESH to display error indications for approximately 1 second. Select HOLD to hold error indications until errors are reset.

11.9 Cable Selection of the Cable Length Meter

You can select the type of cable to which the SDI signal level is converted on the cable length meter on the status display.

From the status display menu, press $\boxed{F \cdot 6}$ CABLE LENGTH to show the cable selection menu of the equivalent cable length meter.



Figure 11.20 Cable selection menu of the cable length meter

11.9.1 HD-SDI Cable Selection

11.9.2 SD-SDI Cable Selection

11.10 Error Reset

[STATUS F·7 ERROR RESET]

To reset the error indication on the status display screen and the error log, press $\boxed{F \cdot 7}$ RESET from the status display menu. The error indication is reset with a beep sound.

12. CAPTURE FUNCTION

The LV 5750 can copy the screen display to the internal memory or a compact flash card. If the screen display is saved to the internal memory, the data is lost when the screen display is changed or when the power is turned OFF. If the screen display is saved to a compact flash card, the data is saved even when the power is turned OFF. The data can be read on a PC as bitmap images.

Press the CAPTURE key (5) to capture the screen image to the internal memory of the LV 5750. The capture menu appears.



Figure 12.1 CAPTURE key 5



Figure 12.2 Capture menu

Function Key	Description
F-1 PUSH HOLD	Captures the screen to the internal memory.
F·2 DISPLAY	Selects to display the captured image.
F·3 CF CARD	Performs compact flash card operation.
F·4 FILE SELECT	Selects the file type to be saved, BMP or BSX.

Table 12.1 Capture Menu Description

Carry out the following procedure to use the capture function.

- (1) Show the display you wish to capture.
- (2) Press the CAPTURE key 5 to capture the displayed contents in the LV 5750.
- (3) Press F·2 DISPLAY to select whether to show the captured image or the current content. You can also select overlay display.
 To store to a compact flash card, continue with the following procedure.

1) Press [2] CE CARD to show the compact flock could store to menu

- (4) Press $F\cdot 3$ CF CARD to show the compact flash card storage menu.
- (5) Press F·1 NAME INPUT to specify the name of the captured image file. After you have specified the name, press F·7 up menu.
- (6) Press F:2 STORE CARD to store the data to the compact flash card.

12.1 Screen capture

To use the capture function, show the display you wish to capture. Then, press the CAPTURE key (5) to capture the displayed contents in the LV 5750.

The image data captured in the internal memory is cleared when you switch the screen display such as when you switch from the waveform display to the vectorscope display. The contents in the internal memory are also cleared when the power is turned OFF.

Simply capturing the display does not show the captured image on the screen. Carry out the procedure described in section 12.2, "Display Selection" to show the captured image.

Press $\boxed{F \cdot 1}$ PUSH HOLD from the capture menu to capture the displayed contents again.

12.2 Display Selection

You can select how the captured data in the internal memory is displayed.

You can select the display method from 1) display the newest information, 2) display the captured image, and 3) display the newest information and captured image overlaid.

When using overlay display, the brightness of the newest information is cut to half, and the brightness of the captured image is cut to half.

DISPLAY selection

REAL: Displays the newest information.

- HOLD: Displays the captured image.
- BOTH: Displays both newest information and captured image overlaid.

If you select HOLD, the video signal waveform display, vectorscope display, audio waveform display, data dump, and picture display show the captured image. However, the scale, readout, and audio level meter continue showing the newest information.

The bitmap data stored to the compact flash card is the information existing at the time the capture operation is carried out.

12.3 Storage to the Compact Flash Card

The image data captured in the internal memory of the LV 5750 is cleared when you switch the display or when you turn OFF the power.

To keep the captured image from being cleared, the data must be stored to a compact flash card. By storing the data in a compact flash card, the image can be displayed on another device such as your PC, or it can be rewritten to the internal memory of the LV 5750 at a later time.

The captured image can be transmitted directly to a PC via the Ethernet network.

* Use FAT16 for the file system on the compact flash card. Writing is not possible on other file systems such as FAT32 and NTFS. ① Showing the compact flash card storage menu

[CAPTURE F·3 CF CARD]

From the error log display menu, press $\boxed{F\cdot3}$ CF CARD to show the compact flash card storage menu.



Figure 12.3 Compact flash card storage menu

② Specifying the file name

[CAPTURE] F·3 CF CARD F·1 NAME INPUT]

From the compact flash card storage menu, press $\boxed{F \cdot 1}$ NAME INPUT to show the file name entry screen.

Enter the file name using the function key ($\overline{F\cdot 1}$ to $\overline{F\cdot 7}$) 2 and the function dial ($\overline{F\cdot D}$) (1). After you finish entering the file name, press $\overline{F\cdot 7}$ up menu. If a space is included in the file name, it is converted to an underscore.



Figure 12.4 File name entry screen

Function Key	Description
F·1 CLEAR ALL	Clears the file name
F·2 DELETE	Clears the character at the file name cursor
F·3 INSERT	Enter a single space at the file name cursor
F·4	Moves the file name cursor to the left by a character
F·5	Moves the file name cursor to the right by a character
F·6 CHAR SET	Applies the character at the character selection cursor
	to the file name.
F·7 up menu	Confirms the file name

Table 12.2 File name entry menu

3 Storing the image data to the card

 $\begin{bmatrix} CAPTURE & F\cdot3 \\ CF CARD & F\cdot2 \\ STORE CARD \end{bmatrix}$ Check that a compact flash card is inserted in the dedicated interface card for the LV 5750. (The interface card is inserted in the external IF slot 24 on the rear panel.) Then press F·2 STORE CARD.

The image data is saved to the compact flash card as a capture data file.

If a compact flash card is not inserted in the dedicated interface card, a message "No Card In Slot" is displayed.

After the writing of the capture data is complete, a file list of the compact card is displayed.

The file list only displays the capture data bitmap files.

* BSX files

When a captured image data is written to the compact flash card, a BMP (bitmap) file and a BSX file are created. The bitmap file is used when displaying the captured image data on another device such as your PC. The BSX file is used to recall the image data on the LV 5750 and contains setup items of the LV 5750 that were used at the time the image was captured in addition to the waveform data.

Bitmap data cannot be used to recall the captured image on the LV 5750. To select whether to save the BSX file or BMP file, carry out the procedure given in section 12.8, "Storage File Selection."

12.4 Recalling from the Compact Flash Card

[CAPTURE F·3 CF CARD F·6 RECALL CARD]

BSX files are needed to recall the image data stored to the compact flash card. Insert the compact flash card containing the BSX file into the CF CARD slot on the rear panel. From the capture menu, press $\boxed{F\cdot3}$ FILE LIST to show a list of captured image files on the compact flash card.

Select the file to be recalled from the list using the function dial ($\overline{F \cdot D}$) (19). Next, press $\overline{F \cdot 6}$ RECALL CARD to recall the captured image into the LV 5750.

The LV 5750 is configured according to the settings included in the BSX file. However, the audio level meter display and the readout values at the time errors occurred cannot be recalled from the compact flash card, because they are not stored in the BSX file.

12.5 File List of the Compact Flash Card

 $\begin{bmatrix} CAPTURE & F\cdot3 \\ CF CARD & F\cdot3 \\ FILE LIST \end{bmatrix}$ To display the file list of the compact flash card, press F·3 FILE LIST from the

If a compact flash card is not inserted in the dedicated interface card, a message "No Card In Slot" is displayed.

compact flash card storage menu. The file list only displays the captured BSX files.

12.6 Deletion of Capture Files

 $\begin{bmatrix} CAPTURE & F\cdot3 & CF & CARD & F\cdot4 & FILE & DELETE \end{bmatrix}$ To delete a capture file, press $F\cdot4 & FILE & DELETE & from the compact flash card storage menu. Then, press <math>F\cdot1 & DELETE & YES & to delete & the selected capture & file. Press <math>F\cdot3 & DELETE & NO & to & not & delete & the & file. The & screen & returns & to & the compact flash card & storage & menu. & file. The & screen & returns & to & the compact & flash & card & storage & menu. & file. & file$

12.7 Auto Increment of File Names

12.8 Storage File Selection

[CAPTURE] F-4 FILE SELECT: BMP&BSX / BMP / BSX]

From the capture menu, press $\boxed{F\cdot 4}$ FILE SELECT. You can select the type of file for storing the captured image data to the compact flash card.

BMP&BSX: Stores the bitmap file and BSX file to the compact flash card. One capture image requires approximately 4.7 MB of free space. The bitmap file can be used to display the image on a device such as a PC, and the BSX file can be used to display the image on the LV 5750.

- BMP: Stores only the bitmap file to the compact flash card. One capture image requires approximately 2.4 MB of free space. The bitmap file cannot be used to display the image on the LV 5750.
- BSX: Stores only the BSX file to the compact flash card. One capture image requires approximately 2.4 MB of free space. The BSX file is used to display the image on the LV 5750.

13. PRESET FUNCTION

You can register up to 30 sets of settings excluding the power ON/OFF setting and Ethernet settings on the LV 5750 and recall them later.

In addition, settings stored to the memory can be recalled via the REMOTE connector by installing the optional Remote Control Unit sold separately into the external IF slot @ on the rear panel.

13.1 Recalling of Settings

Press the RECALL key (2) to display the recall menu. You can recall stored panel settings from this menu. Recall numbers are assigned to the ($\overline{F\cdot1}$ to $\overline{F\cdot6}$) function keys (2). If there are more than six sets of stored settings, press $\overline{F\cdot7}$ more to display them.



Figure 13.1 RECALL key ② MEMORY key ③



Figure 13.2 Recall menu

13.2 Registration of Settings

Press the MEMORY key (3) to display the memory menu. The settings existing at that point can be saved using this menu. You can attach a comment for each setup item.



Memory menu page

Figure 13.3 Memory menu

Function Key	Description
F-1 COMMENT INPUT	Enters comments
F·2 STORE	Stores the settings
F·3 DELETE	Clears the memory
F·4 RECALL	Recalls presets
	(Screen display remains at the preset list)
F·5 ALL COPY	Copies all the presets stored on the compact
CARD-INT	flash card to the internal memory
F·6 ALL COPY	Copies all the presets stored in the internal
INT-CARD	memory to the compact flash card

Table 13.1 Memory Menu Description

13.2.1 Comment Entry

[MEMORY F·1 COMMENT INPUT]

You can enter a comment for each preset number.

To enter a comment, press $\boxed{F \cdot 1}$ COMMENT from the memory menu to display the comment entry screen.

Enter the comment using the function key ($\overline{F\cdot 1}$ to $\overline{F\cdot 7}$) 2 and the function dial ($\overline{F\cdot D}$) (19. When you are done, press $\overline{F\cdot 7}$ up menu to return.



Figure 13.4 File name entry screen

Function Key	Description
F·1 CLEAR ALL	Clears the file name
F·2 DELETE	Clears the character at the file name cursor
F·3 INSERT	Enter a single space at the file name cursor
F·4	Moves the file name cursor to the left by a character
F·5	Moves the file name cursor to the right by a character
F·6 CHAR SET	Applies the character at the character selection cursor
	to the file name.
F·7 up menu	Confirms the file name
F·D	Moves the character cursor

Table 13.2 File name entry menu

13.2.2 Storing of the Settings

After entering the comment, turn the function dial ($\overline{F \cdot D}$) (19) on the memory menu and select the desired preset number for registering the settings. Then, press $\overline{F \cdot 2}$ STORE to store the settings.

13.2.3 Deleting of the Settings

To delete settings that have been registered as a preset, turn the function dial $(\overline{F \cdot D})$ (9) on the memory menu to select the preset number you wish to delete. Then, press $\overline{F \cdot 3}$ DELETE. $\overline{F \cdot 1}$ DELETE YES and $\overline{F \cdot 3}$ DELETE NO appear for you to confirm the operation. Press $\overline{F \cdot 1}$ to delete or $\overline{F \cdot 3}$ to cancel.

13.2.4 Editing of Presets

To edit or recall the preset contents, press $\boxed{F \cdot 4}$ RECALL from the memory menu. The screen display keeps showing the list of presets, but when you exit from the memory menu, the recalled settings are activated.

The recall function here is mainly used to move the preset number or copy the contents to another preset number.

After recalling a preset, select a preset number and press $\boxed{F\cdot 2}$ STORE to copy the contents of the recalled preset.

13.3 Copying of All Presets

If you wish to make the presets on multiple LV 5750 monitors the same, you can use the compact flash card to do so.

* Use FAT16 for the file system on the compact flash card. Writing is not possible on other file systems such as FAT32 and NTFS.

13.3.1 Copying from the CF Card to the Internal Memory

[MEMORY F·5 ALL COPY CARD-INT]

From the memory menu, press $\overline{F\cdot 5}$ ALL COPY CARD-INT to copy all the presets on the compact flash card to the internal preset memories.

13.3.2 Copying from the Internal Memory to the CF Card

[MEMORY] F·6 ALL COPY INT-CARD]

From the memory menu, press $\overline{F \cdot 6}$ ALL COPY INT-CARD to copy all the presets in the internal memories to the compact flash card.

14. ADDITIONAL FUNCTION

14.1 Specifications

14.1.1 General

This section describes specifications and functions that are added when the firmware version 1.x of LV 5750 is updated to 2.1.

When viewing this manual, be sure to also refer to other section.

14.2 New Functions

Additional Video Signal Formats

The following video signal formats have been added: 720p/50, *1720p/30, *1720p/29.97, *1720p/25, *1720p/24, and *1720p/23.98.

Audio CRC Error Detection

CRC errors of the channel status data that is multiplexed in the embedded audio can be detected.

V-ANC Monitor Function

List Display

The presence/absence of ancillary data including V-ANC can be listed.

Format ID Display

Displays the current video mode as defined by ARIB STD B-39 or the PAYLOAD ID packet as defined by SMPTE 352M.

V-ANC Data Display of the ARIB Standard

Displays the inter-stationary control data as defined by ARIB STD-B39. Displays the closed caption data as defined by ARIB STD-B37.

Signal Name	Standard Supported	DID	SDID
Format ID	SMPTE 352M	241h	101h
V-ANC ARIB Standard			
Inter-stationary control	ARIB STD-B39	25Fh	1FEh
data			
Data broadcast trigger	ARIB STD-B35	25Fh	1FDh
signal			
Digital closed caption	ARIB STD-B37	25Fh	1DFh (HD closed caption)
			2DEh (SD closed caption)
			2DDh (Analog)
User data 1	ARIB TR-B23	25Fh	2FCh
User data 2	ARIB TR-B23	25Fh	1FBh

Standard Supported

- Remaining Battery Power Display Displays the remaining power in the IDX or Anton/Bauer *2 battery.
- SNMP Support SNMP (Simple Network Management Protocol) can be used to control the LV 5750 and notify SDI signal errors.
- Time Code Display Displays the time code (LTC or VITC) as defined by SMPTE RP-188.
- Gamut Bar Display Displays a peak level meter for Y, G, B, R, and pseudo-composite signals.
- Data Dump ADF/DID/DBN/SDID Marker Display
- Performance Improvements

Support for large capacity compact flash cards and reduction in the time needed to write data

Magnification of the V scale on the composite waveform display Enhancement to the TRS error detection function (for 525i/59.94)

- *1 Formats whose evaluation verification has not been completed as of March 2005.
- * 2 Installation of the Anton/Bauer battery mount is a special order option.

14.3 Menu Structure

Menus indicated by italics are functions that are added when the firmware version 1.x of LV 5750 is updated to 2.1.











Menus indicated by italics are functions that are added when the firmware version 1.x of LV 5750 is updated to 2.1.

- * 3 Menu shown when the Remote Control Unit that is sold separately is inserted.
- * 4 Menu shown when the Ethernet Unit that is sold separately is inserted.

14.4 System Configuration

14.4.1 Additional Video Signal Formats

14.4.1.1 Video Signal Formats That Have Been Added

The 720p video signal formats 720p/50, 720p/30, 720p/29.97, 720p/25, 720p/ 24, and 720p/23.98 have been added to the conventional 720p/60 and 720p/ 59.94.

However, verifications have not been completed for 720p/30, 720p/29.97, 720p/ 25, 720p/24, and 720p/23.98 as of March 2005.

14.4.1.2 Video Signal Format Setting

(1) Auto Selection

[SYSTEM] F·1 FORMAT F·1 MODE : AUTO]	
From the video signal format setup menu that appears by pressing the SYSTE	M
key followed by F·1 FORMAT, press F·1 MODE to select AUTO. The inp	out
SDI signal can be detected from the supported formats for an automatic setu	ıр.

(2) Manual Selection

[SYSTEM]	F·1 FORMAT	F·1 MODE : MANUAL]
From the video	signal format setup	menu that appears by pressing the SYSTEM
key followed by	/ F·1 FORMAT, pr	ress F·1 MODE to select MANUAL. You can
manually set th	ne signal among th	ne supported video formats.

Here, if you press $\boxed{F\cdot3}$ SCANNING to select 720p, you can press $\boxed{F\cdot4}$ FRM/ FLD FREQ to select from the following: [60 / 59.94 / 50 / 30 / 29.97 / 25 / 24 / 23.98]

For details on the video signal format settings, see section **5.4**, **"Video Signal Format Setting**" in this Instruction Manual.

14.4.2 Time Code

The time code that is multiplexed in the SDI signal can be displayed in the time display area of the information display shown at the top section of the LV 5750 screen. You can select LTC or VITC as defined by SMPTE RP-188 to display the time code.



Figure 14.1 Time Code Display and Information Display Menu

14.4.2.1 Time Code Display

[SYSTEM] F·3 DISPLAY F·1 INFORMATION F·3 TIME : REAL / TIMECODE / OFF]

To display the time code, press the SYSTEM key, press $\overline{F\cdot3}$ DISPLAY, and then press $\overline{F\cdot1}$ INFORMATION to show the information display menu. Then, press $\overline{F\cdot3}$ TIME to select TIMECODE.

REAL: Displays the current time (built-in real-time clock)

TIMECODE: Displays the time code

OFF: Does not display the time

14.4.2.2 LTC/VITC Selection

[SYSTEM] F·3 DISPLAY F·1 INFORMATION F·5 TIMECODE : LTC / VITC]

You can select LTC or VITC as defined by SMPTE RP-188 to display the time code. To make the selection, press $\boxed{F\cdot5}$ TIMECODE from the information display menu to select LTC or VITC.

LTC: Decodes and displays the LTC as defined by SMPTE RP-188.

VITC: Decodes and displays the VITC as defined by SMPTE RP-188.

14.4.3 Remaining Battery Power Display

[SYSTEM] F·3 DISPLAY F·5 BATTERY : IDX / ANTON / OTHERS / OFF] The LV 5750 can display an approximate amount of remaining power in an IDX or Anton/Bauer battery.

This remaining power display is displayed when a battery is used.

To display the remaining battery power, press the SYSTEM key and press $\boxed{F\cdot3}$ DISPLAY.

From the display setup menu that appears, press $\boxed{F\cdot5}$ to specify the corresponding battery manufacturer. If you are using a battery from a manufacturer that is not listed, select OTHERS. If you set to OFF, the remaining power is not displayed.

IDX: Select this when using an IDX battery.

ANTON: Select this when using a Anton/Bauer battery.

OTHERS: Select this when using a 14.4-V lithium ion battery other than IDX or Anton/Bauer such as a battery made by SONY.

The remaining power is not displayed correctly on other types of batteries.

In this setting, the remaining power is detected from the terminal voltage of the battery. Therefore, the indication may not be correct depending on the battery type.

OFF: Does not display the remaining power.

If you are using an AC adapter that connects to the battery mount, select OFF to not display the remaining power.

In addition, if you select IDX or ANTON when you are using a battery other than IDX or Anton/Bauer, the remaining power is not displayed correctly.

The remaining battery power is displayed using four level as shown below.



Figure 14.2 Remaining battery power display

* The OTHERS setting described above is valid on products with firmware version 2.5 or later.

On earlier versions of the product, the remaining power can be displayed only for IDX and Anton/Bauer batteries.

14.4.4 SNMP Support

[SYSTEM] F-4 INTERFACE F-3 SNMP F-1 SNMP : OFF / ONLY /WRITE] If the Ethernet Unit, an option sold separately, is installed in the LV 5750, error notification and remote control can be carried out using SNMP (Simple Network Management Protocol). To use the SNMP function, press the SYSTEM key and press $\boxed{F\cdot4}$ INTERFACE. From the interface setup menu that appears, press $\boxed{F\cdot3}$ SNMP.

Then, press $\boxed{F \cdot 1}$ SNMP to disable or enable the SNMP in read-only mode or read/write mode.

- OFF: Disables the SNMP function.
- ONLY: Enables the SNMP in read-only mode.
- WRITE: Enables the SNMP in readable and writable mode.

The interface setup menu appears only when the Ethernet Unit, an option sold separately, is installed.

14.5 Bar Display

Peak levels of video signals can be displayed in place of the vectors.

The bar display that shows the peak levels can display five signals, intensity signal (Y), green (G), blue (B), red (R), and composite (COMP). The LV 5750 also operates in vectorscope mode to support multi screen display.

[VECTOR F·6 SELECT VECTOR / BAR]

To enable the bar display, press the VECTOR key to display the vector display menu. Then, press $\boxed{F \cdot 6}$ SELECT to select BAR. If you select VECTOR, the vectorscope display is enabled.

The bar display of G, B, R, and COMP signals shows the levels through a matrix conversion of the Y, C_B , C_R of the SDI signal.



Figure 14.3 Bar display

14.5.1 Limit Setting

The bar display is normally shown in cyan, but the section exceeding the specified level is shown in red. The selectable range of limits is as follows:

- YUpper limit:The limit is fixed to 100% and cannot be changed.Lower limit:The limit is fixed to 0% and cannot be changed.
- G,B,R Upper limit: The limit can be set in the range of 90.0 % to 109.4 %.
 Lower limit: The limit can be set in the range of -7.2 % to +6.0 %.
 The limit settings are linked to the gamut error threshold level.
 For the procedure to set the limit, see section 11.7.1 "Gamut Error Threshold Level" in this Instruction Manual.

CMP Upper limit: The limit can be set in the range of 90.0 % to 135.0 %. Lower limit: The limit can be set in the range of -40.0 % to -20.0 %. The limit settings are linked to the composite gamut error threshold level. For the procedure to set the limit, see section **11.7.2**, "**Composite Gamut Error Threshold Level**" in this Instruction Manual.

14.5.2 Gamut Display Frequency Characteristics

To eliminate transient gamut errors such as overshoot on the bar display, the peak level is detected through a low pass filter with the following characteristics.

HD-SDI: Approx. 5 MHz low pass filter SD-SDI: Approx. 1.8 MHz low pass filter

14.5.3 Miscellaneous

Detection is performed on all lines regardless of the line selector ON/OFF setting on the bar display.

14.5.4 Details of Bar Display

Bar Display Contents



Figure 14.4 Bar display contents

Component Bar Display Contents





· Composite Bar Display Contents



Figure14.6 Bar display contents of composite signal

14.6 Status Display

14.6.1 Supplementary Information Concerning the Status Display

(1) Error Display

The status display shows "NORMAL" when there is no error in each measurement item and shows "ERROR" when an error occurs. When an error occurs, the "ERROR" display is held for 1 second.

(2) Compact Flash Card

When saving data such as error logs and data dumps to a compact flash card, use a card of a flash memory type. Micro drives are not supported.

(3) EDH Error Detection

If the EDH error detection is turned ON while applying an SD-SDI signal and a TRS error occurs, the EDH item displays "- - - -". If this happens and the TRS error detection is ON, the error is counted as a TRS error but not as an EDH error. In addition, EDH is not detected when a TRS error occurs regardless of whether the TRS error detection setting is ON or OFF.

(4) Reserved Area Error

The reserved area error is a measurement item that results in error when the data assigned to TRS or ADF such as 000h or 3FFh occurs outside a given area. The reserved area error also occurs when a TRS error occurs.

14.6.2 Audio CRC Detection Function

14.6.2.1 Status Display of the Audio CRC

A CRC error detection function of audio signals has been added to the status display that is shown when the STATUS key is pressed. The audio signal CRC error is detected by calculating the CRC of the channel status bits in the audio signal that is multiplexed in the SDI signal.

However, when the channel status bits indicate the Consumer, error detection is not carried out, but it becomes the display of "WARNING."

1080i / 59.94	4 YCbCr 볩 2004.	/03/01 🕒 10:00:00			
SDI	SIGNAL TRS LINE NUMBER	DETECT NORMAL NORMAL	FORMAT	NORMAL	
		NORMAL			
	GAMUT	NORMAL	COMP GAMUT	NORMAL	
ANC		NORMAL	COMIT: OAMOT	NORMAL	
ANO	CHECKSUM	NORMAL			
	BCH	NORMAL			
/ CDTC	CRC	NORMAL			
	CHANNEL	1,2,3,4,5	5.6.7.8.		
		9.10.11.12.13	3.14.15.16		
ETC	CABLE LENGTH	50m			
	ERROR COUNT	0	FROM RESET	00: 10: 00	
	LOG MODE	NOW LOGGING			
·					
LOG	DATA	AUDIO A	NC ERROR	CABLE	ERROR
	DUMP	PAG	CKET CONFIG	LENGTH	RESET
F·1	F·2	F·3 F		F·6	F·7

The audio CRC detection function is turned OFF by default.

Figure 14.7 Status screen and status display menu

14.6.2.2 Event Log

The result of audio CRC detection can be logged to an event log. When an audio CRC error occurs, "CRC_ERR" is indicated in the area displaying the error information.

For details on event logs, see section **11.2**, "Event Log" in this Instruction Manual.

14.6.2.3 Turning the Error Detection ON/OFF

 $\begin{bmatrix} STATUS & F \cdot 5 \end{bmatrix} ERROR CONFIG & F \cdot 3 \end{bmatrix} ERROR DETECT & F \cdot 6 \end{bmatrix} next menu & F \cdot 6 \end{bmatrix} next menu & F \cdot 3 \end{bmatrix} AUDIO CRC : ON / OFF \\ \hline The audio CRC detection function is turned OFF by default. \\ \hline To start the error detection, press F \cdot 5 \end{bmatrix} ERROR CONFIG from the status display menu (press the STATUS key) of Figure 4-1. Then, press F \cdot 3 BRROR DETECT and F \cdot 6 \end{bmatrix} RROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of Figure 4-1. Then, press F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of Figure 4-1. Then, press F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of Figure 4-1. Then, press F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 AUDIO CRC to select ON. If you select OFF, audio CRC errors are not detected. \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \end{bmatrix} RESTATUS Key of F \cdot 3 BRROR DETECT \\ \hline F \cdot 6 \\ \hline$

14.6.3 V-ANC Monitor Function

14.6.3.1 ANC_PACKET Summary Display

[STATUS F-4 ANC PACKET]

Press $\overline{F\cdot 4}$ ANC PACKET from the status display menu that appears by pressing the STATUS key to show the list of the presence/absence of ANC data that is detectable.

1080i / 59.94 YCbCr 볩 20	04/03/01 🕒	10:00:00			
ANC PACKET SUMMARY AUDIO CONTROL PACKE EDH LTC	DETECT MISSING DETECT				
	DETECT				
	DETECT	DETECT			
V-ANC ARIB V-ANC ARIB V-ANC ARIB CLOSED C CLOSED C CLOSED C NET-Q TRIGGER F USER DAT/	1 ADCAST APTION 1 APTION 2 APTION 3 PACKET A 1 A 2	DETECT DETECT DETECT DETECT DETECT DETECT DETECT DETECT DETECT DETECT DETECT			
FORMAT_ ID	V-ANC ARIB				up menu
F·1 F·2	F·3	F·4	F·5	F·6	F·7

Figure 14.8 ANC Packet Detection Display and ANC Packet Menu

1 AUDIO CONTROL PACKET

Control data for the embedded audio. Multiplex line

- HD-SDI Line 9 and 571 of the Y channel
- SD-SDI Line 12 and 275

* Details of audio control packet are displayed using STATUS F·3 AUDIO.

2 EDH (Error Detection and Handling)

Packet for detecting SD-SDI transmission errors.

When multiple devices are connected, this packet allows you to track the device in which the error occurred.

Error is detected on the full field and active picture.

Multiplex line

• 525/59.94 Line 9 and 272

- 625/50 Line 5 and 318
- * Details of EDH are displayed using STATUS F·4 ANC PACKET F·1 EDH.

For HD-SDI, EDH does not display.

- ③ LTC (Linear/Longitudinal Time Code)
 A type of time code that is multiplexed in the frame once.
 Complies with the SMPTE RP-188 standard.
- ④ VITC (Vertical Interval Time Code)
 A type of time code that is multiplexed in the field once.
 Complies with the SMPTE RP-188 standard.
- (5) PAYLOAD (Payload ID)
 A packet for identifying the SDI video signal format.
 Complies with the SMPTE 352M-2002 standard.
 * Details are displayed using STATUS F·4 ANC PACKET F·2
 FORMAT_ID F·1 PAKET SELECT : SMPTE/ARIB.
- 6 EIA-708

One of the closed caption standards of a U.S. system. Closed caption data for digital video. Multiplexed in the V-ANC area

⑦ EIA-608

One of the closed caption standards of a U.S. system. Originally closed caption data for analog composite signals (multiplexed in line 21).

Multiplexed in the V-ANC area

- (8) PROGRAM (Program Description) Mainly the program description standard of a U.S. system. Multiplexed in the V-ANC area Complies with the SMPTE 334M standard.
- 9 DATA BROADCAST

Mainly the data broadcast packet of a U.S. system. Multiplexed in the V-ANC area Complies with the SMPTE 334M standard.

10 VBI

Multiplexed in the V-ANC area Complies with the SMPTE 334M standard.

(1) CLOSED CAPTION 1,2,3

Closed caption information packet of a Japanese system multiplexed in the V-ANC area. Up to 3 closed caption data can be multiplexed. Complies with the ARIB STD-B37 standard.

Multiplex line

- HD-SDI Line 19 and 582
- SD-SDI Line 18 and 281
- 12 NET-Q (Inter-Stationary Control Data)

NET-Q is called "Inter-Stationary Control Data ". Complies with the ARIB STD-B39 standard. Multiplex line

- HD-SDI Line 20 and 583
- SD-SDI Line 19 and 282
- **13 TRIGGER PACKET**

TRIGGER PACKET is called "Data Broadcast Trigger Signal". Complies with the ARIB STD-B35 standard.

Multiplex line

- HD-SDI Line 20 and 583
- SD-SDI Line 19 and 282
- 14 USER DATA 1,2

As for the content of transmission, it is data not especially decided. Complies with the ARIB STD-B35 standard.

Multiplex line

- HD-SDI Line 20 and 583
- SD-SDI Line 19 and 282

14.6.3.2 Format ID Display

[STATUS F-4 ANC PACKET F-2 FORMAT ID F-1 PACKET SELECT : SMPTE / ARIB]

The format ID is ANC data used to identify the video signal format. It is called PAYLOAD ID in the SMPTE specifications. The format ID on the LV 5750 can be set to SMPTE or ARIB. If SMPTE is selected, the PAYLOAD ID as defined by SMPTE 352 is displayed. If ARIB is selected, the current video mode that is included in the inter-stationary control data as defined by ARIB STD-B39 is displayed.

SMPTE: Displays the PAYLOAD ID as defined by SMPTE 352M. ARIB: Displays the current video mode as defined by ARIB STB-B39.

The format ID display is enabled by pressing $\boxed{F\cdot 2}$ FORMAT ID from the ANC packet menu. Press $\boxed{F\cdot 1}$ PACKET SELECT to switch between SMPTE and ARIB.

1080i / 59.94 YCbCr		
FORMAT ID DISPLAY SMPTE 3 BYTE1 10000111 VERSION ID PAYLOAD ID DIGITAL INTERFACE	52M SMPTE 352M-2002 1128(1080) LINE 1.485Gb/s	
BYTE2 0000110 TRANSPORT STRUCTURE PICTURE STRUCTURE PICTURE RATE	INTERLACED INTERLACED 30/1.001	
BYTE3 00000010 ASPECT RATIO H SAMPLING SAMPLING STRUCTURE	RESERVED RESERVED 4:4:4 GBR	
BYTE4 00000001 CHANNEL ASSIGNMENT DYNAMIC RANGE ASPECT RATIO MAPPING MODE	DUAL LINK A 100%	
BIT DEPTH	10BIT	

Figure 14.9 Format ID Display

Item	Description
BYTE1,2,3,4	The format ID consists of 4-byte data.
	This actual data is displayed in binary.
VERSION ID	Indicates the PAYLOAD ID version. Old version of
	packets are not supported.
PAYLOAD ID	Indicates the video format.
DIGITAL INTERFACE	Indicates the SDI bit rate.
TRANSPORT STRUCTURE	Indicates whether the digital interface uses a
	progressive or interlaced transport structure.
PICTURE STRUCTURE	Indicates whether the picture has been scanned as
	progressive or interlaced.
PICTURE RATE	Indicates the frame rate.
ASPECT RATIO	Indicates the image aspect ratio.
H SAMPLING	Indicates the number of horizontal Y samples.
SAMPLING STRUCTURE	Indicates the sampling structure.
CHANNEL ASSIGNMEMT	Indicates the link during dual link
DYNAMIC RANGE	Indicates the dynamic range of a pixel. This is not
	used in the ARIB specifications.
ASPECT RATIO	Indicates the image aspect ratio. This is not used in
	the ARIB specifications.
MAPPING MODE	Indicates the mapping mode as defined by SMPTE
	349M.
	This is not used in the ARIB specifications.
BIT DEPTH	Indicates the quantization accuracy of a pixel.

Table 14.1 Description of the format ID display

14.6.3.3 Closed Caption Data Display

[SYSTEM] F-4 ANC PACKET F-3 V-ANC ARIB F-1 CLOSED CAPTION]

An analysis (text) display of the header section and dump display of an entire packet can be shown for the digital closed caption packets as defined by ARIB STD-B37. In the ARIB standard, up the three types of digital closed caption packets can be multiplexed. One of the three types is selected and displayed. If data dump is selected, you can select binary display or hexadecimal display.

To display the digital closed caption data, press $\boxed{F\cdot3}$ V-ANC ARIB from the ANC packet menu and press $\boxed{F\cdot1}$ CLOSED CAPTION.

(1) Selection of the Digital Closed Caption Type

Since up to three types of digital closed caption is multiplexed, select the type of digital closed caption you wish to display. Select 1, 2, or 3, which corresponds the order of the closed caption data that is multiplexed in the SDI signal. Press F:2 CAPTION NUMBER to select 1, 2, or 3.

(2) Switching between Text Display and Dump Display

To switch between text display and dump display, press $\boxed{F \cdot 1}$ DISPLAY to select TEXT to enable the analysis display and DUMP to enable the dump display showing the entire packet.

(3) Dump Display Format

If the closed caption data display is set to dump, you can select the display format between binary and hexadecimal. Press $\boxed{F\cdot3}$ DUMP MODE to select BINARY for binary display or HEX for hexadecimal display.

1080i / 59.94 YCbCr 項 2004/0	03/01 🕒 10:00:00	
CLOSED CAPTION DISPLAY LINE NUMBER CLOSED CAPTION TYPE	ARIB STD-B37 19, 582 HD	
HEADER WORD1:00000000 ERROR CORRECTION CONTINUITY INDEX	NO 15	
HEADER WORD2:00000000		
HEADER WORD3:0000000 START PACKET FLAG END PACKET FLAG TRANSMISSION MODE FORMAT ID	1 0 STORAGE HD	
HEADER WORD4:0000000 C.C. DATA ID LANGUAGE ID	EXCHANGE FORMAT LANGUAGE 1	
DISPLAY CAPTION NUMBER TEXT 1	DUMP MODE HEX	up menu
F·1 F·2	F·3 F·4 F·5 F·6	F·7

Figure 14.10 Text display of the closed caption and closed caption data menu

Item	Description
LINE NUMBER	Indicates the line number to which the digital closed
	caption is multiplexed.
CLOSED CAPTION TYPE	Indicates the type of digital closed caption data.
ERROR CORRECTION	Indicates the availability of error correction.
CONTINUITY INDEX	A counter indicating the continuity of packets.
START PACKET FLAG	Indicates the start of the closed caption data group.
END PACKET FLAG	Indicates that the end TS packet is included when
	the packet is divided using MPEG-2 TS.
TRANSMISSION MODE	Indicates the transmission mode.
FORMAT ID	Indicates the type of digital closed caption packet.
C.C. DATA ID	Indicates the closed caption data ID.
LANGUAGE ID	Indicates the language ID used to send closed
	captions in multiple languages.

Table 14.2 Description of the closed caption packet display
1080i / 59.94 YCbCr 宿 2	004/03/01 🕒 10:00:00
CLOSED CAPTION DISPLA	AY ARIB STD-B37
LINE DID SDID DC 1 HEADER1 2 HEADER2 3 HEADER3 4 HEADER3 4 HEADER4 5 DATA1 6 DATA2 7 DATA3 8 DATA4 9 DATA5 10 DATA6 11 DATA7	19, 582 25F 1DF 2FF 200 200 200 200 200 200 20
DISPLAY DUMP	DUMP MODE HEX up menu F·3 F·4 F·5 F·6 F·7

Figure 14.11 Closed Caption Data in Hexadecimal Notation Display

1080i / 59.94	YCbCr 宿 200	4/03/01 🕒 10	0:00:00			
CLOSED CAP	TION DISPLAY	ARIB STD-B3	7			
LINE		19, 582				
DID	10	01011111				
SDID	0.	111011111				
DC	1(011111111				
1 HEADER1	1(000000000				
2 HEADER2	1(000000000				
3 HEADER3	1(000000000				
4 HEADER4	1(000000000				
5 DATA1	1(000000000				
6 DATA2	10	000000000				
7 DATA3	10	000000000				
8 DATA4	1(000000000				
9 DATA5	10	000000000				
10 DATA6	10	000000000				
11 DATA7	1(000000000				
12 DATA8	10	000000000				
]				
DISPLAY		DUMP				up menu
		MODE				
DUMP		BINARY				
F·1	F·2	F·3	F·4	F·5	F·6	F·7
			L			

Figure 14.12 Closed Caption Data in Binary Notation Display

14.6.3.4 Inter-Stationary Control Signal

[STATUS] $F \cdot 4$ ANC PACKET $F \cdot 3$ V-ANC ARIB $F \cdot 2$ NET-Q] An analysis (text) display of the dump display of an entire packet can be shown for the inter-stationary control signal as defined by ARIB STD-B39. If data dump is selected, you can select binary display or hexadecimal display.

(1) Switching between Text Display and Dump Display

To switch between text display and dump display, press $\boxed{F \cdot 1}$ DISPLAY to select TEXT to enable the analysis display and DUMP to enable the dump display showing the entire packet.

(2) Dump Display Format

If the $\boxed{F\cdot 1}$ DISPLAY is set to dump, you can select the display format between binary and hexadecimal. Press $\boxed{F\cdot 2}$ DUMP MODE to select BINARY for binary display or HEX for hexadecimal display.

(3) Turning the Q Signal ON/OFF

The Q bit 0/1 display of the TRIGGER SIGNAL can be turned ON/OFF independently on the text display screen. From the inter-stationary control menu, press $\boxed{F\cdot6}$ next menu. Then, press $\boxed{F\cdot1}$ to $\boxed{F\cdot5}$ to turn Q1 to Q32 ON/ OFF independently.

1080i / 59.94 Y	′CbCr 몝 20	04/03/01	① 10:00	:00				
INTER-STATIO LINE NUMBER ERROR CORR CONTINUITY II STATION COD DATE & TIME VIDEO CURRE AUDIO CURRE DOWN MIX CU	NARY CONTR ECTION NDEX E 2004/07/ NT :1080i/59. NT :5.1+S IRRENT:1/ 2	ROL DATA 20 LEA 05 10:0 94 NE> NE>	A ARIB ST 0,583 NO 15 DER 0:00 (T:1080i/5 (T:5.1+S (T:1/ 2	D-B39 9.94	COUNT	DOWN:25 DOWN:25	54 54	
TRIGGER SIGI Q1: 1	NAL Q2: 1	Q3:1	Q4:1	Q5:1	Q6:1	Q7:1	Q8:1	
Q9: 1	Q10:1	Q11:1	Q12:1	Q13:1	Q14:1	Q15:1	Q16:1	
Q17:1	Q18:1	Q19:1	Q20:1	Q21:1	Q22:1	Q23:1	Q24:1	
	Q26:1	Q27:1	Q28:1	Q29:1	Q30:1	Q3111	Q32:1	
	01:255	02.0	Q3. 0 Q3·255	Q4. 0 04·255				
STATUS SIGNA	AL	Q2.200	Q0.200	Q4.200				
S1:1	S2:1	S3:1	S4:1	S5:1	S6:1	S7:1	S8:1	
S9:1	S10:1	S11:1	S12:1	S13:1	S14:1	S15:1	S16:1	
DISPLAY							next	up menu
TEXT								
E.1	F.2	E.	3	E.4		E.5	E-6	E.7
	F'2			г.4		F.9	L L.0	

Figure 14.13 Inter-Stationary Control Signal Display and Inter-Stationary Control Menu

ltem	Description
LINE NUMBER	Indicates the line number to which the inter-stationary
	control signal is multiplexed.
ERROR CORRECTION	Indicates the availability of error correction.
CONTINUITY INDEX	A counter indicating the continuity of packets.
STATION CODE	Indicates the originating station code. The code is
	expressed using alphabet characters or Japanese
	katakana.
DATE & TIME	Indicates the originating station time. Expressed with
	a date and time.
VIDEO CURRENT	Indicates the current video mode. For details, see
	section 14.6.3.2, "Format ID."
AUDIO CURRENT	Indicates the current audio mode.
NEXT	Indicates the next video/audio mode.
COUNTDOWN	Indicates the countdown of the video/audio mode
	switching.
DOWN MIX CURRENT	Indicates the current audio down mix designation.
NEXT	Indicates the next audio down mix designation.
TRIGGER SIGNAL	Indicates the trigger signal that represents the timing.
COUNTER	Indicates a count up to the required timing for Q1 to
	Q4 of the TRIGGER SIGNAL.
COUNTDOWN	Indicates a count down to the required timing for Q1
	to Q4 of the TRIGGER SIGNAL.
STATUS SIGNAL	Indicates the status. The bit definitions are arbitrary.

Table 14.3 Description of the inter-stationary control signal display

(1) Data Dump Display (HEX)

1080i / 59.94 YCbCr INTER-STATIONARY	個 2004/03/01 CONTROL DATA	() 10:00:00 ARIB STD-B39			
LINE DID SDID DC 0 HEADER 1 STATION CODE1 2 STATION CODE3 3 STATION CODE3 4 STATION CODE4 5 STATION CODE5 6 STATION CODE6 7 STATION CODE7 8 STATION CODE8 9 YEAR	20, 583	25F 1FE 200 200 200 200 200 200 200 200 200 20			
DISPLAY DUM MOE DUMP HE	IP DE X				up menu
F-1 F-2	2 F·3	F·4	F·5	F·6	F·7

Figure 14.14 Inter-Stationary Control Signal in Hexadecimal Notation Display

(2) Data Dump Display (BIN)

1080i / 59.94 YC	bCr 宿 200	4/03/01 🕒 10	0:00:00				
INTER-STATION	ARY CONTR	ol data arib	STD-B39				
LINE DID SDID DC 0 HEADER 1 STATION CO 2 STATION CO 3 STATION CO 4 STATION CO 6 STATION CO 6 STATION CO 8 STATION CO 9 YEAR	20, 58 DE1 DE2 DE3 DE4 DE5 DE6 DE7 DE8	3 1001011111 011111111 100000000 100000000					
DISPLAY	DUMP MODE BINARY						up menu
F·1	F·2	F·3	F·4	F	·5	F∙6	F·7

Figure 14.15 Inter-Stationary Control Signal in Binary Notation Display

14.6.4 Displaying Ancillary Data Markers Using Data Dump

This function detects ancillary data that is multiplexed in the SDI signal and indicates the code using markers.

The detected code is displayed in color as indicated in the following table making it extremely visible.

Detected Code	Color	Description
ADF	Cyan	Displays the 000H, 3FFH, 3FFH (ADF) code from
		the DATA_DUMP data.
DID	Cyan	The DID code is the next data after the ADF code.
SDID	Cyan	If the DID code is smaller than 80H, SDID is
		displayed as data in the second format.
DBN	Cyan	If the DID code is greater than or equal to 80H,
		DBN is displayed as data in the first format.
DC	Cyan	Displays the data count code (DC) following the
		SDID/DBN when the ADF code is detected.
UDW	Cyan	Displays the user data word (UDW) code for the
		data count following the ADF code
CS	Magenta	Displays the CS code immediately after the UDW.
AP	Yellow	If the selected line is in the active picture area,
		the area between SAV and EAV are displayed as
		active picture AP area.

Table 14.4 Marker display of ancillary da	ata
---	-----

525i / 59.	9 省 4 200	4/03 Ō	/ 0110:00:00
DATA	DUMP LINE N	lo. 1	
	SAMPLE	COLOR	DATA
[EAV]	<1440>	Cb	3FF
[EAV]	<1441>	Y	000
[EAV]	<1442>	Cr	000
[EAV]	<1443>	Υ'	2D8
ADF	<1444>	Cb	000
ADF	<1445>	Y	3FF
ADF	<1446>	Cr	3FF
DID	<1447>	Y'	2FF
DBN	<1448>	Cb	2C0
DC	<1449>	Y	224
UDW	<1450>	Cr	290
UDW	<1451>	Y'	20D
UDW	<1452>	Cb	21F
UDW	<1453>	Y	292
UDW	<1454>	Cr	20D
UDW	<1455>	Υ'	11F

(a) Marker display around DID (serial display)

525i 🛅 / 59.94 🕒 20	04 / 03 / 0110:00:00
DATA DUMP LINE No.	1
SAMPLE	Y C. ADR Cb Cr
[EAV] < 720>	000 <360> 3FF 000
[EAV] < 721>	2D8
ADF ADF ADF < 722>	3FF <361> 000 3FF
DID < 723>	2FF
DC DBN DBN < 724>	224 <362> 2C0 290
UDW < 725>	20D
UDW UDW UDW < 726>	292 <363> 21F 20D
UDW < 727>	11F
UDW UDW UDW < 728>	20D <364> 294 11F
UDW < 729>	296
UDW UDW UDW < 730>	21F <365> 20D 170
UDW < 731>	1BC
UDW UDW UDW < 732>	172 <366> 21E 1BC
UDW < 733>	11E
UDW UDW UDW < 734>	1BC <367> 174 11E
UDW < 735>	176

(b) Marker display around DID (component display)

525i / 59.	9 볩 4 200	4/03 🕒	/ 0110:00:00
DATA	DUMP LINE N	o. 1	
	SAMPLE	COLOR	DATA
UDW	<1484>	Cb	200
UDW	<1485>	Y	200
CS	<1486>	Cr	IE9
ADF	<1487>	Y'	000
ADF	<1488>	Cb	3FF
ADF	<1489>	Y	3FF
DID	<1490>	Cr	IFE
DBN	<1491>	Y'	26A
DC	<1492>	Cb	206
UDW	<1493>	Y	277
UDW	<1494>	Cr	177
UDW	<1495>	Y'	2BB
UDW	<1496>	Cb	1BB
UDW	<1497>	Y	200
UDW	<1498>	Cr	100
CS	<1499>	Y'	1D2

525i 💼	/ 59.94 🕒	2004 / 03 / 0	110:00:00
DATA	DUMP LINE N	lo. 1	
	SAMPLE	COLOR	DATA
[SAV]	<1712>	Cb	3FF
[SAV]	<1713>	Y	000
[SAV]	<1714>	Cr	000
[SAV]	<1715>	Υ'	200
AP	< 0>	Cb	200
AP	< 1>	Y	040
AP	< 2>	Cr	200
AP	< 3>	Y'	040
AP	< 4>	Cb	200
AP	< 5>	Y	042
AP	< 6>	Cr	200
AP	< 7>	Y'	059
AP	< 8>	Cb	200
AP	< 9>	Y	0D6
AP	< 10>	Cr	200
AP	< 11>	Y'	1F6

(c) Marker display around CS (serial display)

(d) Marker display around AP (serial display)

Figure 14.16 Marker display screen of ancillary data

15. SDI-EXT REF PHASE DIFFERENCE DISPLAY FUNCTION

For Software Version 2.8 and later

15.1 General

Followings are functions that are added when the software version of LV 5750 is updated to 2.8 and later.

This chapter describes the additional functions.

The SDI-EXT REF phase display function shows the phase difference between the SDI signal and the external sync signal (EXT REF).

15.2 Features

• Displays numerically and graphically the phase difference between the SDI signal and external sync signal (EXT REF).

Conventionally, the phase of the SDI signal was adjusted by switching between internal synchronization and external synchronization and monitoring the phase difference in the waveforms on the WFM display. The SDI-EXT REF phase display allows you to easily check and adjust the phase difference.

- Displays the phase difference between SDI signals
 The relative phase difference between SDI signals can be displayed by setting
 the phase difference between an arbitrary SDI signal and EXT REF signal to
- zero.
 Records up to eight phase differences
 Because up to eight phase differences of SDI signals can be stored, the phase data of eight systems of SDI signals from the switcher can be held.

15.3 Specifications

- Supported Formats 1080i/60,59.94,50 1080p/30,29.97,25,24,23.98 1080PsF/30,29.97,25,24,23.98 720p/60,59.94,50,30,29.97,25,24,23.98 525i/59.94 625i/50
- * When applying an HD tri-level sync signal to EXT REF, the frame and line frequencies between the EXT REF and SDI signal must be the same.
- * When applying a B. B signal to EXT REF, the synchronizable formats are as follows:

B. B Format	SDI Format
525i/59.94	1080i/59.94, 1080p/29.97, 1080PsF/29.97, 720p/59.94, 525i/59.94
625i/50	1080i/50, 1080p/25, 1080PsF/25, 720p/50, 625i/50

Phase Difference Display Range

V direction ± 1 frame with respect to the center (for interlace)

±1/2 field (for progressive)

H direction $\pm 1/2$ line with respect to the center

The direction of V and H both are expressed with the Delay axis between approx. +1/2 frame and the center, and, similarly, are expressed with the Advance axis between approx. -1/2 frame and the center.

* The phase difference display in the H direction may fluctuate in the range of ±1 clock when the signal is switched.

	Displayed with an Advance				vance axis				
Format				Displayed with a			Delay axis		
Format	V PHASE	H PHASE		V PHASE	H PHASE		V PHASE	H PHASE	
	[Lines]	[us]		[Lines]	[us]		[Lines]	[us]	
1080i/59.94, 1080p/29.97, 1080PsF/29.97	-562	-29.645	~	0	0	~	562	0	
1080i/60, 1080p/30, 1080PsF/30	-562	-29.616	~	0	0	~	562	0	
1080i/50, 1080p/25, 1080PsF/25	-562	-35.542	~	0	0	~	562	0	
1080p/23.98, 1080PsF/23.98	-562	-37.060	~	0	0	~	562	0	
1080p/24, 1080PsF/24	-562	-37.023	~	0	0	~	562	0	
720p/59.94	-375	0	~	0	0	~	374	22.230	
720p/60	-375	0	~	0	0	~	374	22.208	
720p/50	-375	0	~	0	0	~	374	26.653	
720p/29.97	-375	0	~	0	0	~	374	44.475	
720p/30	-375	0	~	0	0	~	374	44.430	
720p/25	-375	0	~	0	0	~	374	53.319	
720p/23.98	-375	0	~	0	0	~	374	55.597	
720p/24	-375	0	~	0	0	~	374	55.542	
525i/59.94	-262	-63.518	~	0	0	~	262	0	
625i/50	-312	-63.962	~	0	0	~	312	0	

Details of display range of the Delay axis and Advance axis are as follows.

Graphic Display

In case of the direction of H, the Delay axis, and the Advance axis, the distance from the center to the edge is equivalent to the time of 1H.

In case of the direction of V, the Delay axis, and the Advance axis, the distance from the center to the edge is equivalent to the time of approx. 1/2 frame.

In the phase difference of approx. $\pm 1/2$ frame from the center, the display position of the marker in which phase difference is shown is moved from a Delay axis to an Advance axis (or the contrary).



Figure 15.1 Graphic display



The marker turns green when the V marker is within ± 0 lines of the center and the H marker is within ± 3 clocks of the center.

15.4 MENUE Structure



15.5 Operation

Press VECT on the panel, and press $F \cdot 6$ SELECT to choose EXTPHASE. The SDI-EXT REF phase display of Figure below appears. This display shows the timing of the SDI signal with respect to the EXT REF signal. If the SDI signal lags the EXT REF signal, Delay (+) is indicated. If the SDI signal is ahead of the EXT REF signal, Advance (-) is indicated.

If you press $\overline{F \cdot 7}$ next menu then $\overline{F \cdot 1}$ EXTREF PHASE on the SDI-EXT REF phase display, the EXTREF PHASE function menu appears.



Figure 15.2 SDI-EXT REF phase display and function menu

- (1) The phase difference between the SDI and EXT REF signals that are currently applied is shown numerically in the CURRENT PHASE section on the left side of the display shown in Figure 15.2. In the TOTAL PHASE item, the phase difference is displayed as time of total.
- (2) The graph on the right shows the line difference in the V direction along the vertical axis, and the time difference in the H direction along the horizontal axis. The phase difference is zero when the two circles that indicate the V and H phase differences overlap at the center.

The marker changes from white to green when the phase between the SDI and EXT REF signals comes within ± 3 clocks.

The H marker turns green when it comes within ± 3 clocks of the center.

The V marker turns green when it comes within ±0 line of the center.

- (3) The default zero reference on the SDI-EXT REF phase display is defined on the condition that the SDI output of the LEADER's LT 443D MULTIFORMAT VIDEO GENERATOR and the REF B. B signal with no timing offset are connected to the LV 5750 with cables of equal length.
- (4) Pressing the USER REF SET key sets the phase difference between the SDI and EXT REF signals that are being applied to zero. This function allows the reference to be set arbitrarily to match your system environment.

This function cannot be set when the phase difference is zero.

Pressing the REF DEFAULT key sets the phase difference to factory default zero reference.

Since it is displayed on the item of the REF display as follows, it can be recognized which standard is chosen.

EXT REF is a B.B signal, and when DEFAULT is set effectively, REF EXT BB > DEFAULT is displayed.

EXT REF is a B.B signal, and when USER REF is set effectively, REF EXT BB > USER REF is displayed.

(5) SDI PHASE MEMORY is a function used to store the phase difference that is being measured in CURRENT PHASE.

Up to eight instantaneous measured values can be stored such as when adjusting the phase by changing systems with a switcher.

Press SDI MEMORY to store the current measured value in the box of the selected SDI NUMBER.

(6) REF Display Description

The REF display at the lower left of the display shown in figure 15.1 automatically indicates the following according to the synchronization (REFERENCE) setting.

In the case of internal sync, the following is shown.

REF INT

If you select external sync and the external sync signal is HD tri-level sink, the following is shown.

REF EXT HD

If you select external sync and the external sync signal is SD bi-level sink, the following is shown.

REF EXT BB

If you select external sync but there is no external sync signal, the following is shown.

REF NO SIGNAL

15.6 Zero Reference

The default zero timing reference of the SDI and EXT REF signals is defined on the condition that the SDI output of the LEADER'S LT443D MULTIFORMAT VIDEO GENERATOR and the REF B. B signal with no timing offset are connected with cables of equal length.

- * The zero reference measurement may not be exactly zero due to the variation in the delay of the LV 5750.
- * The phase difference display in the H direction may fluctuate in the range of ±1 clock when the signal is switched.

For other configurations, the zero reference will be offset. Therefore, use USER REF SET.